



OCCASION

This publication has been made available to the public on the occasion of the 50th anniversary of the United Nations Industrial Development Organisation.



DISCLAIMER

This document has been produced without formal United Nations editing. The designations employed and the presentation of the material in this document do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations Industrial Development Organization (UNIDO) concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries, or its economic system or degree of development. Designations such as "developed", "industrialized" and "developing" are intended for statistical convenience and do not necessarily express a judgment about the stage reached by a particular country or area in the development process. Mention of firm names or commercial products does not constitute an endorsement by UNIDO.

FAIR USE POLICY

Any part of this publication may be quoted and referenced for educational and research purposes without additional permission from UNIDO. However, those who make use of quoting and referencing this publication are requested to follow the Fair Use Policy of giving due credit to UNIDO.

CONTACT

Please contact <u>publications@unido.org</u> for further information concerning UNIDO publications.

For more information about UNIDO, please visit us at www.unido.org

21947-E

Distr.
RESTRICTED

GM/R.11 21 April 1997

UNITED NATIONS INDUSTRIAL DEVELOPMENT ORGANIZATION

ORIGINAL: ENGLISH

Industrial Development Decade for Africa (IDDA) II

Report of the Mid-Term Programme Evaluation*

Prepared by:
Organization of African Unity (OAU)
Economic Commission for Africa (ECA)
United Nations Industrial Development Organization (UNIDO)

^{*}This document has not been formally edited.

Contents

		Page
AC	RONYMS	5
SU	MMARY OF CONCLUSIONS AND RECOMMENDATIONS	7
1.	INTRODUCTION	10
2.	CONCEPTUAL AND POLICY FRAMEWORK FOR THE SECOND IDDA A. Lagos Plan of Action B. The First IDDA C. The Second IDDA	13 13 13 13
3.	BACKGROUND TO AFRICA'S INDUSTRIALIZATION IN THE 1990s A. Internal Factors B. External Factors	17 17 19
4.	ASSESSMENT OF THE PROGRAMME IMPLEMENTATION OF IDDA II A. Mechanism for Implementation, Monitoring and Coordination	23
	of the IDDA II Programme	23
	Programmes	25
	C. Assessment at Sub-regional/Regional Level	31
	D. Joint Secretariat Support to Implementation of the IDDA II Programme	35
	i. UNIDO Support Programme	35
	ii. ECA Support Programmeiii. OAU Support Activities	40 42
5.	CONSTRAINTS TO THE IMPLEMENTATION OF THE PROGRAMME	
	OF THE SECOND IDDA	44
	A. At National Level	44
	B. At Sub-regional Level	47 50
6.	MAJOR FINDINGS AND CONCLUSIONS	51
	A. Perception of the IDDA II Programme	51
	B. Relations between Government and Private Sector	52
	C. Consolidation and Expansion Programmes	54
	D. Economic Liberalization & Industrial Competitiveness	55
	E. Financial Resource Mobilization for Industry	56
	F. Agro-Industry Linkages	58
	G. Human Resource and Entrepreneurship Development	59 61
	H. Physical and Institutional Infrastructure	62

		Page		
7.	THE WAY FORWARD: Lessons and Recommendations	65		
	A. Private Sector Development	68		
	B. Human Resource Development	69		
	C. Regional and Sub-regional Co-operation and Integration	70		
	D. Environmental Concerns	71		
	Financial Sector Development	72		
	Physical Infrastructure Development	72		
	Governance, Political Liberalization and Commitment	72		
	Alliance for Africa's Industrialization	73		
	Annexes			
1.	Terms of Reference	75		
2.	Field Missions			
3.	\boldsymbol{c}			
4.	a. List of IDDA Technical Assistance (XA) projects 1994-1995	87		
	b. List of IDDA Supplementary Activities (YA) Projects 1994-1995	92		
	c. List of IDDA Technical Assistance (XA) projects 1996-1997	96		
_	d. List of IDDA Supplementary Activities (YA) Projects 1996-1997	98		
5.	List of People Met in the Secretariats of OAU, ECA and UNIDO	100		
6.	List of People Met and Institutions visited during the Field Mission	101		

ACRONYMS

AAI Alliance for Africa's Industrialization
ACBF African Capacity Building Foundation

ADB African Development Bank
AEC African Economic Community

AIDF African Industrial Development Fund AIDMO Arab Industrial and Mining Organization

AIHTTR African Institute for Higher Technical Training & Research

AMU Arab Maghreb Union

ARCEDEM African Regional Centre for Engineering, Design and Manufacturing

ARCT African Regional Centre for Technology

ARNAM Arab Regional Network for Agricultural Machinery
ARSO African Regional Organization for Standardization

ASEAN Association of South-East Asian Nations

BDEAC Banque de Développement des Etats de l'Afrique Centrale

BERETEC Technical Design and Construction Office (Congo)

BMC Botswana Meat Corporation
BOT Build-Operate-and-Transfer

CAFRAD Centre Africain de Formation et de Recherche Administrative pour le

Développement

CAMI Conference of African Ministers of Industry

CAPME National Centre for Assistance to Small and Medium Size Enterprises

(Cameroon)

CEAO West African Economic Community

CEMAC Communauté Economique et Monétaire de l'Afrique Centrale CENEEMA Centre for Research and Agricultural Machinery and Tools

CEPEX Export Promotion Centre (Tunisia)

CEPGL Economic Community of the Great Lake Countries
CFA Comité Monétaire de la Zone Francophone Africaine
CNCC Technical Assistance to National Centre for Leather Shoes

COMESA Common Market for Eastern and Southern Africa

EAC East African Community

EADB East African Development Bank ECA Economic Commission for Africa

ECCAS Economic Community of Central African States

ECDC/TCDC Economic Co-operation among Developing Countries/Technical Co-

operation among Developing Countries

ECOWAS Economic Community of West African States EDP Entrepreneurship Development Programme

EPZ Export Processing Zone

ESABO Eastern and Southern Africa Business Organization

FDI Foreign Direct Investment

FEAP Family Economic Advancement Programme

FSFC Federal Super Phosphate Fertilizer Company (Kaduna, Nigeria)

GDP Gross Domestic Product

HRD Human Resource Development

IDA International Development Assistance

IDB Islamic Development Bank

IDDA Industrial Development Decade for Africa

IDEP African Institute for Economic Development and Planning

IGO Intergovernmental organization
ILO International Labour Organization
IMF International Monetary Fund
IMP Industrial Master Plan

ISTA Sub-regional Multisectoral Institute of Applied Technology and Project

Planning and Evaluation (Gabon)

LDCs Least Developed Countries

LLPI Leather and Leather Products Institute

LPA Lagos Plan of Action

MCI Ministry of Commerce and Industry MERCOSUR Southern Core Common Market

MIGA Multilateral Investment Guarantee Agency

MRU Mano River Union

MTC Metallurgical Technology Centre

MULPOC Multinational Programming and Operational Centre

MVA Manufacturing Value Added

NAFCON National Fertilizer Company of Nigeria NAFTA North American Free Trade Agreement

NCID National Committee on Industrial Development

NDC Namibia Development Corporation NGO Non-Governmental Organization

NIPRD National Institute for Pharmaceutical Research and Development

OAU Organization of African Unity
ODA Overseas Development Assistance
ODC Offshore Development Company

PTA Preferential Trade Area for Eastern and Southern African States

R&D Research and Development

RIPCO Botswana Technology Centre - Rural Industries Promotion Company

RMC Resource Mobilization Committee

SADC Southern African Development Committee

SAP Structural Adjustment Programme

SIRDCO Scientific and Industrial Research and Development Centre (Zimbabwe)

SMEs Small and Medium Enterprises SMI Small and Medium Industry

SMID Strategic Management of Industrial Development
TPCS Training and Production Centre for Shoe Industry
UDEAC Central African Customs and Economic Union
UEMOA Union Economique et Monétaire Ouest Africaine

UNDP United Nations Development Programme

UNIDO United Nations Industrial Development Organization
UNTACDA United Nations Transport and Communication Decade

WAMA West African Monetary Agency

WB World Bank

WTO World Trade Organization
ZIC Zimbabwe Investment Centre

SUMMARY OF CONCLUSIONS AND RECOMMENDATIONS

- 1. The mid-term evaluation reviews the implementation of the IDDA II Programme over the period 1993-1996 at the national, sub-regional and regional levels, explains the difficulties that have been encountered and outlines proposals for the remaining period of the Development Decade.
- 2. The IDDA Programme was launched at a time when many African countries had already initiated action programmes and policies in response to the crises of the 1980s.
- 3. The crises of the eighties which had economic and political dimensions were endogenous and exogenous in nature, and therefore the African response had to be found within the international context.
- 4. Economic reforms and policy approaches were adopted by many countries of the region in coping with the crises which had been attributed principally to heavy external debt and the servicing burden, collapse of world commodity prices and accompanying deterioration of terms of trade, and unfavourable climatic conditions resulting in food shortages, poor physical and institutional infrastructure for lack of maintenance and of new investments, and the application of inappropriate policy measures which often created instability in the main macro-economic variables. Such reforms overshadowed the Programme.
- 5. The adoption by many African countries of Structural Adjustment Programmes was one of the main responses to the crises. This was intended to liberalize economic activity in general by the introduction of such measures as exchange rate adjustments, removal of subsidies, market pricing of export commodities, development of the private sector as the main engine of growth, and expansion of its scope through privatization of public assets.
- 6. The second major response was the transition from one-party to multi-party democracy which was expected to bring about political stability through good governance, the rule of law and the protection of human rights.
- 7. The perception of the IDDA as an action-oriented programme for industrialization had dwindled to the point where in most countries segments of government other than Ministries of Industry and Trade were unaware of the Programme's existence. This lack of awareness is even more pronounced within the private sector.
- 8. The dim perception and unawareness of the IDDA Programme was found to be directly related to the absence of special mechanisms for the implementation, coordination and monitoring of the Programme. IDDA's existence would have been kept alive if the recommended mechanisms had been operational and their activities publicized.
- 9. The private sector in most countries is weak and incapable of assuming responsibility as principal engine of industrial growth. Many governments have, however, articulated policies and introduced measures for private sector development by way of institutional support, training and establishment of consultative mechanism between public and private sectors on policy formulation and implementation.
- 10. Privatization has been widely and sometimes indiscriminately adopted by many African countries. Reasons for privatization include: governments ridding themselves of state involvement in productive sectors of the economy, particularly manufacturing; widening the scope of private sector

activities; attracting fresh foreign investments, technology and management skills into the economy; getting rid of unprofitable and loss-making state enterprises; etc.

- 11. Structural Adjustment Programmes have been adopted by many governments with varying degrees of success. Among the adjustment measures is economic liberalization which has had both positive (industrial competitiveness) and short-term adverse results on some existing industrial enterprises, particularly textiles, leather and leather products.
- 12. Though dependence on external sources of financing persists, the recent slackening of ODA and FDI will impose additional strain in programme implementation. Mobilization of domestic financial resources has not been enhanced for lack of new approaches to domestic savings, remittances, etc.
- 13. Although agro-based industries have high priority in most countries (by virtue of forward and backward linkages) agricultural production, by which industrial raw materials could be provided, still lags behind. Dependence on rainfed agriculture has placed most countries at the mercy of weather conditions, yet large-scale irrigation schemes are too costly and far beyond the reach of many countries.
- 14. Industrial development has been hindered, to a large extent, by lack of adequately trained manpower, highly motivated entrepreneurs and skilled management personnel. Although efforts are being made in almost all countries to rectify these deficiencies through institutional support and relevant training schemes, the situation is far from satisfactory.
- 15. Inadequate physical infrastructure development and institutional capacity building have remained a major hindrance to socio-economic progress. Industrial development will hardly move ahead unless and until physical and institutional infrastructure as well as human resource development, which together constitute essential prerequisites, are given maximum support in financial terms by development partners.
- 16. Sub-regional institutions which have been created to support the industrial development efforts have in most cases found to:
 - lack adequate financial support;
 - lack skilled manpower;
 - lack financial incentive and motivation resulting into high turnover of competent staff;
 - lack adequate support facilities;
 - compete for external resources, functions/roles with other overlapping/duplicating institutions:
 - lack co-ordination and harmonization of programmes between themselves.

While these responses and other prevailing factors were taken into account in formulating the IDDA II priorities, which were based on the objectives of IDDA I and the general principles of the Lagos Plan of Action, the emergent situation from the application of reform measures and policy approaches has shifted the focus from the IDDA Programme. The most notable example is the commitment to the private sector and the redefinition of its role as the principal engine of growth with private enterprises as determinants of industrial priorities. Governments are to create the enabling environment including

physical infrastructure and human resource development that will support and sustain the private sector and its business community in the industrialization process.

- 17. The projects developed, designed and implemented by UNIDO in the various sectors and subsectors at the national and sub-regional levels were compatible with the overall goals/objectives of the IDDA II national programmes objectives. Within the limited resources available, projects were developed as far as practical to meet the broad range of activities envisaged for implementation under the Programme. Thus projects which were delivered were relevant to IDDA Programme objectives. While these individual projects were normally efficiently delivered, the evaluation could not conclude how much the results achieved contributed to the overall goals of the Programme.
- 18. The concept of the "seed money" within the IDDA framework had a marginal effect in terms of attracting funding from other sources.
- 19. The evaluation provides some guidance for the relationship between the IDDA and the Alliance. During their implementation, national governments, with eventual support from the participating organizations, should ensure that IDDA and the Alliance are complementary. In short, the Alliance should be used as a tool to reach the goals of IDDA.

1. INTRODUCTION

- 20. The Conference of African Ministers of Industry, at its twelfth meeting held in Gaborone, Botswana in June 1995, in its resolution 1(XII) requested African countries, the Economic Commission for Africa (ECA), the United Nations Industrial Development Organization (UNIDO) and the Organization of African Unity (OAU) to undertake in 1997 a mid-term evaluation on the implementation of the Programme for the Second IDDA.
- 21. The Fourteenth Session of the Industrial Board which was held from 26-30 June 1995, adopted decision IDB.14/Dec.16 to conduct an in-depth evaluation of the IDDA Programme and its activities.
- 22. The evaluation was jointly guided by the Secretariats of OAU, ECA and UNIDO. The Secretariats prepared a detailed terms of reference, which elaborated the purpose and scope of the evaluation as well as the methodology adopted for the exercise. (See Annex 1: Terms of Reference)
- 23. The purpose of the mid-term evaluation as defined in the Terms of Reference is to enable African member states, UNIDO, OAU and ECA to make an objective assessment of the achievement of the Programme against set objectives and to make forward looking recommendations to take informed decisions on the future course of action.
- 24. The Terms of Reference described the scope of the evaluation to be:
 - to assess the achievements of the Programme against its stated objectives and expected outputs;
 - to identify and assess the factors that have facilitated the achievement of the Programme's objectives, as well as those factors that have impeded the fulfilment of those objectives;
 - to propose a course of action, both at policy and programme implementation level and to draw lessons therefrom for the design, implementation and monitoring of the Programme for the future; and,
 - to examine the relationship of the IDDA II Programme to the "Alliance for Africa's Industrialization", the new initiative that was launched by African Heads of State and Government, in Abidjan, Côte d'Ivoire, on 23 October 1996.
- 25. The evaluation was conducted in three phases.
 - Phase 1: The conceptual stage;
 - Phase 2: Field Missions; and
 - Phase 3: Preparation of the final evaluation report.
- 26. Two consultants were recruited for the exercise:
 - Ambassador K. Adusei Poku (Team Leader)
 - Mr. Guy M'Bengue
- 27. Phase 1: The conceptual stage (desk review) was conducted at the level of the Secretariats. It was devoted to reviewing the IDDA Programme to date and making an assessment of available

documentation and information prepared to promote, coordinate, implement, monitor and evaluate the IDDA Programme.

- 28. Phase 2: Field missions were undertaken to assess on the spot, progress towards industrialization and the impact of IDDA II at country and sub-regional levels. The field missions as stipulated in the terms of reference, covered 21 African countries selected from all sub-regions. The evaluation team was supported by the staff of the three Secretariats which divided itself into six groups so as to cover the countries in the four sub-regions (see Annex 2). Criteria for selecting countries were: countries which prepared national IDDA II programmes; geographical distribution, financial and time constraints. In most of the countries visited, the evaluation team was assisted by a national consultant specially recruited to make a pre-report on national IDDA programmes.
- 29. **Phase 3:** The final phase was devoted to reviewing all the findings that emerged from the first two phases and to the drafting of the evaluation report.
- 30. The progress and status of undertaking of each phase were presented to the UNIDO Advisory Committee established for the evaluation exercise by the representatives of OAU, ECA and UNIDO and mission team members. The mid-term evaluation of IDDA II has been carried out during the period of mid-January 1997 to end-March 1997.
- 31. Permanent representatives of the African missions to UNIDO Headquarters in Vienna were briefed on the evaluation methodology and Terms of Reference on 15 January 1997.
- 32. When the field missions were completed, the evaluation team had different meetings in Addis Ababa and at UNIDO Headquarters in Vienna with ECA, UNIDO and OAU officials to synthesize the findings and recommendations.
- 33. A presentation of the main findings and conclusions of the evaluation was made during the CAMI 12 Bureau meeting in Mombasa, 13-14 March 1997. The 12th CAMI Bureau fully endorsed the evaluation's main findings and conclusions. An individual briefing of the evaluation was also made and discussed with selected major donors Germany, France and Denmark on 24 March 1997.
- 34. The final report was drafted in end-March at UNIDO Headquarters, Vienna.
- 35. The mission team wishes to acknowledge the full support and co-operation received from government and private business officials met during its field mission as well as UNDP officials and UNIDO field offices.
- 36. The report has been organized in seven chapters. Chapter 2 covers the conceptual and policy framework for the Second IDDA. It deals with the background to the Second IDDA, objectives at the national, sub-regional and regional levels and the envisaged institutional framework for the implementation, monitoring and evaluation of the Programme. Chapter 3 reviews the process of industrialization in Africa in the context of the changing international environment. It provides an overview of the internal developments and changes in the continent of Africa as a response to instabilities and macro-economic imbalances. It also highlights new external developments taking place at the international level that have had profound effects on Africa's global economic relations.
- 37. Chapter 4 of the report provides an assessment of the implementation of the Programme of the Second IDDA. It examines the implementation of the Programme at the national and sub-regional levels covering sector priority areas, and attempts to provide an analysis on the performance of the Programme, results achieved and shortcomings observed.

38. Chapter 5 covers the range of constraints which affect Africa's industrialization and impeded implementation of the IDDA II Programme. Chapter 6 provides an analysis of the findings of the evaluation along the lines indicated in the terms of reference drawing from the assessments made at the national and regional levels. Chapter 7 makes recommendations for the effective implementation of remaining portion of the IDDA II Programme.

2. CONCEPTUAL AND POLICY FRAMEWORK FOR THE SECOND IDDA

A. Lagos Plan of Action

- 39. The Lagos Plan of Action, designed jointly by all member states of Africa, constitutes a charter for the development of the continent for the period 1980-2000. The Final Act of Lagos, adopted in 1980, generated a feeling of optimism and expectation that the last two decades of the century would, through the individual and collective efforts of countries, witness a major breakthrough in African economic and social development and establish a firm foundation for progress in the twenty-first century.
- 40. The underlying spirit of the Lagos Plan of Action was the vision of African self-reliance and self-sustainment, while committing the member states to the concept of regional and sub-regional cooperation.

B. The First IDDA

- 41. From the concepts and objectives of the Lagos Plan of Action emerged the idea of a decade specifically devoted to translate the goals of the Lagos Plan of Action into industrial programmes and projects. Proposals for an Industrial Development Decade for Africa (IDDA), 1980-1990 were adopted at the Sixth Conference of African Ministers of Industry, held at Addis Ababa, Ethiopia in November 1981.
- 42. The first IDDA which was an outcome of the Lagos Plan of Action and an integral part of it was then launched with the main objective of using industrialization as a means of attaining self reliance and self-sustainment.
- 43. However, the decade of the 1980s was disastrous for Africa. In terms of economic performance, the record was very disappointing, the industrialization of Africa did not become a reality. Africa by the end of the 1980s was even poorer than it was at the commencement of the Decade in terms of per capita income.
- 44. In accordance with the proposal of the Eighth Conference of African Ministers of Industry, held at Bujumbura, Burundi, in September 1986, for an independent evaluation of the IDDA I, an appraisal of the Decade Programme was made in 1988 by an independent team of experts. The report of the evaluation which detailed the weaknesses and limited results of the Programme, was endorsed by an Intergovernmental Group of Experts and finally by the Conference of African Ministers of Industry at its ninth meeting held in Harare, Zimbabwe from 29 May to 1 June 1989.

C. The Second IDDA

a. Legislative Decisions

The Eighth Conference of African Ministers of Industry, held at Bujumbura, Burundi, in September 1986, recommended the proclamation of a second IDDA, to accelerate the industrialization of Africa launched during the first one. The meeting also requested the Secretariats of ECA, OAU and UNIDO to undertake, in consultation with member states, African Organizations and other relevant United Nations agencies, the preparation of a programme for IDDA II.

- The Conference of African Ministers of Industry, at its Ninth Meeting held from 29 May to 1 June 1989, in Harare, Zimbabwe adopted resolution 2(IX) concerning the proclamation of a Second Industrial Development Decade for Africa and the formulation of a programme for the Decade.
- 47. The General Conference of UNIDO at its third session, adopted resolution GC.3/10 of 23 November 1989, on the proclamation of IDDA, and later the General Assembly, in resolution 44/237 of 22 December 1989, proclaimed the period 1991-2000 the Second Industrial Development Decade for Africa.
- 48. After the proclamation of the Decade, the General Assembly in resolution 47/177 of 22 December 1992, stressing the need to integrate the programme for the Second Decade into the overall framework of the United Nations New Agenda for the Development of Africa in the 1990s, decided to adjust the period for the Programme for the Second Decade established in its resolution 44/237, to cover the years 1993-2002, and adopted the Programme for the Second Industrial Development Decade for Africa including national, sub-regional and regional components of the Programme.

b. Goals

49. The basic goals of the Second IDDA are not fundamentally different from those adopted by the first decade. They spring from the concepts of the Lagos Plan of Action. The vision continues to be that of a programme to end the over dependency which African countries have on the industrialized world, to promote internal engines of growth, to build on Africa's wealth and natural resources and progressively to achieve self-reliance and self-sustainment.

c. Programme Components

- 50. Preparations of the national programme for the second IDDA were moved to the national level. Each member state had undertaken the task of framing a national programme for the second IDDA using the services of local experts. Building up from the nationally felt problems should ensure a sense of pragmatism. To assist the member states in the preparation of national programmes, the three Secretariats of OAU, ECA and UNIDO issued a framework and guidelines for such work. The framework emphasized that the basic objectives of the first IDDA remained valid, and categorized the proposed programmes and activities into few distinct groups.
- 51. Accordingly, the interim period (1991-1992) before the start of the Second Decade in 1993 was used to prepare national, sub-regional and regional programmes.
- 52. The sub-regional integrated programmes were adopted at sub-regional meetings, in which priority projects were identified as well as the sponsoring countries. Following the recommendation of the Eighth Meeting of African Ministers of Industry for the proclamation of a Second IDDA, the integrated sub-regional programmes were revised between 1988-1991. The *sub-regional components* of the Programme for IDDA II are, therefore, based on the revised programmes. Moreover, they have been discussed and updated with the relevant sub-regional organizations.
- 53. The regional projects that were selected for the *regional component* of the Programme of the first IDDA were planned for implementation during the second IDDA. The industrialization of the region was expected to contribute to the widening of sub-regional markets, to the establishment of production links between countries and to the creation of a common market and later of an African Economic

Community (AEC) whose treaty was signed on 3 June 1991 at Abuja (Nigeria) by the African Heads of State and Government.

d. Strategic Approach

- 54. While the First Decade Programme was designed and formulated at a centralized level by the OAU, ECA and UNIDO, and approved by the Conference of African Ministers of Industry, the Second Programme's preparation took place at the national and sub-regional levels taking into account the realities, environmental circumstances, natural resources and priorities of each member state. Secondly because of the experiences of the first decade, the changing world environment and the adoption by a large number of African countries of the World Bank sponsored Structural Adjustment Programmes there has been a perceptible change in the strategic approach to achieve the goals of IDDA II. The modified strategic approach includes the following components:
- optimizing interlinkages between industry and agriculture;
- concern at the poor performance of existing industrial investment plans for rehabilitation, and need to improve public enterprise performance;
- strengthening the African entrepreneurship base;
- the importance of the informal and small-scale sectors;
- a new emphasis on market orientation;
- stimulating an industrial culture;
- promoting regional and sub-regional co-operation;
- environmental considerations.
- 55. The action-oriented programme for the Second IDDA encompasses:
 - a) Consolidation Programmes
 - b) Industrial Expansion
 - c) Promotion of the Small-Scale and Medium Sector and entrepreneurship Development and
 - d) Support Services

e. Policy Approach

56. While all the member states have continued the practice of preparing national development plans, there has been a major attitudinal change in approach. There is a strong move towards market orientation and the promotion of competitive forces. There is a clear trend towards opening up and liberalization of the economies and the promotion of African entrepreneurship. At the same time, there is a trend towards reducing investments of the public sector in industry.

f. Envisaged Implementation, Coordination and Monitoring Mechanisms

57. The Programme of the Second IDDA put a greater emphasis on the mobilization of resources for the promotion of investment in national, sub-regional and regional multinational projects. The Programme categorized the financial needs in five areas and noted that it would be difficult for Africa to depend solely on its resources especially in view of heavy debt burden, the budgetary deficits and the weakness of domestic savings in almost all the African countries. The Programme thus assumed that foreign assistance and foreign investment would be necessary to finance its implementation.

- 58. With respect to the institutional mechanism for the follow-up on the implementation of the Programme at *the national level*, it was proposed to set up a National Coordinating Committee with the broadest base possible, including all relevant economic ministries and supporting infrastructure. At the highest political level, the appointment of a Cabinet Committee had been proposed with the ministers of industry, finance, planning and agriculture as members, to provide overall coordination, examine progress reports and authorize any necessary adjustments in the approved Programme.
- 59. In order to monitor progress in the implementation of the IDDA Programme at the sub-regional level, the Programme proposed that an Industrial Promotion Coordinating Committee be established within the framework of existing organs such as the MULPOCs and IGOs whose function would be to coordinate the formulation and implementation of all industrial programmes in the sub-region and report periodically to the legislative body of the sub-region. Such a body could be supported by a number of Sub-sectoral Strategic Management Committees in such sub-sectors as metallurgical, engineering, agrochemical industries, basic chemicals and pharmaceuticals, building materials industries, etc.
- 60. At the regional level, a Committee of Ten composed of ten member countries selected in accordance with a balanced geographical distribution was set up to monitor the progress in the implementation of the Second IDDA Programme. Due to financial constraints and lack of follow-up, the Committee was not able to meet regularly and its contribution was ineffective. The 12th meeting of the Conference of African Ministers of Industry dissolved the Committee of Ten in 1995. The Bureau of CAMI assisted by the Joint Secretariat and national coordinating committee was given the mandate to monitor the progress of the implementation of IDDA Programme and report to subsequent meetings of CAMI.

Conclusions

- 61. The goals and objectives of the Programme were relevant and compatible with the prevailing conditions in Africa.
- Though the proposal for establishing an implementation and monitoring mechanism at the national, regional and sub-regional levels was envisaged, in practical terms it was almost non-existent or did not function;
- despite that clear knowledge prevailed among member countries that external financial resources would be required to support implementation of the Programme, no such strategy was incorporated in the Programme;
- the Programme did not consider any risk factors both internal and external which could hinder/derail Programme implementation and subsequently no alternative measures were designed to mitigate those risks;
- the Programme existed almost as a desire rather than an action-oriented programme backed by adequate financial resources with effective implementation teams;
- large bilateral and multilateral programmes were under implementation alongside the IDDA Programme. This contributed to the IDDA Programme being submerged and marginalized;
- while global changes occurred, little programme adjustments were made to allow entry of new development partners into the process of a dialogue for the design and implementation of the Programme.

3. BACKGROUND TO AFRICA'S INDUSTRIALIZATION IN THE 1990s

A. Internal Factors

a. Socio-Political Conditions

62. In response to continued political and social instabilities as well as macroeconomic imbalances, the overwhelming majority of African countries are in the processes of implementing political and economic reforms. In the political arena, the democratization process underway in the majority of African countries has led to a transition from long-established one-party rule to multi-party democratic governance. This process lends its support to national consensus building in policy formulation and decision making.

b. Economic Transformation and Adjustment

63. In the economic sphere, reform measures focusing on fiscal austerity and restrictive monetary policies, and the liberalization of foreign exchange and interest rates are being carried out. Most of the reform measures are aimed specifically at liberalizing factor and product markets, removing distortions, empowering the private sector, and making the public sector more efficient, in spite of the enormous short-term costs and the social consequences of such measures. Demand management in the form of budgetary reforms and restrictions on credit and monetary expansion constituted one of the widely used policy instruments. Containment of the growth of public expenditure and the application of austerity measures and civil service reforms aimed at the rationalization of the structure of the public sector were pursued with vigour in many countries of the region, with the brunt adjustment continuing to fall on necessary investment in vital services and infrastructure and limitations on budgetary subsidies. Privatization and commercialization of poorly run public enterprises was contemplated also in some countries, although concrete achievements and outcomes fell far short of plans in many cases. In general, the main focus has been the creation of a stable macroeconomic environment for growth. However, most countries are yet to make the necessary transition from overwhelming preoccupation with adjustment and stabilization to the long term concerns with sustainable development and structural transformation.

c. New Initiatives on African Development

64. Since the adoption of IDDA II, a number of changes and developments have taken place at the international level, and within the continent of Africa as well. Within this period African countries have also adopted various economic initiatives of which the following are the most important.

i. The Treaty Establishing the African Economic Community

65. The treaty establishing the African Economic Community, signed in Abuja, Nigeria on 3 June 1991, having as a major objective the promotion of economic, social and cultural development in Africa is a significant step towards economic and industrial integration in Africa. The Abuja Treaty which came into force in 1993 is yet the most comprehensive expression of Africa's political commitment to build a viable, continent-wide economic community; it places great emphasis on the role of integration in promoting economic self-reliance and an endogenous and self-sustained development. The Treaty assigns top priority to the development of regional production structures, with appropriate supportive infrastructure, and the coordination and harmonization of economic and social policies within and between the sub-regional communities. This is expected to serve as a vehicle for the rapid expansion of intra-African trade, and for enhancing Africa's effective participation in the evolving global linkages and interdependence of production enterprises.

66. In the context of the Abuja Treaty, therefore, the purpose of promoting industrial development is, among others, to harmonize the industrial policies of the member states, create a solid basis for industrialization, promote collective self-reliance, modernize the priority sectors and establish joint industrial development projects at regional level as well as to create African multinational enterprises in priority industrial sub-sectors.

ii. The Cairo Agenda for Action

- 67. The Cairo Agenda for relaunching Africa's economic and social development adopted by African heads of State and Government in Addis Ababa, Ethiopia, in June 1995, is a significant step forward towards economic and industrial integration in Africa.
- 68. The Cairo Agenda identified a set of priority issues which governments were committed to address and for which international support was requested in order to promote growth and economic recovery. These issues were: (i) Governance, peace stability and development; (ii) Food security; (iii) Human resources development and capacity building; (iv) Resource mobilization; and (v) Regional economic co-operation and integration.
- 69. In this regard, industrial development is centred to structural change and transformation of Africa's economies, to the increase in incomes and employment and to the diversification of exports.

iii. The UN System-wide Plan of Action for African Economic Recovery and Development

- 70. The objectives of the System-wide Plan of Action for African Economic Recovery and Development are to provide a dynamic and flexible framework for concerted and coordinated action by the United Nations system to assist Africa in achieving sustained and sustainable growth and development in the 1990s and beyond, assist in the creation of a more supportive international economic environment for economic growth and social development, assist in the restructuring of African economies, help promote employment generation, channel the sectoral expertise of the United Nations organs, agencies and bodies in efforts to foster African recovery and development, and enhance global awareness of the continent's economic crisis.
- Based on the latest assessment of Africa's socio-economic situation and the continent's critical issues and challenges, and taking into account the directives of intergovernmental bodies, the revised System-wide Plan which was adopted in 1994, identifies key priority areas in which common strategies and shared objectives could be pursued through cooperative or joint action by several organs, organizations and bodies of the United Nations system, and also draws a closer linkage between the System-wide Plan and the United Nations New Agenda for the Development of Africa in the 1990s by reflecting in the System-wide Plan the new imperatives that have emerged from the review of the New Agenda. In recognition of the dynamic and diversified role of women as development actors at all levels, the revised Plan incorporates the role of women in key areas such as agricultural production, human resource development and capacity-building, and deals with the mainstreaming of women in the development process.

iv. Alliance for Africa's Industrialization

72. The Alliance for Africa's Industrialization which was launched by the African Heads of State and Government in Abidjan on 23 October 1996, is positioned to boost the implementation of IDDA II in light of meeting the new challenges emerging from the Uruguay Round agreements and the increasing globalization of the world economy.

- 73. The Alliance seeks to provide a framework under which African countries can formulate competitive national industrial development strategies based on private sector-led development, focusing on comparative advantage, and then see these translated into concrete, practical action programmes.
- 74. The Alliance is also complementary to the United Nations System-wide Special Initiative, synthesizing the industrial aspects of the Special Initiative into a coherent programme, incorporating key elements such as a private sector-led solution, promotion of investment, reduction of poverty and increased food security through agro-based industries.

B. External Factors

a. Changing world Economy

- *75*. The early years of the 1990s have witnessed major changes in the international environment that are likely to have profound effects on Africa's industrial development and international economic relations in the coming years and beyond. The developed countries appeared to have emerged from a protracted recession that had previously put so much strain on national economies and caused a good deal of disruption in the international economy. Further, the globalization of world production has gathered new strength, pushing further to the periphery of the world economy those countries that have not developed the capacity to compete strongly in the world market or participate effectively in the new global economic order which is characterized by interdependent production structures cutting across national boundaries. In response to growing world economic recovery, increasing privatization of public enterprises, emerging financial markets, and a wide range of incentives provided to private investment, there has been a massive surge in private capital flows, which would mainly benefit counties with attractive investment environment and markets. African countries have not benefited as other developing countries have. The present strengthening of commodity prices has been beneficial to many producing countries, although there are fears that this has only been a temporary deviation from the long term declining trend.
- 76. These and other developments are evidently fraught with innumerable challenges and concerns with great opportunities of far-reaching consequences that African countries must effectively manage if they are to mobilize fully their resources for development, and reverse the present tendency of marginalization in the world economy. This is not, however, an easy task knowing the serious obstacles that African countries face in their drive to meet the challenges of domestic development, including the reorganization and transformation of their economic structures.

i. Uruguay Round/WTO

77. World trade is more and more globalized, reinforced by the fact that it increased more rapidly than world production, through the creation of the World Trade Organization and conclusion of the Uruguay Round. This liberalization of world trade will be an opportunity for African states in as much as they carry out reforms intended to transform their competitive base.

ii. Regional Blocs

78. World trade liberalization process is a result of the appearance and reinforcement of regional economics blocs (European Union, ASEAN, NAFTA, MERCOSUR, AEC etc.). The final objectives of these regional arrangements are economic integration, improved economies of scale and opportunities for specialisation with optimal allocations of regional resources. As far as Africa is concerned, the

implementation of Abuja Treaty establishing the African Economic Community should prepare the region for the global economy.

iii. <u>Technological Changes and Competitiveness</u>

79. Specially in today's globalized economy, the potential for economic prosperity depends on technological, organizational and human capital innovations that enhance the productivity of any factor of production. The emerging new industrial competitiveness increasingly depends on technological capabilities and innovation, and on the ability to apply new technologies in production, organization and marketing, and to establish appropriate linkages with global corporations in the form of subcontracting, production mandates, FDI, licensing, strategic alliances, etc. In such a scenario, international competitiveness in a large number of industries will be determined by the ability of firms to apply, at least selectively, new generic technologies in their production and organization. In this regard, Africa must face the challenge by defining technology policies that would enable it to enhance industrial capabilities and hence industrial competitiveness.

iv. Environmental Concerns

80. The new approach on environmental pressures is another challenge that faces African industry at the international level. This new approach has emerged as a result of stringent environmental regulations on pollution prevention and control. The approach allows industry to conserve resources and reduce pollution during the whole production process rather than only at the end. These environmental pressures urge governments to set up policies that ensure ecologically sustainable industrial development especially those that promote cleaner production technologies and reduction of waste during the production process.

b. Investment Flows Assistance

i. Official Development Assistance

While it is estimated that Africa's annual external financial requirements for development come to about US\$ 60 billion, total net resource flows amount to only one third, i.e. US\$ 21,7 billion in 1993 down from US\$ 22.9 billion in 1990. The major part of these flows is made up of multilateral and bilateral loans. Grants also declined from US\$ 21.5 billion in 1990 to US\$ 19.7 billion in 1993. This decline is due mainly to budgetary constraints in many of the donor countries, the growing demand from other competing areas such as Eastern Europe, and the reorientation of assistance policy towards aid administration and greater effectiveness in its utilization. The stagnation or slowing down of official development assistance comes at a time when many recipient countries in Africa are undergoing periods of economic stress caused by macro-economic instability, civil strife, deteriorating terms of trade, increases in population growth as well as heavy debt burden. While these constraints are being addressed by many African countries through such measures as structural adjustment programmes and political reforms, however, overseas development assistance will continue to be a major player in addressing the above problems.

ii. Foreign Direct Investment

82. The flow of FDI has not essentially favoured African low-income countries because of both external and internal conditions and low credit rating. Much of the flow is directed mostly to South-East Asian countries and Eastern Europe which are stronger competitors for direct foreign investments. Africa's ability to reverse this situation depends on the extent to which internal conditions could be improved. These conditions include political instability and uncertainty, increasing levels of external

debt, inadequate commitment to reform measures and the unattractive conditions of physical and institutional infrastructure. These internal disincentive factors are now being addressed through various international initiatives on Africa's recovery and development. The macro-economic reform programmes, attention on capacity building of African institutions, human resource development and the development and enhancement of the private sector taking place in many member countries are together expected to improve the general economic and political climate and stimulate investment flows to African countries.

iii. External Trade

83. The marginalisation of Africa in the global economy is reflected in her share in world merchandise trade which has been declining over the past decade, and now accounts for only 2.5%. In 1994, total African exports amounted to US\$ 89.6 billion while its imports rose to US\$ 97.8 billion, thereby worsening the current account deficit which has been deteriorating from US\$ 6.2 billion in 1992 to US\$ 10.5 billion in 1994. On the whole about 30% of Africa's export earnings goes into debt servicing, worsening further the capacity to import the necessary machinery, spare parts and other inputs for development, apart from food requirements. It is also expected that the Uruguay Round Agreements will have some adverse impact on many African countries in the initial stages trying to adapt to the new market access conditions. These conditions are said to have eroded some of the preferences that were enjoyed under the Generalized System of Preferences (GSP) and the Lomé Convention, and in particular the cases of tropical products and natural resource based products. Because of the narrow export base, the erosion of preferential margins ranging from 50% to 100% will have a great impact on the overall export earnings of the majority of African countries even though some gains will be registered in improved access in other categories of products.

iv. Debt Burden

84. Many African countries had little or no debt at the time of achieving their political independence. Today, almost all of them are heavily burdened by international indebtedness. Causes for this indebtedness vary from country to country, but on the whole Africa's ability to finance its development programmes has been partly hampered in various ways by debt obligations. Africa's total debt stock was projected to increase from US\$ 297 billion in 1993 to US\$ 313 billion in 1994. These debts originate from the three major sources of external loans namely, bilateral (40.6%), multilateral (21%) and commercial/private (19.5%). International efforts have been made and are still continuing to write off, reduce or swap these debts. Whatever successes are achieved, Africa's international indebtedness will continue to be a constraint in the process of economic and social development.

Conclusions

a. Internal

- 85. Internal conclusions are as follows:
- Africa must provide political and socio-economic stability to attract investments;
- expansion of intra-African trade is a pre-requisite for enhancing Africa's effective participation in the evolving global linkages and in inter-dependence of production enterprise;
- it is necessary to create the necessary environment to attract multinational enterprises;

- industrial policies should be harmonized to facilitate a strong functioning regional economic community;
- no matter how many resolutions and action plans may be adopted, they shall however remain as wishes unless coordination is effected and the barriers lifted to promote inter-African economic activities.

b. External

- 86. Africa urgently needs to meet the challenges of domestic development and a reorganization and transformation of its economic sectors:
- reforms must be carried out to meet the requirements of Uruguay Round which is liberalization of trade to facilitate competitiveness both within and outside the region;
- participation in the regional blocs should be strengthened to promote economic integration with optimal allocation of regional resources;
- in the face of technological changes and competitiveness, Africa must face the challenge by defining technology policies that would enable it to enhance its industrial capabilities;
- the concept of sustainable human development is one of the key issues which African countries must gear themselves to with ecologically sustainable industrial development in force such issues must be addressed in future policy and decision making. Countries visited are yet to come up with any significant environment policy;
- development assistance should emphasise its intervention on capacity building of human resources in African institutions which in turn could provide the multiplier effect.

4. ASSESSMENT OF THE PROGRAMME IMPLEMENTATION OF IDDA II

- 87. It is recalled that the purpose of the evaluation exercise is to enable African member states, UNIDO, ECA and OAU to make an objective assessment of the Programme against set objectives, and to make decision on future course of action.
- 88. In pursuit of this goal the evaluation attempted to review the IDDA II Programme implementation within the overall objectives as outlined in the national and sub-regional programmes. The evaluation methodology has been to examine the IDDA Programme goals as stipulated, the mechanism for implementation and to assess implementation performance in contributing towards industrialization of the continent. In assessing Programme achievements, the evaluation sought to identify various factors that have facilitated these achievements as well as those factors that have impeded the fulfilment of the Programme objectives with a view to proposing a course of action both at policy and implementation levels for future revision/design, implementation and monitoring of the Programme.
- 89. The assessment was constrained by the following factors:
 - i. the limited amount of time allotted to field missions and the lack of perception of the IDDA Programme at the national level did not facilitate the collection of sufficient data and information;
 - ii. high turnover of staff both at ministerial/government bodies and plant levels did not help the mission team to gather information and data within the 2-3 days provided for each country visit;
 - iii. other bilateral/multilateral programmes which were overlapping with IDDA Programme activities.
- 90. The general observation is that the IDDA II Programme implementation as originally envisaged has so far not materialized, and consequently has not been able to generate the expected impact. However, this is not to say that some positive changes have not occurred. The evaluation team strongly believes that today African policy and decision makers are more aware and committed in terms of what should be done to achieve and sustain a progressive development path than in the past.

A. Mechanism for Implementation, Monitoring and Coordination of the IDDA II Programme

i. At National Level

- 91. Learning from the lessons of the last decade, the programmes for the Second IDDA were prepared at the national level, while member states themselves undertook the responsibility for ensuring Programme implementation and coordination. The mechanism proposed for this purpose and to publicize the national programmes was:
 - a national agro-industries commission;
 - a national commission to revitalize public sector enterprises;
 - a specialized department for rehabilitation programmes, preventive maintenance and manufacture of spare parts:
 - an industrial manpower task force; and
 - an inter-disciplinary team to upgrade physical infrastructure.

- 92. The purpose of these bodies was to build teams out of existing organs for better coordination, inter-disciplinary understanding, adjustment and promotion of an integrated programme approach, and secure an effective implementation mechanism in their respective fields.
- 93. The evaluation team noted that none of these mechanisms proposed to publicize and coordinate the national programmes was established. This was felt necessary to avoid multiplicity of committees and agencies that would overtax the limited human resource capacity within the governmental system. Existing national machinery, i.e. the Ministries of Trade and Industry or Planning were expected to perform the implementation and coordination functions of the Programme as part of their normal activities, but even that did not work.
- 94. Apart from the failure to establish special mechanisms for Programme implementation, many countries do not maintain national records of industrial statistics and data that would facilitate assessment of industrial performance within the economy and impact of the Programme. This added to the difficulty of making any reasonable appraisal of the status of industry and its contribution to overall economic growth.
- Despite the absence of centralized mechanisms for the implementation of the overall national IDDA II programme, specialized agencies exist in many countries for the promotion of specific segments of the Programme. For example, a number of promotion agencies have been established to mobilize local and foreign financial and technological resources through investment promotion activities and services. Development banks and financial markets are already operating in some countries for domestic savings mobilization and channelling of multilateral and bilateral funds into specific sectors of the economy. In the area of SMI and entrepreneurship development, there are special institutions like EMPRETEC which are also engaged in the entrepreneurship development and although privatization was not included in the initial IDDA II Programme, its subsequent adoption has led to the establishment of implementation agencies to oversee government divestiture and privatization programmes, the success of which will broaden the base and scope of private sector operations in the economy.

ii. At sub-regional Level

- 96. IDDA II sub-regional programmes were prepared by the Joint Secretariat which took into account the existing programmes of sub-regional economic groupings. At different sub-regional meetings convened by ECA, UNIDO and relevant sub-regional economic groupings, the programmes were revised as priority projects and host countries were identified. Major responsibility for implementing sub-regional projects was left to individual governments and their private sectors with the support of sub-regional groupings, ECA, UNIDO, OAU and development partners. Since 1995, the Bureau of CAMI has replaced the Committee of Ten to monitor and follow up on Programme implementation.
- 97. Coordination of sub-regional programmes was to be pursued by the Sub-regional Economic Committees through follow-up meetings with the assistance of the Joint Secretariat.
- 98. A Resource Mobilization Committee (RMC) comprising of relevant sub-regional economic groupings and financial institutions, ECA, UNIDO and the World Bank to be chaired by ADB was to be set up as the main mechanism for coordinating the mobilization of resources for IDDA II implementation at sub-regional level. Somehow the RMC was never constituted, leaving financial resource mobilization at sub-regional level at bay and related technical assistance programmes un-coordinated and unfocused.

B. Sectoral Assessment of the Implementation of the IDDA II under National Programmes

i. <u>Consolidation Programmes</u>

- 99. This consisted of the rehabilitation of existing industries and the improvement of performance of public sector enterprises. Poor performance of most African industries are traced to a number of factors, the most prominent being underutilization of capacity. The evaluation of IDDA I came out forcefully on the issue and recommended that IDDA II place strong emphasis on the rehabilitation and regeneration of existing industries and to ensure that these national assets are not wasted away.
- *100*. Rehabilitation and performance improvement of existing industries: One of the strongest points in almost all national IDDA II programmes is the highest priority given to industrial rehabilitation utilization of enterprises. Causes of poor performance have been identified and remedial measures are already underway in many countries. Ethiopia has registered some successes in its rehabilitation efforts and consequently raised capacity utilization to an average level of 75% to 80%. Some of the rehabilitated industries have now been privatized. In Kenya, a number of non-strategic parastatals have successfully been rehabilitated and privatized. East African Portland Cement Factory, Kenya Meat Commission and a number of sugar and tea factories are being rehabilitated, some of which have been privatized. In Ghana, the UNIDO/ILO Industrial Sector Programming Review Mission undertook an extensive analysis of the industrial sector including the IDDA II Programme and synthesized the focus of industrial development into key areas and proposed a series of programmes likely to attract donor technical assistance for rehabilitation. The Review Mission identified factors responsible for poor industrial performance and capacity underutilization in both public and private sectors. Some of these factors were high dependence on imported inputs, small domestic markets, little or no export outlets, obsolete equipment and weak management. In tackling these weaknesses, as part of the rehabilitation programme, Ghana has established the Rehabilitation Advisory Services Agency (RASA) responsible for identifying rehabilitation needs and resource requirements as well as providing direct on-the-job assistance to selected enterprises.
- 101. In Mali, two textile industries (ITEMA and COMATEX) have been rehabilitated by 1994. In Niger, two enterprises (Nouvelle Imprimerie du Niger and La Société des Produits Alimentaires) have been rehabilitated out of twelve enterprises earmarked for rehabilitation. In Guinea, most of the enterprises originally identified for rehabilitation were later privatized for lack of funds for renovations.
- 102. Many other countries, particularly those implementing Structural Adjustment Programmes, have gone ahead to privatize industries without any attempt to first rehabilitate them, as in the case of Guinea, where the record of privatization has been disappointing.
- 103. Countries that have seriously tackled their rehabilitation programmes and also adopted measures for improving performance of public sector enterprises have been more successful at their privatization exercise.

ii. Industrial Expansion Programmes

104. In addition to improving the quality and performance of existing industries, national IDDA II programmes placed emphasis on expanding the industrial base. The number of industrial enterprises in most African countries is so small that no critical mass of industries could exist that will grow into self-sustaining complex of supportive industries and services in different branches. Industrialization therefore can hardly take root where needed enterprises are not being created and/or existing ones being expanded.

105. Industrial sectors identified under the IDDA II Programme for expansion have however received mixed responses in different countries where a large number of state-owned enterprises exist in ailing conditions with little or no funds for rehabilitation and improvements. Governments are sceptical about expansion programmes. The approach adopted by some countries is to proceed with rehabilitation where possible, and then privatize those which are not of strategic importance while encouraging the indigenous entrepreneurs to move into new industrial sectors or expand their existing small and medium-size enterprises.

a. Metallurgical Sector

- 106. This is one of the core industries identified in many national programmes. Activities within this sector include:
 - rehabilitation, modernization and diversification of existing integrated iron and steel plants;
 - capacity expansion of existing integrated steel plants and steel rolling mills;
 - upgrading of existing smelters for copper and aluminium and establishment of new smelters:
 - establishment of a Metallurgical Technology Centre for the Preferential Trade Area for Eastern and Southern African States;
 - establishment of an Aluminium R&D Centre.
- 107. Due to the heavy capital outlay involved in the development of projects within this sector, only a few African countries have been able to move into this area of industry. Integrated iron and steel complexes have been established in only six countries and utilize about 50% of capacity on coverage. Twenty-two countries are already operating about 30 mini-steel plants and forty-five rolling mills.
- 108. Facilities for producing quality iron and steel castings are found in four countries (Algeria, Egypt, Kenya and Zimbabwe). It is estimated that only 40% of installed capacity of Africa's foundries totalling 500,000 tons per annum is utilized. On the whole the development of viable engineering industries is being hampered by the discouraging condition of the steel and metal-working enterprises in all sub-regions of the continent.
- 109. Africa's potential for the development of the metallurgical sector is considerable, but lack of financial resources and technological knowhow required for rehabilitation of existing plants and expansion where appropriate will continue to constrain the development of the sector in the near future.

b. Engineering and Allied Metalworking Sector

- 110. Areas within this sector earmarked in many national programmes are:
 - manufacture of tractors and agricultural implements;
 - production of pumps; and
 - manufacture of household appliances.
- 111. While this sector commands an important place in the industrialization process of African countries, its development has also been restrained by lack of financial resources. Existing plants and installations are yet to be rehabilitated in order to enhance capacity utilization and satisfy the expanding domestic needs especially in the agricultural sector. The Akaki Spare Parts and Hand Tools Factory of Ethiopia offers a good example of a huge plant with capacity to satisfy some of the requirements of a number of African countries in terms of agricultural implements and various spare parts. Yet lack of

financial resources for rehabilitation and market expansion programmes has kept this relatively new plant to a production level below breaking point. Production costs are high in this and similar plants, with the result that most countries meet their domestic requirements through imports from non-African countries.

112. Zimbabwe's production of non-electrical machinery (machine tools, guiding machines and sanders) could be expanded to increase its exports beyond PTA countries. Both examples point to the importance of sub-regional co-operation in the promotion of large-scale manufacturing plants to take advantage of bigger markets and economies of scale.

c. Chemical Sector

- 113. This sector comprises production in industrial gases, acids, plastics, soap and detergents, cement, pharmaceuticals, fertilizers, pesticides, etc. Apart from a few countries which have the markets and are capable of establishing basic and intermediate level chemical plants, all others may have to content themselves with formulation, blending and packaging activities.
- 114. Pharmaceutical, cement and fertilizer production has been the concern of many member countries some of which have already gone far in this direction. In Nigeria which is endowed with large deposits of phosphate rocks and cheap gas, a number of fertilizer plants have been established. The development of NAFCON II (National Fertilizer Company of Nigeria) in Onne, Port Harcourt and NAFCON III in Abuja has considerably reduced fertilizer importation which stood at about US\$ 350 million annually. In Kenya, there are already three cement producing plants namely: Bamburi Portland Cement Company Ltd., Caloleni Lime and Cement Works Ltd., and East African Portland Cement Company Ltd. The last one is already in the process of converting the plant from wet to dry process and to expand production from the installed capacity of 350,000 tons to eventually 700,000 tons. With annual consumption increase of about 7%, Kenya plans to establish two additional plants with combined capacity of nearly 1 million tons in the near future.
- 115. In the field of pharmaceuticals, Kenya ranks high among COMESA countries. Out of 50 recognized pharmaceutical manufacturers in the COMESA sub-region, 24 are located in Kenya. Capacity utilization of these establishments is around 40% and about 95% of raw material inputs are imported.
- 116. With its endowment of medicinal and aromatic plants, Kenya plans to develop further its pharmaceutical capability using this natural resource base. The Kenya Medical Research Institute has been carrying out research into plant extracts, and commercial operations in traditional medicines and extraction of active ingredients offer new investment opportunities.

d. Agro-Industries and Food Production

- 117. All national programmes gave high priority to agro-industrial development concentrating essentially on:
 - edible oil mills;
 - fruit and meat preservation;
 - fish processing;
 - milk, sugar, coffee and salt production.
- 118. The importance of this sector was noticed in all the countries visited during evaluation missions. In francophone Africa, agro-industries account for more than 40% of total number of industries. In Mali, for example 38% of MVA is derived from agro-industry and similar observations were made in Niger, Guinea and Senegal.

- 119. In all these countries, the beverages sub-sector has been the fastest growing in the manufacturing sector. Uganda has two major beer production plants, the Nile Breweries and Uganda Breweries with combined industry capacity of 700,000 crates of beer per month. Expansion programmes are already underway including a new bottling assembly line.
- In most countries food processing sub-sector accounts for over 60% of the manufacturing value 120. added. The production at cottage or household level results in large wastage and post-harvest losses amounting in some cases to about 40%. The problem of post-harvest losses is being tackled by the construction of silos, storage facilities and cold stores as well as farm-gate purchases (fruits and vegetables) by government-supported marketing organizations. The processing of cocoa beans, for example, constitutes a small fraction of total production for which Africa accounts for 56% of world production and 60% of world exports. Processing plants exist in Ghana and Côte d'Ivoire, the latter with the largest grinding capacity in Africa ranking sixth in the world. Ghana's cocoa processing plants have been rehabilitated and expanded in the hope of increasing not only the production of cocoa butter and other intermediaries but also the manufacture of chocolates. Coffee processing faces a similar situation. Only about 100,000 tons of coffee extracts including instant coffee is locally processed out of Africa's annual green coffee production of about 1.25 million tons. Sugar and milk production, salt manufacturing and fish processing are sub-sectors where many countries pay serious attention. Uganda operates three major enterprises: Sugar Corporation of Uganda, Kakira Sugar Works and Kinyala Sugar Works Ltd., all of which have recently undergone major rehabilitation and expansion programmes.
- 121. Fish processing has been of major importance to a country like Namibia where canning, preservation and processing account for 26% of total manufacturing output, 30% of value added and 41% of employment in the manufacturing sector. Government policy is to keep emphasis on growth of local processing facilities with no more licenses for off shore processing.

e. Forest-based Industries

- 122. Under national IDDA II programmes many countries have identified need for investments and expansion in the following industrial activities:
 - manufacture of doors and windows;
 - pulp and paper;
 - plywood, flush doors, multi-ply panels;
 - fibre board.
- 123. In order to increase value added in the exploitation of their natural resources, some countries like Ghana have taken steps to ban the export of logs in favour of locally processed timber. Finished wood products such as wood-based panels and flush doors and furniture have great potential in many forest areas of Africa both for local and export markets.
- 124. Over 5 million tons of paper and packaging materials are expected to be imported yearly by the year 2000 even though many African countries have the potential to develop pulp and paper industries that would close the import gap. Only a dozen sub-Saharan countries including Kenya, Swaziland and Zimbabwe presently produce an average of 6,000 tons of paper annually. In view of the fact that pulp and paper industries require heavy investments and modern technology to be competitive, the problem and concern of desertification and because many African countries are timber-deficit, there is need for a few large-scale manufacturing plants on sub-regional joint-venture basis in partnership with external sources for finance and technology.

f. Leather and Leather Products

- 125. The continued development of this sector of industry is vital to many African countries especially those endowed with natural conditions for livestock development. These countries obtain over 5% of total export earnings from this industrial sub-sector. With improved infrastructure, improved methods of animal husbandry, increased animal population and meat consumption, Africa now accounts for about 18% of world livestock population. Despite the potential for the growth of this industrial sub-sector in world trade, not enough has been done because of the lack of technological knowhow, and an appropriate policy framework that will upgrade and expand this industry.
- 126. The leather industry has seen a steady growth in African over the past few years. With increases in hides and skins, the production of all categories of leather in African countries has been advancing at a faster rate than global averages. In Kenya, livestock production constitutes 26% of total income from the agricultural sector. On annual basis, livestock provides the tanning industry with 1.6 million raw hides and 4.3 million raw skins for processing into semi-finished and finished leather. The drawback has been the low capacity of operation averaging about 30% to 40% of installed capacity due mainly to obsolete technology and liberalization policy which has flooded the local market with cheap secondhand goods essentially footwear. Further assistance is required for rehabilitation and upgrading of institutions such as AHITI and KITI, to strengthen their capacity to train trainers and leather goods manufacturers. Kenya also produces 200,000 metric tons of fish annually from which 12,000 tons of fish skins are derived. These skins which are dumped as waste, could be processed and brought into the mainstream of leather production.
- 127. The importance of the leather industry has been underscored by the establishment of the Leather and Leather Products Institute (LLPI) based in Addis Ababa with units in some of the countries covered by COMESA. LLPI promotes productivity and competitiveness within the leather sector by providing training, information and consultancy services.

g. <u>Textiles</u>

- 128. Textiles industry is one area where many African countries have a long-standing tradition. In the national programmes, many countries placed emphasis on reactivating activities in this sector, including:
 - rehabilitation of existing textile mills to achieve optimum production capacity;
 - new investments for expansion of sector production;
 - intensification of garment production;
 - promotion of exports in the garment industry.
- 129. Being a net importer of textiles and clothing, African countries are in a position to rapidly expand the industry taking advantage of sound raw material base of natural and man-made fibres. The major problems facing most of these countries are related to obsolete machinery and outmoded technology as well as lack of quality control and funds to initiate modernization. With liberalization in the foreign exchange market, many textile manufacturers are now able to acquire foreign exchange to place orders for replacement parts and other rehabilitation equipments. This same liberalization policy has led to a massive influx of imported textile products including used apparel which in some instances has caused the collapse of local manufacturing plants.

130. In some countries, cotton production has not caught up with industry demand, resulting in the import of raw material inputs, at a time when foreign exchange restrictions were still in force. Countries which are determined to ensure growth of their textile industry have already put in place policy measures for reorganizing existing cotton growing schemes and developing new ones, encouraging textile manufacturers to engage in raw material production (backward linkages), encouraging the establishment of viscose making units where appropriate, and providing mini-irrigation schemes for cotton growing areas.

h. Construction Materials

- 131. While construction is among the fastest growing industries in many African countries, the development of building materials for the industry is still lagging behind. The manufacture of cement, bricks, sheet glass tiles and other roofing materials, ceramics, wood panelling, building components of iron and steel, aluminium and other metals, etc. are only partially reflected in national programmes. Some of these products are firmly in the hands of private sector operators who have responded to the increasing demand resulting from the expansion of the construction industry. The gap is filled by imports of items which can very well be produced locally.
- 132. In most countries, governments have not come out with concrete policies and measures that will stimulate the construction material industry and thereby bring supply in line with the demand of the expanding construction industry. For example, the exploitation of clay deposits for the manufacture of roofing tiles and other ceramic products has not been strongly encouraged. Since most governments are the largest users of such products (for construction of schools, hospitals, offices and other public buildings, etc.), it is expected that government policy will direct producers into the sector through incentive measures or direct government intervention where appropriate.
- 133. It has been observed that large amounts of foreign exchange are spent in the construction industry not only for the imports of materials, but for the payment of contract fees made to expatriate construction companies. The existence of local contractors should be recognized by awarding them construction contracts where their capabilities could permit. Their capacity building should also be the concern of African governments through assistance in training and special seminars, upgrading of skills and the award of joint-venture contracts for local and expatriate firms working together on the same project. Capacity building of African construction companies is strongly recommended, while efforts are being made to enhance the production of materials for the construction industry.

Conclusions

- 134. <u>Consolidation Programmes</u>: Countries that have seriously tackled their rehabilitation programmes and also adopted measures for improving performance of public sector enterprises have been more successful at their privatization exercises.
- 135. <u>Industrial Expansion Programmes</u>: Often industrial sector is so small that no critical mass can exist and grow. Industrialization can hardly take root where new enterprises are not being created and/or existing ones are not being expanded.
- 136. Metallurgical Sector: Africa's potential for the development of the metallurgical sector is considerable but lack of financial resources and technological knowhow required for rehabilitation of existing plants and expansion will continue to strain the development of the sector in the future. Presently on the average, only 40% of installed capacity is being utilized.

- 137. Engineering and Allied Metalworking Sector: Capacity utilization is low, production costs are high, most countries meet their domestic requirements through inputs from non-African countries. The need to promote intra-regional trade and sub-regional co-operation is essential.
- 138. <u>Leather and Leather Products</u>: There is lack of technological knowhow and appropriate policy framework that will upgrade and expand the industry. Assistance is required for upgrading of institutions such as AHITI and KITI to strengthen their capacity to train trainers and in leather goods manufacturing.
- 139. <u>Textiles</u>: Lack of quality control measures.
- 140. <u>Construction</u>: Capacity building of African construction companies is required to reduce use of foreign construction companies.
- 141. While agro-based industries are important, the linkage between agriculture and industry can only be strengthened and integrated provided the agriculture base is sound and consistent in terms of production and timely delivery. However, it is to be noted that agriculture in Africa is still dependent on nature's timely rainfall the agriculture system is yet to be developed along the lines of a modern irrigation system.

C. Assessment at Sub-regional/Regional Level

- 142. The IDDA II programmes at the sub-regional level comprised the drawing of sub-regional industrialization plans and strategies in the following priority sub-sectors:
 - Agro-industries
 - Metals and engineering
 - Chemicals
 - Building materials
- 143. There were differences in emphasis and order of priority in the different sub-regions. These plans included programmes on capacity building (human resources, energy, technology/standardization) and physical infrastructure.
- 144. Projects for promotion and implementation were identified by different sub-regions.
- 145. In Central African sub-region 37 projects were identified in the priority sub-sectors and 12 support projects. Apart from the fact that the projects were too many, they were not selected in a methodical manner using sub-regional selection criteria provided under the Programme. Furthermore, no concerted efforts were made at the country or sub-regional level to mobilize resources for implementing the projects. As a result, only a small number of projects were implemented. These projects have been implemented through assistance by UNIDO and UNDP. These are:
 - Assistance to the Central African Customs and Economic Union (UDEAC)
 - Assistance to the Economic Community of the Great Lakes Countries (CEPGL)
 - Production of Human Vaccines
- 146. The low implementation level of projects is partly due to weak sub-regional organisations and support institutions. ECCAs, UDEAC and BDEAC have almost collapsed or are being rendered

ineffective. Other constraints to programme and projects implementation are poor infrastructure, dearth of skilled manpower and general political and social instability in the sub-region.

- 147. Private sector contributions to sub-regional projects have been negligible because of the high risks involved and lack of appropriate sub-regional incentives.
- 148. A number of studies have been made and publications and technical reports issued by UNIDO and ECA including UNIDO programme review missions to Cameroon and Central African Republic and ECA pre-feasibility studies on three sub-regional projects. Various workshops, follow-up meetings and expert group meetings have been held for the sub-region. Two follow-up meetings on the implementation of the sub-regional programme for IDDA II were held in 1993 in Libreville and Brazzaville. And while a number of recommendations were adopted, they have so far not been followed-up.
- 149. In <u>Eastern and Southern Africa</u> some 43 priority and 45 optional projects were listed for implementation in the IDDA II sub-regional programmes.
- 150. Selected sub-regional programmes were included in some national development plans because their governments were given the lead role in the implementation of these programmes. Since these governments are now implementing SAPs, the lead role for implementing IDDA II programmes and industrial projects has shifted to the private sector and little advance in such programmes has taken place.
- 151. Mechanisms are being put in place to involve the private sector in policy formulation, implementation and monitoring at the national and sub-regional levels. At the sub-regional level, there is the Eastern and Southern Africa Business Organization (ESABO), an organization of national chambers of commerce and industry and manufacturing associations.
- 152. Within the private sector only one sub-regional core project (iron and steel production) has been implemented on joint venture basis between Zimbabwe and Kenya.
- 153. In the area of institutional and infrastructure development the Eastern and Southern Africa subregion has implemented a number of projects.
- 154. 1. COMESA has promoted sub-regional investments and trade through reduction of tariff and non-tariff barriers, information, workshops and technical reports. It has also promoted the establishment of a number of sub-regional institutions such as:
 - PTA Development Bank
 - PTA Clearing House
 - PTA Reinsurance Company (ZEP-RE)
 - Eastern and Southern Africa Business Organization (ESABO)
 - Leather and Leather Products Institute (LLPI)
 - Metallurgical Technology Centre (MTC)
- 155. 2. SADC has drawn up a sub-regional industrial strategy and is involved in the construction of multinational industrial and trade corridors.
- 156. 3. The EAC is reopening transport and communication systems in East Africa while various sub-regional and regional institutions are also active in promoting industrialisation. Foremost amongst them are:

- African Regional Organization for Standardization (ARSO)
- African Capacity Building Foundation (ACBF)
- African Institute for Higher Technical Training & Research (AIHTTR)
- East African Development Bank
- Eastern and Southern Africa Minerals Resources Centre
- Crossborder Initiative
- 157. While developments in institutional and infrastructure development are encouraging, the technical institutions are weak as a result of under capitalization and lack of linkages with industry.
- 158. In North Africa, the IDDA II sub-regional programme was directed at the implementation of sub-regional and multinational projects with the assistance of international and sub-regional institutions. The programme lists 37 core and 15 support projects.
- 159. Implementation of sub-regional core projects has not been possible because of governments' withdrawal from implementing industrial projects in favour of the private sector.
- 160. Sub-regional institutions which support industrialisation are:
 - Arab Industrial and Mining Organization (AIDMO)
 - Arab Regional Network for Agricultural Machinery (ARNAM)
 - Arab Union for Iron and Metal Industries
 - Islamic Development Bank
 - Arab Maghreb Union.
- 161. In <u>West Africa</u>, some 43 core projects and 17 support projects were identified in the IDDA II sub-regional programme.
- 162. These projects were expected to be implemented, coordinated and monitored by governments, sub-regional economic groupings and institutions including ECOWAS, CEAO, MRU, ARCEDEM, ARCT, Leather Research Institute, Niamey and MULPOC. The regional economic groupings have proved to be weak in implementation and coordination activities. Although ECOWAS revised its treaty in 1993, it has not managed to establish a customs union nor a common market. Decisions at the high official level are not translated into action. There are still a number of tariff and non-tariff barriers. Some financial institutions established by ECOWAS, such as the West African Monetary Agency, ECOWAS Fund for Co-operation and Development are still to make an impact on sub-regional trade and industrialisation. There are promising developments in UEMOA which has a single currency CFA Franc. The Mano River Union is weak and hampered by political and social instability in member states.
- 163. A number of meetings on implementation of IDDA II in West Africa have been held including a follow-up meeting (Abidjan 1993) and expert group meetings. The ECA has conducted studies on establishment of diverse plants such as iron and steel, and millet and sorghum. Both UNIDO and ECA have several reports for the promotion of industrialisation at the sub-regional level.
- 164. Specific sub-regional activities such as technical meetings, workshops, for and studies by UNIDO, ECA, OAU and others under IDDA have been effectively and efficiently implemented. There have however been serious weaknesses in follow-up and linkages with the productive sectors. Some studies have remained on the shelf. Investment for a follow-up activities have been non-existent. Technical support in transferring technology to private sector industries has been weak. Most of these

weaknesses stem from meagre financial resources applied and lack of effective coordination and cooperation among the stakeholders.

- 165. At the level of sub-regional intergovernmental organisations there have been weaknesses due to lack of resources, dependence on external resources, lack of political commitment and effective implementation mechanisms. There is also a lack of coordination with national IDDA implementation agencies in promoting sub-regional projects.
- 166. The private sector has not put in place mechanisms for implementation of sub-regional projects. Plans are under way to establish sub-regional capital markets and sub-regional private sector associations.
- 167. The projects identified in the IDDA Programme were not formulated using set criteria and were not focused. Resource mobilization mechanisms were not institutionalised. The limited resources available were not focused to implement core projects. The assessment of the overall Programme of IDDA is that although some projects were effectively and efficiently implemented, the Programme was not efficient or effective as reflected in the deindustrialization of Africa, weak economies and deteriorating incomes.
- 168. The Programme has however put into the fore the importance of industrialisation as the engine of growth and the importance of regional co-operation and integration in trade and industrialisation.
- 169. At the sub-regional level, the IDDA Programme impact can only be assessed by looking at sub-regional economic and industrial performance over the decade. Aggregate analysis of sub-regional macro-economic indicators however ignore wide variations between countries. These should be captured in the assessment of national programmes.
- 170. For instance, the output in <u>Central Africa</u> declined by 5.9% in 1993 and by 2.5% in 1994. In 1995 there was a modest growth of 1.1%. The sub-region accounts for only 8.7% of African GDP. The MVA growth rate has been declining since 1992.
- 171. The poor performance of the economy and the manufacturing sector is attributable to civil wars and political tensions in the sub-region. A number of manufacturing firms have closed down partly due to civil strife and liberalisation programmes. Capacity utilization is below 30 per cent in many enterprises.
- 172. <u>Eastern Africa</u> has recorded good GDP growth rates of 2.5%, 4.2% and 4.8% in 1993, 1994 and 1995 respectively compared to a decline of 1.3% in 1992. While agriculture contributed largely to overall output, the manufacturing sector also performed well. Manufacturing value added grew by 6.1% in 1994 and is estimated to have grown by 3.9% in 1995.
- 173. The Southern Africa sub-region recorded modest growth rates of 1.7%, 2.6% and 2.9% in 1993, 1994 and 1995 respectively. Manufacturing value added grew by 0.5%, 2.6% and 6.9% in the corresponding years.
- 174. The Eastern and Southern Africa sub-region is in the process of implementing Structural Adjustment Programmes. The entry of South Africa in the sub-regional economy has brought new opportunities and competition.
- 175. The institutionalisation of consultation mechanisms on industrial policies, strategies and implementation will be as crucial as the harmonisation and rationalisation of economic blocs.

- 176. North Africa accounts for over half of African GDP. The GDP growth rate was 1.1% in 1993 and 2.9% in 1994. The manufacturing sector share in GDP is about 15.5% and MVA growth rate was 1.5% in 1992, 1.9% in 1993 and 6.6% in 1994. Although capacity utilization in industries is estimated at an average of 50%, intra-sub-regional trade has been growing slowly due to the fact that the sub-region is oriented more and more towards Europe and the Middle East.
- 177. The West African sub-region's economy has been growing modestly over the decade. The GDP growth rate in 1993 was 1.4%, in 1994 it declined to 1.0% and was estimated to grow by 4.1% in 1995. Per capita GDP at 1990 prices was estimated at US\$ 362 in 1994. Estimates of 9.8% growth of MVA appear to be over optimistic. Utilization of installed capacities is estimated to average 30 per cent.
- 178. Economic performance in West Africa seems to reflect developments particularly in Nigeria where there has been a drop in oil production, political instability and a reversal of liberalization policies. In the CFA region a massive devaluation of the CFA Franc occurred at the beginning of 1994 with important consequences.

Conclusions

- 179. The low level of implementation of the sub-regional programmes is partly due to weak sub-regional economic organizations, poor infrastructure, dearth of skilled manpower and general political/social instability in the region.
- 180. Sub-regional technical institutions have been created however, these institutions are weak as a result of under capitalization, lack of linkages to industry and non-availability of financial resources.
- 181. IDDA sub-regional programmes and projects have been incorporated into national programmes and are shaped by SAPs which focus on stabilization, liberalization and privatization.
- 182. Most governments gave priority to the implementation of SAPs while the implementation of the IDDA II Programme was left to be led by the private sector.

D. Joint Secretariat Support to Implementation of the IDDA II Programme

i. UNIDO Support Programme

- 183. The UNIDO Programme in support of the Second IDDA is primarily directed towards:
 - the building up of viable production capacities in the form of entrepreneurship development, training, and small-scale industry development in such key industrial subsectors as agro-related industries;
 - assistance in building up competitiveness of African industry embracing industrial rehabilitation, technology upgrading and environmental protection;
 - assistance in regional and sub-regional co-operation.
- 184. Though the IDDA II Programme covers the period of 1993-2002, effective implementation of projects commenced in 1994.

- 185. UNIDO from its Regular Budget allocates funds every biennium specifically for the implementation of projects and programmes under the framework of IDDA. At the UNIDO Secretariat, the IDDA Programme is managed by a coordinator under the guidance and supervision of the Chief of the Africa Bureau. The financial allocation of the IDDA II Programme for the period 1994-1995 was US\$10,536,900 and for the period 1996-1997 is US\$10,600,000.
- 186. The UNIDO IDDA Support Programme comprises of the following types of services:
 - a. Technical Co-operation Activities (XA)
 - b. Supplementary Activities (YA)
 - c. Short-Term Technical Advisory Services (STAS)
- 187. a. <u>Technical Co-operation Programme (XA)</u> are projects carried out to support industries that establish linkages between industry and agriculture with emphasis on food production and processing, between industry and transportation and communications. It also included cross sectoral priorities such as human resource development, technology development, acquisition and transfer, formulation of industrial strategies and policies, integration of women in industrial development, energy and environment.
- 188. b. <u>Supplementary Activities (YA)</u> are designed to help specific IDDA objectives in respect of skills development and promotional activities intended to encourage access to technology and investment resources.
- 189. c. <u>Short-term Technical Advisory Services STAS (YA)</u> respond quickly to specific requests from African countries and organizations to provide on the spot advice on issues related to industrial policy, programme development, private sector development, investment promotion, training and sub-regional cooperation, technology and resource mobilization.
- 190. Financial earmarkings for the above-mentioned types of services were as follows:

Financial Allocation and Distribution in US dollars				
	1994 - 1995	1996-1997		
a. Technical co-operation (XA)	4,137,500	5,509,310		
b. Supplementary activities (YA)	4,216,200	3,350,900		
c. STAS (YA)	1,183,200	1,684,742		

- 191. The formulation and implementation of the UNIDO Support Programme for the Second IDDA took into consideration the following IDDA goals and objectives as reflected in the IDDA II Programme:
 - a. Structural transformation of African economies:
 - Industrial strategies and policies;
 - Development of industrial sub-sectors, especially agro-industries;
 - Environment and energy.

- b. Linkages between industry and other economic sectors:
- Linkages between industry and agriculture;
- Support to transport and communications and linkage with UNTACDA II.
- c. Mobilization of financial resources for:
- Private sector development;
- Human resource development and training
- d. Promotion of regional/sub-regional industrial co-operation:
- Support to regional and sub-regional organizations;
- Coordination with OAU, ECA and ADB.
- e. Co-operation with other technical organizations

192. During the period of 1994 - 1995 a total of 185 projects/activities were undertaken, broken down as follows:

During 1994-1995	Total	Regional Projects	National Projects
a. Technical co-operation	52	27	25
b. Supplementary activities	52	41	11
c. STAS	81	-	-

193. The 1996-1997 (as of 28 February 1997) projects/activities are broken down as follows:

During 1996-1997	Total	Regional Projects	National Projects	Global
a. Technical co-operation	36	12	23	1
b. Supplementary activities	37	11	26	-
c. STAS	66	-	-	-

- 194. UNIDO offered a wide range of technical co-operation services within the framework of the Second Industrial Development Decade for Africa (IDDA). (For details of UNIDO projects and activities carried out in 1994-1995 and 1996-1997, please see Annex 4a, 4b, 4c and 4d.) These services were provided with consideration of the objectives and areas of concentration of the United Nations New Agenda for the Development of Africa in the 1990s.
- 195. The support programme in Africa focused on strengthening industrial institutions, developing strategies and policies for accelerated industrial development; private sector development with emphasis on strengthening private sector institutions; small and medium industry development, rehabilitation of key industries, including the introduction of cleaner production techniques in some countries; developing

human resources for industry; assisting women to meet the challenges of industrial and technological change in a competitive, global economy; and promotion of technology and investment.

UNIDO Assistance to Sub-regional Projects and Activities

- 196. Implementation of projects in countries of central Africa sub-region faced a number of constraints because of prevailing conditions which did not allow execution of projects approved for countries Angola, Burundi, Rwanda and Zaire. The conditions in Angola have somewhat improve.
- 197. In the North Africa region which is mostly constituted by middle income countries the proportion of projects carried is relatively low compared to Western or Eastern/Southern Africa subregion. However, a number of regional projects were carried out which include industry-related services (in Tunisia, Morocco, Algeria); investment forum in Casablanca, Morocco, May 1996; SME umbrella programme and the automotive study (Egypt), and project on INTIB (Egypt and Algeria). The situation in Somalia did not allow the execution of projects and activities in this country.
- 198. With regard to Eastern/Southern and West African sub-regions despite the high proportion of projects and activities in those sub-regions, the internal situation in some of the countries in the sub-regions also made it difficult for projects to be carried out especially in Mozambique, Liberia and Sierra Leone.

IDDA Projects and Activities in LDCs

- Africa has 33 of the least developed countries out of the world total of 48. These are distributed among the sub-regions as follows: 3 in the North African sub-region (Djibouti, Somalia and Sudan); 10 in the Eastern/Southern region (Comoros, Eritrea, Ethiopia, Lesotho, Madagascar, Malawi, Mozambique, Uganda, United Republic of Tanzania and Zambia); 8 in the Central African sub-region (Angola, Burundi, Central African Republic, Chad, Equatorial Guinea, Rwanda, Sao Tome and Principe and Zaire); and 12 in Western African sub-region (Benin, Burkina Faso, Cape Verde, Gambia, Guinea, Guinea-Bissau, Liberia, Mali, Mauritania, Niger, Sierra Leone and Togo).
- 200. It is significant to note that LDCs accounted for a high proportion of projects and activities financed from the funds allocated for the IDDA. 63% of all STAS activities were accorded to the LDCs. The LDCs were also accorded high priority in the regional and sub-regional projects and activities financed under the technical component of XA and supplementary activities (YA) funds.

Conclusions

- 201. The individual projects which were implemented are in line with goals/objectives established under the IDDA II Programme.
- 202. The concept of the "seed money" within the IDDA framework had a marginal effect in terms of attracting funding from other sources.
- 203. Projects implemented were relevant and though in most cases efficiently delivered, the likelihood of positive effect and impact was hindered given the political, economic and social constraints prevailing in the continent. This is particularly true for projects which were involved in the development of

manufacturing processes. Consequently, there were difficulties in effectively transferring the technologies to the private sector.

- 204. While the investment for awere effective and efficient in bringing the prospective investors together, very few if any, investments and joint ventures actually took place.
- 205. The industrial sector review missions undertaken were in many ways helpful in the identification of industrial constraints in the manufacturing industry and in highlighting measures to mitigate those problems identified. However, there were little follow-up activities actually undertaken to address the problems to improve the manufacturing industry.
- 206. At the project level, a number of institutions were implementing projects financed by IDDA II. However, most of these institutions, while aware that funds were coming from UNIDO, did not specifically associate them with IDDA II funds.
- 207. The IDDA II is not adequately publicized or sold at the national level, which may be attributed to meagre financial resources available for the Programme.
- 208. The projects developed, designed and implemented by UNIDO in the various sectors and subsectors at the national and sub-regional levels were compatible with the overall goals/objectives of the IDDA II national programmes objectives. Within the limited resources available, projects were developed as far as practical to meet the broad range of activities envisaged for implementation under the Programme. Thus projects which were delivered were relevant to IDDA Programme objectives. While these individual projects were normally efficiently delivered, the evaluation could not conclude how much the results achieved contributed to the overall goals of the Programme.

Guidance for the Future

- 209. In future undertakings within the IDDA framework, UNIDO should take into account that all partners in the development process (the private sector, Chambers of Commerce and Industry and industry associations) have a role to play in the formulation and implementation of national IDDA programmes.
- 210. An effective dialogue should be maintained as part of a continued development process between all actors with greater participation of the UNIDO field offices.
- 211. A greater involvement of the UNIDO field offices is required to facilitate specific assistance in identifying impediments in industrialisation efforts and in providing advisory services on strategies required to overcome them in coordination with Africa Bureau at UNIDO Headquarters.
- 212. Given the limited resources available to UNIDO's IDDA Support Programme, a strategy should be developed at an early stage in focusing levels and degree of UNIDO assistance on policy, institutional and plant levels and in determining an effective linkage between these levels.
- 213. There should be an effective follow-up system to ensure that results gained from the implementation of projects can be maintained and sustained.
- 214. The mission did not find any evidence that a participatory approach was undertaken with beneficiaries in the formulation of projects.

- 215. UNIDO IDDA Programme should co-operate more with national and regional development banks to facilitate the mobilizing of funds to industrial projects.
- 216. Despite the fact that many projects addressed issues of structural transformations at policy level and in strengthening industrial human resource development, the fact remains that many countries visited still do not have an industrial policy/strategy. The capacity and capability of personnel are still weak in not being able to adapt to the changing needs and demands of today. Shortage of adequate manpower skills in management, policy/decision-making is still lacking. Thus effective guidance cannot be extended towards industrial development.

ii. ECA Support Programme

- 217. ECA has undertaken a number of studies in support of IDDA II Programme implementation. These studies were:
 - a. possibilities and perspectives for the production of basic chemicals from natural gas in Africa;
 - b. promotion of investment and industrial projects in the context of the Second IDDA;
 - c. technology transfer, acquisition and negotiation;
 - d. development of African metal industries based on South-South Co-operation;
 - e. utilization and perspectives for integrated development of natural resources and for the production of phosphate;
 - f. manual for trainers for small-scale industries;
 - g. manufacturing of irrigation equipment in the Sahel countries;
 - h. the leather and leather products industry in African: current status and prospects;
 - i. measure to consolidate privatization in the African industrial sub-sector with special emphasis on the Second IDDA;
 - j. the study of institutions and modalities for promoting the financing of small and cottage industries in Africa.
- 218. Technical studies and ad hoc expert group meetings were also organized during the period under review in order to give opportunity to high level African experts to discuss, share their experiences on specific topics and review the studies before finalization, reproduction and distribution to African endusers including planners, policy-makers, industrialists, manufacturers and potential investors.

Technical Publications

219. Within the context of the Second IDDA, the Secretariat of the ECA has produced various technical publications. In the field of industrial policies and planning, the thrust of ECA's activities was directed at assisting African countries in reformulating existing industrial policies and plans as well as

formulating new policies whenever necessary with a view to reorienting such policies and plans towards facilitating the implementation of the Programme for the Decade. The activities of ECA took the form of recurrent and non-recurrent publications.

- 220. Four issues of the bulletin, Focus on African Industry were published. The articles in the four publications covered a wide spectrum of issues including the critical role of the informal sector in the development of indigenous industrial capabilities, development of international competitiveness, private sector and entrepreneurship development, privatization in Africa, the Uruguay Round Agreements and implications for African industry and the present status of African metallurgy industries and prospects for the year 2000.
- 221. In order to assist African countries in the development and promotion of entrepreneurial and technological capabilities, two Directories of Project Profiles for Small-scale Industries were produced. The two directories provided techno-economic information on, among other things, products to be manufactured, process description, capacity and specifications for plant and machinery, raw materials and other inputs, financial requirements in the form of fixed and working capital, marketing and product distribution, and a programme to assist and advise small-scale industrial entrepreneurs in their manufacturing activities. A manual for Trainers in the Development of Entrepreneurship in Small-Scale Industries was produced and disseminated.
- 222. In order to assist member countries in their efforts to develop and promote industrial cooperation at sub-regional and regional levels, the Secretariat of the Commission prepared in 1993, a technical publication on Industrial Development Priorities and Sub-regional Co-operation in the context of the Second IDDA. The document identified industrial priorities in each sub-region and made concrete recommendations for accelerating African industrial development.

223. Other technical publications include:

- Investment and Financial Policies and their Impact on the Development of Indigenous Industries;
- Lessons from Selected Newly Industrialized Countries for the Effective Implementation of the Second IDDA;
- Sustainable Financing of Selected Second IDDA Multinational Industrial Projects in African sub-regions;
- Selected technologies available with or without Licensing in the context of the Second IDDA;
- Factors in Enterprise Formation in African Countries;
- Possibilities and Perspectives on the Development of Petrochemical Industries in North Africa on the Basis of Local Reserves of Oil and Gas;
- Building Materials and Construction Industries in Africa: Constraints and Prospects for Socio-Economic Development;
- Study on African Potential Industrial Comparative Advantage in an Emerging New Economic Order;

- Modalities for South-South Co-operation;
- Approach on the Formulation of Comprehensive Construction Industries Planning and Implementation Policies in Africa;
- Lessons for Africa from Selected Case Studies on the Development of the Mining and Metal Industries;
- Possibilities of Utilization of Biomass for the Production of Basic Chemicals in Africa;
- Survey of Africa's Basic Industries;
- Survey of Africa's Agro-based Industries;
- Promotion of African Fertilizers and Petrochemical Industries.

These were some of the non-recurrent publications produced with a view to providing African planners and policy-makers with necessary information for accelerating the industrialization process of the continent.

224. As basic industries require massive capital outlay and larger markets beyond the capacity of individual African countries and can therefore be optimally established through sub-regional and regional co-operation arrangements of multinational/transnational industrial enterprises, ECA's activities in the field of chemicals, metallurgical, engineering and agro-industries were to assist member countries in their efforts to lay a solid foundation for the development and promotion of those industries. The publications listed above, therefore, dealt with specific problems such as production and utilization of fertilizers in Africa and perspectives for their integrated development, manufacture of agricultural tools, implements and low cost transport equipment by small-scale engineering industries, development and repair and maintenance of industrial facilities in Africa and building materials and construction industries in Africa.

Technical Advisory Services

225. ECA undertook a number of advisory missions to about eighteen countries during the period under review with a view to facilitating improvement in capabilities for formulating policies and strategies within the context of the Second IDDA. Following the missions, detailed reports with proposals on the formulation of industrial policies and development plans were prepared and sent to the concerned countries. The reports also provided guidance for the implementation of national programmes for the Decade. In addition, the Secretariat of the Commission continued to provide the African Regional Centre for Engineering Design and Manufacturing (ARCEDEM) and the African Regional Centre for Technology (ARCT) with assistance in administrative backstopping and training of trainers in the two centres.

iii. OAU Support Activities

OAU has undertaken a number of activities in support of the IDDA II Programme implementation. The following are worth to be mentioned:

226.a. A comprehensive report on industry-related activities on ways and means of mobilizing resources, financial, human, institutional and infrastructural developments was presented at the

64th OAU Council of Ministers as well as the 32nd OAU Summit Conference held in Yaoundé, Cameroon in July 1996. During the conference, the resolution prepared by OAU and UNIDO on the Alliance for Africa's Industrialization was adopted.

227.b. The OAU organized the First ECOSOC Ministerial Meeting of the African Economic Community in Abidjan, Côte d'Ivoire (November 1996) during which a number of resolutions were adopted, some of which addressed industrial development issues.

The major outcome of that meeting was, inter alia, the adoption of strategies and approaches for the implementation of the Treaty and the adoption of the Protocol governing the relations between the African Economic Community (AEC) and the Regional Economic Commission (REC). In this connection, the OAU is planning to organize the first meeting of a Committee on Industry, Energy, Science and Technology, Natural Resources and Environment during which the draft Protocol on Industry to be annexed to the Abidjan Treaty of the AEC will be discussed.

- 228.c. The OAU has paid working visits and participated in the sub-regional meetings of COMESA and SADC in the framework of the implementation of the Programme of the Second IDDA.
- 229.d. The OAU initiated actions and studies for the creation of the African Energy Commission for which consultations are on-going with the African Development Bank and the ECA for the organization of an inter-agency meeting to adopt the required studies and time table.
- 230.e. The OAU organized the All-Africa Trade Fair in Kaduna, Nigeria in March 1997. It has also organized, in co-operation with the League of Arab States, the 2nd Afro-Arab Trade Fair which took place in Johannesburg, South Africa in October 1995. It is also organizing in November 1997 in Dubai, United Arab Emirates, the 3rd Afro-Arab Trade Fair. In these trade fairs, the OAU organizes for a on industrial investment for Africa.

5. CONSTRAINTS TO THE IMPLEMENTATION OF THE PROGRAMME OF THE SECOND IDDA

- 231. The Second IDDA was launched in 1993 and its implementation covers a period of barely four years. Although national programmes were prepared on the basis of common framework and guidelines, they reflected different political, economic and social diversities and scenarios that portrayed national circumstances and levels of economic and industrial development. National programme implementation has therefore differed from country to country with successes and failures originating from different country-specific factors, and although there are general impediments and constraints to Africa's industrialization as a whole, this chapter deals with constraints of common traits which appear to have impeded implementation of national and sub-regional programmes of IDDA II.
- 232. A summary of factors which are considered to have impeded the IDDA Programme implementation is provided below:
 - no methodologies and strategies were developed to facilitate the formulation, development and implementation of the IDDA Programme by national governments, most of whom simply followed the general guidelines provided by the Joint Secretariat.
 - no mechanism existed to ensure that the IDDA objectives and aims were part of the national development plans, backed up by adequate financial and human resources;
 - frequent changes of government personnel at various ministry levels did not contribute to confidence building towards the IDDA Programme activities as planned;
 - a lack of commitment and understanding on how to proceed with IDDA Programme implementation on the part of the national governments;
 - despite liberalization of policies, investments (internal and external) did not materialize;
 - poor infrastructure facilities were also bottlenecks to attracting investments and in promoting inter-regional trade;
 - the UNIDO Support Programme, which encompassed delivery of some 185 projects during the 1994-1995 period and another 100 during the 1996-1997 period as of February 1997, wherein each project on the average had about a financial outlay of US\$100,000, was not seen by the national governments as specific IDDA but was considered within the overall context of UNIDO-financed projects in a given country:
 - programmes with much larger financial allocation coming from the World Bank and IMF addressing issues of poverty alleviation, education, infrastructural and utility development attracted much more attention of national governments; consequently, the IDDA Programme was no longer a priority but was gradually marginalized, existing only as a side line within the overall development process.

A. At National Level

Out of the above summary, the following constraints have been singled out for emphasis.

1. Non-existence of National Machinery for Programme Implementation

233. National programmes of IDDA II were broadbased, encompassing multi-disciplinary areas of operation. They dealt not only with purely industrial programmes such as rehabilitation and expansion of manufacturing enterprises, but also non-direct industrial activities at the plant and human resource development levels. The range of activities under the programmes were such that existing government departments like the Ministry of Industry and Trade could hardly be capable of adding to its traditional functions the responsibility of managing the IDDA II Programme. This is why in the design of the Programme, special mechanisms of implementation, monitoring and co-ordination were proposed. While governments may have had good reasons for not creating these additional committees and commissions, their non-existence has become a serious constraint to the implementation of the national programmes. Again, the non-existence of these implementation mechanisms, most of which were envisaged to be inter-ministerial and inter-disciplinary, contributed to the declining perception of the IDDA II Programme both in the public and the private sectors.

2. <u>Inadequate Financial Resources</u>

- 234. Due to the far-ranging scope of the IDDA II Programme, no estimate of implementation outlay was made. In fact, this could not be made over a ten-year period with any measure of realism. National governments were however, expected to have assessed their own capability for mobilizing external and domestic funds for Programme implementation. On the basis of specific programmes and projects within the overall national programmes, governments were expected to seek financing from bilateral sources of Overseas Development Assistance (ODA), multilateral funding institutions and relevant UN agencies. The fact that in some countries infrastructural developments as well as capacity building and human resource development through institutional support facilities and training are actively underway, shows that those governments have succeeded in mobilizing resources for those segments of the IDDA Programme. Unfortunately, these successful cases are few and far apart. The majority of countries are facing financial constraints evidenced in either total lack of infrastructural facilities (roads, ports, communications, etc.) or the poor state of disrepair of these facilities.
- 235. Mobilization of financial resources to implement the IDDA Programme is dealt with elsewhere in the Report (see Chapter 6). This section touches on the inadequacy of the stock and flow of funds for implementation in the industrial sector. Inadequate financial resources is a major constraint and so is the lack of capacity not only to mobilize, but also to adequately and responsibly utilize the resources. The severity of their constraint may be minimized if member governments had taken action to strengthen their machinery for financial resource mobilization.
- 236. First, negotiation machinery for multilateral and bilateral financing needs strengthening. Negotiators must be assisted with good and realistic projects and programmes and not just development ideas that no longer convince the donor community.
- 237. Second, machinery for domestic resources mobilization must be strengthened, including expansion of the tax net and deficit reductions.
- 238. Third, capacity to utilize resources timely and responsibly especially programme and project-related loans and aid. A programme for capacity-building in this regard is most essential and could itself be financed with ODA funds.

3. Human Resource Constraint

239. The level of human resources has hampered in the implementation of national development policies and programmes in African countries. Of all the constraints to the IDDA Programme implementation, the most critical and devastating is the lack of adequately prepared human resources, e.g. trained manpower, entrepreneurs, skilled managerial personnel, engineers, technologists, etc. After almost four decades of political independence, the majority of African countries have not been able to develop the critical mass of skills, knowledge and attitudes which enable societies to transform their natural resources into value-added commodities and products for themselves and for export. Low level of human resource development is the main cause of poor level of economic activity. Lack of trained manpower also seriously affects the quality of the public service. The importance of human resource development has been underscored in the IDDA Programme. The member countries have assigned very high priority to this area whose development is strongly supported in the assistance programme of such agencies as ILO, UNDP and the World Bank.

4. Infrastructure Constraint

- 240. Lack of physical infrastructure particularly transport and communication services, is another serious impediment to the Programme implementation. Lack of physical infrastructure is often associated with low standards of education especially in the rural areas of African societies.
- During the last fifteen years, physical infrastructure has seriously deteriorated and many African countries have decreased resource allocation to the sector. Road transport, railways, energy, electricity and water supply, telecommunications and air links have deteriorated in the face of growing population and development needs. Even where these facilities are available, such as water and electricity, they are not accessible to the majority of the population which has also contributed to the decline in living standards. They also become high-cost factor inputs to industry and tend to discourage potential investment. In countries where an investor has to wait for months, if not years, to have a telephone line, water or electricity supplied to his plant, the prospects for a serious business activity is not bright, for hardly any foreigner will invest/venture into such an area. Infrastructure development, both physical and institutional is a prerequisite to industrialization and African countries cannot move further without considerable investments into this sector. The IDDA II Programme has laid strong emphasis on this sector, and governments must increase their efforts in mobilizing agency and other donor-support for development of the physical infrastructure.

5. Political Instability and Lack of Continuity in National Policy Making

242. Political stability and continuity in government policy making have a direct bearing on the implementation of any development programme or national agenda. In between the preparatory phase of the IDDA (1991-1992) and the period of mid-term evaluation (1993-1997), African countries and subregions have witnessed a number of local and national political upheavals, and sub-regional conflicts and civil wars. Three countries in North Africa have been affected by civil war and one by an international embargo. In Central Africa, civil conflicts in Rwanda, Burundi, Zaire and their spill over effects on neighbouring countries in terms of displaced people and refugees, in West Africa, civil wars in Liberia and Sierra Leone, in East and Southern Africa, civil strikes in Somalia, Mozambique and Angola and the uneasy relations between Sudan and its neighbours all have had heavy toll on national economic resources. In their persistence, these factors have constrained economic development and have directed valuable and scarce resources into conflict management and unproductive activities. As result, many African countries have failed to mobilize and organize human resources for national development.

- 243. From 1991 to the beginning of 1997, a number of countries in the region went through at least two to three changes of government. Many of these changes of government came with changes in national policies and implementation strategies. Implementation of the IDDA Programme suffered from this lack of policy continuity.
- 244. On the other hand, the recent move towards democratization is having a positive impact on the management and transparency in development policy. It has brought an appreciable degree of positive inter-relationships between government and the private sector. It is leading to a consensus building on the national development agenda. This will have a tremendous effect on national policy direction and the way national development is perceived by all partners. The implementation of development programmes such as the IDDA Programme will be greatly enhanced by stability and some measure of continuity in policy making.

6. Devaluation of CFA Franc

- 245. The devaluation of the CFA Franc was the natural outcome in the implementation of structural adjustment programmes. The affected countries export mostly agricultural raw materials (one or two) whose prices are fixed on the world market.
- 246. In reality, the devaluation of the CFA Franc has meant for the industrial sector of the countries concerned the doubling of, the cost of imported inputs. This has meant resources constraints and much less liquidity for entrepreneurs. The devaluation has resulted in credit squeeze, less investment in the industry and treasury problems for many enterprises. This monetary adjustment has shot up prices and interest rates. Interest rates rose in between 20% to 35%. In the process, the funding and implementation of national development plans have been adversely affected, including the implementation of the IDDA national programmes.

B. At Sub-regional Level

- 247. To a great extent, the constraints in the implementation of the Second IDDA Programme at the national level are also applicable to the sub-regional level. The understanding at the inception of the sub-regional programme of the Second IDDA was that each country in which a sub-regional project is located shall take the primary responsibility of mobilizing the factor inputs and of monitoring the implementation of the project while associating other countries partly to it. As in the case of the implementation of IDDA national programmes, similar constraints were encountered in the implementation of sub-regional programmes. The constraints are described below:
- 248. First, the awareness of the sub-regional programmes of the Second IDDA was limited to the Ministry of Commerce and Industry of the member states and did not receive adequate diffusion.
- 249. Secondly, political instability in many African countries and frequent change of government officials and civil servants have had serious bearing on the knowledge and follow up in the sub-regional programmes.
- 250. Thirdly, as in the case of national IDDA programmes, the implementation of structural adjustment programmes did not focus on industrial development. Moreover, as per SAP's prescriptions governments are no more to play an investor role in industrial projects but that of a facilitator and coordinator. Resources under SAPs are earmarked for private sector programmes and projects. Furthermore, as most African countries have limited financial resources, one should not expect government resources being allocated to sub-regional projects. Besides, governments themselves did not

seem committed to sub-regional and regional industrial co-operation projects. The sub-regional IDDA programmes and projects were not subjected to national governments consideration and approval.

- 251. Fourthly, the structural adjustment programmes sponsored by the World Bank and IMF have compelled African governments to disengage from direct industrial manufacturing activities and privatize public enterprises. The process of privatization is still on, some countries have gone half way in it, some are having second thoughts while others have just started venturing into it. The whole and long exercise of state disengagement, privatization and laying a sound ground for the private sector to take the lead have taken African policy makers to task. Governments are much aware that until such conditionalities are met, World Bank and IMF will continue putting the pressure. One wonders if in such atmosphere interest in sub-regional IDDA projects can be sustained.
- 252. Fifthly, the collapse in some cases and inefficiency of many sub-regional economic groupings such as sub-regional economic communities, development banks and co-operation institutions have brought a new dimension in the perception of sub-regional industrial projects and their implementation.
- 253. In Central Africa, the Economic Community of the Great Lakes Countries (CEPGL) and affiliated co-operation programmes and institutions under it, viz the Development Bank of Great Lakes States (BDEGL), the Great Lakes Energy (EGL), the Agricultural and Stock-Breeding Research Institute (IRAZ) have collapsed as a result of civil wars. The PTA Development Bank in Bujumbura has relocated to Nairobi and the ECA Mulpoc sub-regional office in Gisenyi in charge of sub-regional industrial cooperation relocated in Addis Ababa.
- 254. The Economic Community of Central African States was expected to be the overall umbrella organization in the implementation, co-ordination and monitoring of sub-regional co-operation programmes, of which the second IDDA Programme has virtually collapsed and its offices in Gabon closed. The Central African Customs and Economic Union (UDEAC) in existence since almost two decades had no impact and is grappling for survival. UDEAC is in the process of being transformed into a Treaty establishing an Economic and Monetary Community for Central Africa along the same lines as the Economic and Monetary Union of West Africa (UEMOA) of CFA Franc Zone.
- 255. The Bank of Central Africa States (BDEAC) in Brazzaville has been restructured and has ceased to provide loans to national and sub-regional economies. The Bank is surviving on collection of loans disbursed before 1992 and on income from previous investments.
- 256. The Multisectoral Institute for Applied Technology for Planning and Evaluation of Projects (ISTA) in Gabon and the Sub-regional Institute for Statistics and Applied Economics (ISEA) in Cameroon struggle to survive, having serious financial and management problems.
- 257. The major problems which impede the implementation of sub-regional industrial programmes and economic activities in general are poor infrastructure; roads are accessible only during dry seasons, railway systems are outdated and telecommunication systems and connectivity are poor. The financial systems are operating inefficiently with some banks collapsing and/or providing lengthy and slow money transfer within and across countries.
- 258. Shortage of skilled manpower is another major constraining factor in industrial development. There is a serious dearth of qualified manpower at all levels but specially technical skills and entrepreneurial capabilities. In some countries of the sub-region, it is very difficult to find technicians such as plumbers or welders.

- 259. Out of forty-nine industrial and support projects listed in the IDDA II sub-regional programme, on the whole, none has been implemented.
- 260. In the Eastern and Southern Africa sub-region, the Common Market for Eastern and Southern Africa (COMESA) has been active in promoting subsidiary development institutions while sub-regional and regional institutions relevant to the IDDA programmes are not functioning properly and have not been effective due to lack of financial resources and commitment from member states. These include the African Regional Organization for Standardization (ARSO), the African Institute for Higher Technical Training and Research (AIHTTR), the Leather and Leather Products Institute (LLPI), the Metallurgical Technology Centre (MTC) and the Mineral Resources Centre (MRC).
- 261. Another constraint which has adversely affected the implementation and interest to sub-regional IDDA projects is the proliferation of co-operation institutions. Such a situation may create rivalry, duplication of efforts, lack of focus, overlapping of functions and send wrong signals to member states and development partners. Some member states belong to as many as five sub-regional economic groupings and are not able to contribute effectively to their programmes.
- 262. In the North African sub-region, industrial development has reached a higher level of performance and sophistication than in Sub-Saharan Africa. North African industrial development is much concerned now in the new requirements for global competitiveness through the application of requisite skills, technology, management and targeting growth industries based on international and domestic demand over the next twenty years. National and sub-regional industrial efforts are directed towards meeting ISO 9000 and ISO 14000 requirements and certification.
- 263. Sub-regional industrial co-operation has not made substantial headway in spite of a great number of technical and financial institutions created in virtually all the industrial sub-sectors to promote regional co-operation. North African countries are presently more European and Mediterranean-oriented in their economic, industrial and trade relations.
- 264. Out of seventeen core projects and fifteen support projects listed in the IDDA sub-regional programme, practically none has been effectively implemented. The reason for poor level of implementation given in the case of sub-regional projects in other sub-regions apply equally to sub-regional core and support projects in North Africa. In addition, uneasy relations among the countries, political instability and sanctions have surely contributed to sub-regional co-operation and implementation of sub-regional industrial projects not having made substantial headway.
- 265. In West Africa, the implementation of SAPs policies has taken precedence on issues of subregional industrial co-operation and implementation of the IDDA II sub-regional programme. The Economic Community of West African States (ECOWAS) has lost momentum and does not play any tangible role in fostering and strengthening sub-regional industrial co-operation.
- 266. As in the case of other sub-regions, out of 43 core projects and 17 support projects identified in the IDDA II Programme for West Africa, equally none has been implemented. The views of member countries is that in the prevailing context of structural adjustment programmes and market economy, IDDA II sub-regional industrial projects fall hence in the domain of the private sector.
- 267. Information seems to indicate that some projects identified in the IDDA II Programme have been implemented by the private sector. The project on the salt of Tidekelt in Niger is a case in point.
- 268. The major constraints in the implementation of IDDA II subprogramme were, inter alia, political instability which affected many countries in the sub-region, the adverse effects of the implementation

of SAPs policies, to certain extent the apathy to sub-regional co-operation and the creation of parallel sub-regional economic co-operation groupings and programmes during the period of the IDDA II Programme implementation.

- As stated above, a number of parallel sub-regional initiatives came to existence during the period 1993-1997. This has led to duplication, overlapping of efforts and resources, and a degree of confusion and distraction in member states and policy makers, and a dwindling interest in the IDDA programmes at national, sub-regional and regional levels.
- 270. As a matter of fact, the Treaty establishing the Economic and Monetary Union (UEMOA) of seven West African countries was signed in January 1994 by Heads of State and Government with the objective of creating a common market based on free movement of people, goods, services and capital flows. The Union having a common currency opens better opportunities for sub-regional co-operation. Similarly, a Treaty instituting the Economic and Monetary Community for Central Africa (CEMAC) was signed in March 1994. The countries, members of the Community have CFA Franc currency in common as well as other financial institutions and economic co-operation groupings.
- 271. While the new sub-regional initiatives have more chances of success in view of their better cohesion and immediate common interests, they have nevertheless lessened member countries interest in IDDA Programme and dampened their support to it.

C. Regional Level

- 272. The essence of the regional programme of the IDDA II Programme lies in the objectives of the Treaty establishing the African Economic Community (AEC) which advocates the economic, social and cultural integration of sub-regional economies through Regional Economic Communities (RECs). The IDDA regional programme derived from national and sub-regional programmes with multinational projects as its pillars.
- 273. In the implementation of the IDDA regional programmes, a number of technical and sectoral institutions have been set up to support and provide the main factor inputs for the industrial development of the region. These institutions are, inter alia, IDEP, ARCT, ARCEDEM, AIDF, ADB, AIHTTR, ARSO, etc.
- 274. However, due to time constraint, the evaluation exercise could not cover in detail the regional component of the second IDDA Programme. What could be stated here is that parallel to the fielding of IDDA evaluation, ECA with UNDP assistance has mounted a mission on the rationalization and streamlining of the above institutions. The reasons being that over the years member states have withdrawn their financial and to some extent their political support to those institutions. Because of their acute financial problems, they have become ineffective and the continuity of many of them is being questioned.
- Again here, the lack of effectiveness and impact of most of these regional institutions on member countries economic development process has over the years contributed to the erosion of member countries' interest in them.

6. MAJOR FINDINGS AND CONCLUSIONS

- 276. The IDDA II Programme was prepared on the basis of national programmes. The prevailing principle was that African governments would take the lead role in formulating industrial policies, build the required infrastructures and create an enabling environment in partnership with the private sector. Goals set under IDDA II included the rehabilitation of existing industries, expansion of specific industrial sub-sectors, promotion of the SMI sector with emphasis on entrepreneurial development and the provision of physical and institutional infrastructure. For achieving these goals institutional mechanisms were to be put in place to ensure effective implementation, monitoring and coordination of the programme both at national and sub-regional levels.
- 277. Meanwhile certain internal and external developments (see Chapter 3) have taken place that have changed both the scenario and the role of the actors.
- 278. These developments, some of which were already in progress at the launching of IDDA II, have underscored the recognition that it is now the private sector that should determine which industrial priorities to focus, based on world market conditions. The private sector broadly defined as all nongovernmental segments of society, thereby assumes the responsibility for leading the industrialization process. With these changes, the IDDA II programme and its priorities appeared to have been overshadowed as government after government struggled to find responses to the pressures of the fast changing internal and external conditions.

A. Perception of the IDDA II Programme

- 279. In the course of the evaluation missions it was noted that the perception of IDDA II as an action-oriented programme for industrialization had dwindled. Apart from ministries of industries which were responsible for preparing the national programmes, the message of the IDDA Programme was hardly disseminated to other segments of government. Most private sector organizations were unaware of the IDDA II programmes and their implications for their own activities. Both public and private sector operators were more concerned with such matters as privatization and ownership, liberalization and impact of structural adjustment programmes on every aspect of economic life.
- 280. The evaluation tried to identify reasons for the waning awareness of the IDDA II Programme:
- In some countries the IDDA II national programmes were regarded as industrial sector contribution to national development plans. Hardly were there any instances where the IDDA II programmes were submitted to established organs of state for approval and adoption. Consequently, the national programmes were not provided with annual budgetary allocations for implementation. In some countries, this lack of support was due to the external inputs and initiatives to prepare these programmes. Despite being "nationally" prepared, they were regarded as "externally" driven.
- Governments have often taken the position that initiatives such as the IDDA II Programme are likely to be implemented with external funds. Some referred to the Economic Recovery Programmes and/or Structural Adjustment Programmes which are funded with loans and grants from the World Bank, IDA and IMF. In spite of the fact that governments accepted responsibility for implementation of their own programmes, expectations still lingered that IDDA II would be implemented with financial support from the international community. Although some projects were financed with seed money provided under IDDA, there was little

or no differentiation between this source of financing and regular UNIDO support activities. Publicity of the IDDA II Programme at the national level was inadequate.

- All African governments accepted fully the central theme/message of IDDA II, namely that industrialization is one of the most important instrument for economic and social development; but each government adopted its own path to growth and development according to national circumstances, and therefore implementation strategies advocated by IDDA were not adopted.
- The industrial sector is only one of the many economic sectors demanding national attention. While industry may have been accorded high priority in the articulation of development policies, in practice it is normally the social and infrastructure sectors that received greater external financial support. Public declarations in support of the industry were hardly translated into action.
- 281. The evaluation therefore took the position that what mattered most was the adoption of the goals and objective set out in the IDDA II Programme. Most of these have been reflected in some national development plans and incorporated in long-term perspectives exemplified in Nigeria as Vision 2010 and Vision 2020 in Kenya and Ghana. The team, therefore, examined the contribution of the industrial sector to the national development effort, irrespective of the perception of IDDA II.
- 282. The evaluation drew the conclusion that unless there is the political commitment to channel budgetary allocations and international resources to industrialization, the IDDA II Programme will suffer the fate of IDDA I.

B. Relations between Government and Private Sector

- 283. In practically all African countries governments have declared that the private sector should receive utmost attention and should be involved as partner in overall development of the economy; and in particular to act as the principal engine of industrial growth. This declaration notwithstanding, many African governments are yet to articulate and formulate policies specifically directed towards enhancing the role of the private sector. However, some countries have established a consultative mechanism between the government and the private sector.
- 284. In many of the countries visited it was noted that the emergence of a strong and dynamic private sector has been hampered by a number of factors:
- First, the private sector organizations realized that they have not been prepared to assume the new responsibility in the industrial process. Where associations of manufacturers etc. exist, they cover primarily the large and medium-scale industries. Small and micro-industrial operators are yet to be organized into viable entities capable of participating in policy formulation.
- Second, there is still a feeling of reluctance on the part of the public sector operators to share the responsibility of industrial development with the private sector. Free flow of industrial information from government ministries and agencies to the private sector operators is lacking. But policy measures are now in place to reduce and eventually abolish bureaucratic controls on enterprises: and private sector representations are gradually being brought into formal discussions with governments and public officials. What is still lacking in many member countries is the evolution of effective consultation mechanisms by which industrial policies and implementation strategies will be discussed and agreed upon in an atmosphere of mutual confidence and trust.

- Third, the private sector representatives often display lack of confidence in dealing with their governments. They often approach discussions and consultations from positions of weakness. This results from lack of knowledge, inability to research and prepare technical background papers and the inability to present a united front for the private sector as a whole. In countries where the public sector dominance of the economy is only now being relaxed and where the private sector was considered unimportant, it still regards with suspicion government efforts to involve it in the development process.
- Fourth, the lack of indigenous pool of entrepreneurs with requisite capacities, managerial and financial, has been a constraint to the rapid expansion of private sector participation in industrial ventures, particularly the small and medium enterprises where indigenous entrepreneurs have capacity to manage.
- Fifth, commercial and development banks do not facilitate access to long-term credits for industrial projects.
- 285. In enhancing the role of private sector, while extricating itself from industrial production, governments have embarked on the process of privatization which will widen the scope of private sector operations. However, privatization has not been successful in many countries because most of the public enterprises were not in marketable condition, that there were not sufficient number of capable entrepreneurs and technical expertise to take over the management of these enterprises, and that sufficient funds for expansion/modernization were not available.
- 286. It was also noted that privatization has highlighted additional issues of concern to both governments and the private sector. Since governments are now working within liberal economic environment, privatization has been opened to foreign investors as well. That many large-scale industries have gone into the hands of foreign nations (sometimes multinationals) has created a problem of new ownership structure with social and economic implications which are becoming worrisome to governments and the general public. Concern has been expressed about eventual repatriation rather than reinvestment of profits, and the concentration of economic power in a few wealthy hands (usually naturalized nationals) and thereby defeating efforts to empower indigenous people and narrow disparities in society.
- 287. Concerns have also been raised in some countries as to whether potential buyers of successfully rehabilitated enterprises should not be induced to invest in new enterprises in order to expand the industrial base rather than changing ownership with the public sector. There was, however, the feeling in some countries, particularly among government establishments that privatization of large-scale projects to foreign investors brings along not only fresh financial resources but also tends to promote the transfer of technology and competitive management in the economy.
- 288. The evaluation concluded that privatization as a national policy is likely to continue in most African countries. But in the pursuit of this policy governments may wish to bear some of these points in mind:
- where wholly state-owned industrial projects have been rehabilitated and are showing evidence of vitality and promise other options of privatization than outright sale may be considered, namely; joint ventureship, management contracts, leasing, subcontracting, etc.;
- the conversion of state-owned rehabilitated enterprises into limited liability companies through public subscription of shares, with reasonable proportion of guaranteed shares for employees;

new investments should be encouraged and directed, through attractive packages of incentives, into industrial priority areas to create opportunities for additional employment and to promote industrial diversification.

C. Consolidation and Expansion Programmes

- 289. Rehabilitation and regeneration of existing industries was one of the major goals of the Second IDDA. This was also a concern expressed by Africa's Priority Programme for Economic Recovery. Obviously, until existing enterprises have been rehabilitated and rationalized, it is unsound to embark on further large-scale investments, especially in the public sector. Under-utilization of installed capacity has become a common characteristic of most industries in Africa. Foreign exchange was unavailable for renewal of old machinery and equipment and purchase of spare parts; raw materials (of foreign origin) were in short supply; electricity supply was either inadequate or frequently interrupted. Markets were lost to competitors for lack of quality improvements and inability to fulfil orders, all of which were underpinned by poor and sometimes corrupt management practices.
- 290. IDDA II national programmes placed special emphasis on rescuing these national assets (public sector enterprises) which otherwise were not only a drain on the treasury but would eventually waste away. Remedies proposed included diagnostic studies of all ailing industries to ascertain factory-specific reasons for under-utilization and to find appropriate solutions. This rehabilitation process was not to be confined to state-owned enterprises but to include private-sector industries where appropriate since they were also victims of the foreign exchange shortages.
- 291. Related to the rehabilitation exercise was the improvement of performance of public sector enterprises. Performance pertained to quality of management, availability of skilled supporting staff and workers as well as the type of technology that would enhance managerial capacity and competitiveness. Political interference in the day-to-day operations of enterprises, and the mix-up of social and commercial objectives were all contributing factors to poor performance of public sector enterprises. Measures for overcoming these deficiencies and to improve productivity and overall operational performance were proposed by the IDDA Programme.
- 292. Evaluation findings concluded that member countries have adopted different remedial approaches to this common problem.
- 293. In countries where structural adjustment programmes were being implemented, privatization was the remedy adopted. The merits and problems of privatization are discussed under Chapter 6.B of this report.
- 294. Some countries like Ethiopia adopted rehabilitation policies even before IDDA II. Operational improvements have been recorded, with some enterprises reaching 80% of capacity utilization and declaring profits. As a second step, some public sector enterprises of less strategic importance were selected for privatization. As profit-making concerns or at least as non-loss-making enterprises, chances of their disposal have been increased.
- 295. Some countries are saddled with state-owned enterprises, which can neither be got rid of or rehabilitated. Nigeria is a case in point. Selected state enterprises iron and steel, petrochemicals, cement, sugar etc. have been opted for privatization, but the private sector both foreign and local, has shown little interest. On the other hand, the state has not been able to mobilize the financial and managerial resources to rehabilitate and regenerate them so that they can be efficiently and profitably

operated. The result is that national assets of considerable value are gradually being wasted, and yet they cannot be left in their ailing conditions. Continued support by the state has become inevitable.

- 296. The IDDA II Programme also stressed the need for industrial expansion of selected core subsectors of industry with multiplier effects, and where member countries have comparative advantages. These sub-sectors included: metallurgical; engineering and allied metalwork; chemicals; agro-industries and food processing; forest-based industries; leather and leather products; industrial systems for fisheries; textiles; construction materials and packaging.
- 297. All these sub-sectors of industry are being pursued in one form or the other in member Countries depending on comparative advantages and country circumstances. In many of the countries visited it was noted that agro-based industries and food processing, forest-based industries, leather, fisheries, textiles, construction materials and packaging are among those that are fairly well established. These areas normally fall within the competence of many indigenous private sector companies, and the technology involved is often not sophisticated.
- 298. With regards to large-scale projects like metallurgical, petrochemical and engineering industries many of which are in the public sector, any expansion works are expected to be preceded by rehabilitation, rationalization and better coordination of existing enterprises. With many state-owned enterprises being privatized the wisdom for further expansion as proposed under IDDA II is being questioned. A new form of collaboration with the private sector for gradual transfer of some of the large-scale enterprises may be considered, bearing in mind that getting rid of ailing industries can be painfully slow and expensive.

D. Economic Liberalization & Industrial Competitiveness

- 299. Economic recovery measures taken by some African governments in mid and late eighties have been followed by structural adjustment programmes which are still being implemented in many countries of the region. Realistic and stable exchange rates, removal of subsidies, reduction of budget deficits and other macroeconomic measures have been introduced. Moreover, removal of bureaucratic controls, access to foreign exchange and liberalization of economic activity in general have been strongly advocated under adjustment programmes.
- 300. In some countries capital/stock markets have also been established. Industrial information is becoming readily available on new markets, sources of industrial raw materials and potential new products. Exchange bureaux have been established for the sale and purchase of foreign currency. All these measures have immensely contributed to the increasing tempo of commercial activity in some African countries.
- 301. Positive as these measures have been in removing import and export restrictions, the evaluation noted that they have also subjected the industrial sector, particularly manufacturing, to serious international competition. For example, it was noted that in some countries the importation of used garments and clothing at lower prices than locally manufactured products has seriously affected the very survival of the textile industry in countries where textile was once a leading industrial activity. Governments may wish to seek protection for such sectors of the economy in consultation with WTO.
- 302. Foreign competition at domestic markets is welcomed by citizens as a whole but private sector operators contend that its introduction should have been gradual, especially at a time when many local enterprises are engaged in rehabilitation, redesigning of their products, and general quality improvements to meet the challenges of local and foreign competition.

303. Some governments are already responding to the challenges of industrial competitiveness by a combination of policy measures including the provision of institutional support services for industry in such areas as research and development, technological development and adaptation to local industry requirements, quality standards, and export promotion. Increased competitiveness is inevitable. This is an area where both the government and the private sector may fruitfully cooperate in searching for the types of technology that would enable local industries to compete and survive.

E. Financial Resource Mobilization for Industry

- 304. The major constraint to industrialization in Africa is lack of financial resources. All the technical advisory services, guidelines, project documents, international proclamations and initiatives will come to naught if money is not available for implementation. This is the case with IDDA II.
- 305. Various financing proposals were made under the Programme in the hope of mobilizing funds for implementation. Foremost among these is the emphasis on domestic resource mobilization, by which governments were expected to demonstrate their determination to shoulder their own responsibility without undue dependence on foreign sources. This demonstration is reflected in the manner by which governments manage their budgetary resources. Budget deficit reductions to achieve public savings are often recommended but not easy to achieve. Savings can be recorded by reductions in military expenditures, reductions in subsidies to public establishments with commercial orientation, reductions in the size of the public service and elimination of wasteful and otherwise avoidable expenditures.
- 306. It was noted in the course of the evaluation that many governments have taken these bold steps as reflected in savings under current expenditure budgets. These measures are essential parts of policies being implemented under structural adjustment programmes. In Ghana and Uganda, for example, massive retrenchment of officials in the civil and public services has taken place. Subsidies to state enterprises have been reduced or withdrawn while privatization is progressively taking place. Unfortunately military budgets in most countries, if not all, are considered to be "state secrets" and therefore any reductions or otherwise in these budgets are kept away from the public. In a few countries military reinforcement is said to be taking place.
- 307. Another step in mobilizing funds through effective budgetary management is the raising of additional revenue. It has been noted that some countries have already reviewed their existing taxation structures and collection practices. Incentives are offered to revenue collectors while malpractices are gradually being reduced. Tariff structures on public utilities have also been reviewed with a view to removing subsidies and thereby increasing revenues.
- 308. One area where much needs to be done is the elimination of wasteful and sometimes avoidable public expenditures. These include such entrenched practices as innumerable range of allowances and entitlements (housing, transport, vacation, etc.) that public servants enjoy, some of which are legacies of the colonial past.
- 309. Surpluses of public enterprise operations constitute a substantial source of public revenue. Not all state-owned enterprises are inefficient and loss-making. Countries like Zimbabwe and Botswana which have followed the (Industrial) Development Corporation model and operate on purely commercial lines have demonstrated efficiency and registered high returns on public investments.
- 310. Public enterprises should have their functions redefined, their commercial operations divorced from social considerations and be freed from political interferences. Given a free hand to compete freely

with counterparts in the private sector, revitalized public enterprises could be a substantial source of revenue for expansion and new investments.

- 311. Domestic resources can also be mobilized from household and corporate savings, and from financial intermediaries. In countries where average income per capita may not exceed US\$ 300 per annum, and where immediate survival needs can hardly be met, household savings are scarce. Yet a large segment of self-employed persons in the informal sector traders, artisans, crop farmers are known to be capable of saving substantial amounts of their earnings either for refinancing and expansion of their operations or for building rental dwellings. To arrest the taxes evasion, some governments have devised measures for levying rates on these economic operations. Efforts are also being made to register these operators, assist them with consultancy services to keep proper books of accounts and encourage them to develop a culture of savings.
- 312. The evaluation team also noted that in some African countries there are a few individuals who are investing abroad or channelling large sums of money into foreign banks for fear of confiscation, should the political environment become hostile. Alongside this group of relatively wealthy individuals, there is a group of African professionals and others living and working abroad and who occasionally make substantial remittances to their countries. Where studies have been conducted, it has been revealed that in a some cases the volume of these remittances, which normally go into real estate development, equal if not exceed the sum total of grants and aid from the donor community bearing in mind that a good proportion of donor assistance is in the form of payments for expert and advisory services.
- 313. The evaluation observed that these potential sources of funds (clandestine funds outflows and remittances) have not been captured in an organized manner. Inducements should be provided to increase the flow of remittances into productive sectors of the economy. Resource outflows into foreign banks is a touchy issue but the view held in many countries is that financial flight is essentially associated with political uncertainty. The capacity of African countries to mobilize internal financial resources for their own development, therefore depends largely on the extent of financial discipline that can be imposed and the ability to overcome political instability and uncertainty through the establishment and promotion of democratic institutions. In this regard the evaluation noted that a large number of member states have accepted the principles of self-determination, pluralism and greater democratization as necessary condition for development in general.
- 314. Some African countries, such as Nigeria, Uganda, Ghana and Kenya, have also instituted financial sector reform measures with a view to restructuring their financial institutions, introducing measures of deregulation in their operations, freeing interest rates, checking excessive growth in money supply, and strengthening the supervisory role of central banks. A few of these countries have created domestic capital markets or strengthened their institutional, legal and operational framework in order to generate greater public confidence in their financial intermediation work. These confidence building measures are all geared towards more effective and efficient mobilization of domestic savings and channelling them into priority investments. In this regard, the restructuring of financial institutions may be considered to enable the provision of long-/medium-term credits.
- One area which is not often associated with domestic resource mobilization is the exploitation of natural resources for which Africa has the most visible comparative advantages.
- 316. The enormous hydro-electric potential, oil and natural gas, minerals and endowment in livestock and forest resources, huge resources of rivers and lakes, together constitute a vast natural resource base that must be adequately exploited, processed and transformed for Africa's sustainable development. Due to the shortage of development finance, lack of technological knowhow, inadequate industrial capacity for transformation into finished products, and the urgent need for funds, Africa's enormous natural

resource endowments are exported in raw form, leaving the transformation process and value added to the industrialized countries as the major beneficiaries.

- 317. In order to arrest and reverse this trend, African countries are expected to acquire the necessary technological knowhow and critical capacities for industrial transformation of natural resources into finished products that would enhance manufacturing value added and industrial competitiveness. The evaluation team concluded that governments should cooperate with external sources of finance and technology, which are usually the multinational and transnational corporations, and seek the assistance of such international intermediaries as the International Finance Corporation and MIGA of the World Bank.
- 318. Mobilization of external resources has been crucial in Africa's development. Multilateral assistance, bilateral aid, foreign direct investments and foreign commercial loans have been the principal sources of external funding. With the present grim financial condition of African countries resulting from heavy indebtedness, adverse terms of trade due to falling commodity prices, inability to raise additional taxes, etc., it may safely be concluded that in the foreseeable future Africa's continued dependence on external sources of financing can hardly be reversed. Unfortunately this is the time when these flows are showing signs of stagnation.
- 319. The evaluation team has observed that while external official flows can hardly be influenced, foreign direct investments can to a large extent. Foreign investments may be attracted if proper conditions, namely favourable investment climate and stable political environment exist.
- 320. To this end, a number of countries have revamped their investment promotion agencies and rewritten their investment codes removing legal and administrative obstacles, offering financial and other incentives, creating export processing zones and trying to improve general conditions of physical infrastructure. Many countries have received international assistance for organizing national and subregional investment promotion fora where foreign investors and local partners meet to discuss and promote specific investment projects. But in all these the results have been minimal. The evaluation team came to the conclusion that the provision of liberal incentives may not necessarily promote foreign investments; and that African governments should look for more deciding factors such as political stability, legal protection of investors and their investments. The traditional approach to Investment Promotion through meetings would need to be revised, bearing in mind that serious investors can no longer be enticed with any half-baked project ideas. In a competitive market one can sell only what appeals to the customer.
- 321. Another observation regarding resource mobilization is the ineffective utilization of funds. One of the complaints of the donor community is the incapability of existing governmental machinery for funds utilization. But although slow disbursement often results from lack of absorptive capacity, it can also be related to cumbersome administrative procedures imposed by donor countries. The evaluation took note of the fact that in some countries machinery for consultation already exists whereby representatives of donor and recipient countries meet at country level with UN Resident Coordinators to review disbursement and other obstacles to funds utilization.

F. Agro-Industry Linkages

322. Linking industry and agriculture is recognized to be in line with the potential for long-term comparative advantage of agro-based production. In Africa there is clear justification for an agriculture-led industrial development strategy with the ultimate goal of faster growth of both the agricultural and industrial sectors, in order to increase supply of domestic food and generate income and employment

through competitive processed agricultural products. Forward and backward linkages between industry and agriculture and their great impact on poverty alleviation and food security make both sectors complementary.

- 323. However, the development of agricultural production in African countries leaves much to be desired. The pressure of population growth and the constraints on agricultural land call for strategies involving in-depth agricultural reforms by the introduction of modern, commercial techniques and application of fertilizers and building of irrigation schemes. Today only a few African countries are self-sufficient in food production and even that varies from year to year depending on weather conditions. Rain-fed agriculture is the practice in most of the countries of the region. The level of irrigation is negligible. Irrigation schemes for large-scale farming are expensive to build; but until this is done agricultural growth will only remain a wish. The evaluation found that there is an urgent need for African countries at national and sub-regional levels to address the issues of water and irrigation policies in order to achieve a breakthrough and a degree of self-reliance and food security.
- 324. It was noted in the course of the evaluation that all member states regard agro-industrial development with high priority. This includes food processing, forest-based industries, textiles, fisheries, leather and leather products and packaging. Since most of these sub-sectors are served by small and medium-sized enterprises, attention is being focused on identifying and supporting indigenous entrepreneurs who are traditionally engaged in these industrial activities. In food-processing there are many examples in branch activities, such as edible oil mills, fish processing, milk production, fruit and meat presentation, salt production, coffee and sugar production. In forest-based industries there are new investments in rehabilitation and expansion of plywood and flush doors, paper and pulp factories, fibre board plants. Many countries have plans to establish or revamp their tanneries, leather finishing, footwear and leather goods industries.
- 325. Agro-based industrial development is one area where many African governments have gone beyond policy formulation, and have set in motion support services and action programmes for implementation. Many countries have ongoing projects specially geared toward enhancing the role of women entrepreneurs who mostly work at the cottage level. NGOs and many international agencies have strong support programmes in many countries for women entrepreneurship development. It is noted, however, that overenthusiasm may produce unnecessary, overlapping activities, and government's coordination role has become important.
- 326. Some few African countries have recognized the need to provide modern inputs to the agricultural sector with a view to boosting agricultural production and interlinking more and more industry and agriculture.

G. Human Resource and Entrepreneurship Development

327. Human resource development, particularly entrepreneurial training and managerial capacity building, and its effective utilization is one of the most critical factors of Africa's industrialization and the path to national and regional self-reliance and self-sustainment. To harness this potential input for development, African governments have long accepted responsibility for providing formal education; and of late they have buttressed this with vocational, technical and professional training institutions. These institutions, depending on their nature are expected to produce the type and quality of manpower whose training and expertise are specifically related to requirements of industry and industry-supported organizations and institutions. Manpower and skills required for industry are the industrial managers, high-level professionals, middle-level specialists, supervisors and skilled workers. It is the combination

of these skills in appropriate quantities that ensures and directs the production process. Supporting these skills are other specializations in marketing, research, procurement, maintenance, etc.

- 328. Evaluation findings are that increasing number of member countries have created institutions for developing the necessary expertise and skills that would eventually meet manpower requirements of industry. The question that remains is whether the numbers being trained are adequate and whether the quality of their training meets the standards normally required by industry whose operations are constantly being influenced by rapid technological changes. That large number of expatriate experts are still employed in many countries for middle and high level positions in industry testify to the inadequate number and quality of trained manpower in the system. The fact that many graduates and secondary school leavers remain unemployed for long periods after graduation is also an indication of the mismatch between education and working requirements of industry. This calls for a serious look at national educational systems with a view to restructuring institutions and teaching programmes to reflect current and future requirements.
- 329. Much more critical is the shortage of industrial entrepreneurs and top managerial personnel on whose shoulders rest the responsibility for guiding Africa's future industrialization. The responsibility for developing those rare qualities of entrepreneurship which are said to be innate and not simply taught in classrooms will continue to be that of the government for some time to come. In this regard, it is encouraging to note that machinery for identifying indigenous entrepreneurs and assisting them now exist in such countries as Nigeria, Tunisia, Mauritius, Ghana, Zimbabwe and Senegal.
- 330. Indigenous entrepreneurship is most needed if efforts to develop small and medium industries are to succeed. At a time when governments are divesting themselves from operation of the productive sectors of the economy, especially manufacturing, the need for trained entrepreneurs and top industrial managers has become imperative. In spite of financial constraints, African governments have little choice but to continue appropriating adequate budgetary resources for developing their manpower resources in order to cope with advancement in production, management and information technologies.
- 331. While every effort is being made to harness human resources for development, the evaluation team observed that a good portion of these resources was being wasted through unemployment and underemployment. Urban congestion, increasing crime rate and creation of city slums are becoming characteristics of African cities. Although governments, NGOs and some international organizations have expressed concern over these developments, no solution appears to be readily in sight. It has been noted that in the final analysis member countries are losing funds by:
 - i. educating and training people in areas who cannot be absorbed into the active labour force;
 - ii. fighting crime and social disturbances in urban centres;
 - iii. coping with increasing health and other social problems created by overcrowding and congestion.

Resource mobilization becomes counter-productive if available funds are utilized for unproductive purposes; and the evaluation concluded that it is an urgent problem that calls for concerted action of African governments and the international community.

H. Physical and Institutional Infrastructure

- 332. The provision of physical infrastructure, consisting mainly of transport and communications, water, electricity and energy supply, roads, railways and ports, has been a traditional role of African governments. The role should continue into the distant future if Africa's industrial transformation should become a reality. Coordination of these services at the national level is of utmost importance not only to ensure timely supply of material inputs and market outlets for finished products, but also to enhance productivity if not the survival of industrial undertakings.
- 333. Sub-regional co-operation in the provision and maintenance of common physical infrastructure is virtually essential in promoting the establishment of joint venture projects within sub-regions and in fostering trade among neighbouring countries, taking into account the need for market expansion, and the advantages of large-scale production and economies of scale.
- 334. The evaluation has confirmed that governments have been constrained for lack of resources to make adequate headway in the provision of these services both at the national and sub-regional levels. Road and railway networks linking parts of the same country and with neighbouring countries are only partially developed thus restraining the movement of goods and people from pursuing their economic functions. As if these constraints were not enough, business people and traders are often subjected to cross-border restrictions and customs harassment thus frustrating and impending trade, even between neighbouring countries.
- 335. However, the provision of physical infrastructure alone, without the support of institutional infrastructure cannot achieve much of desired results. Consequently the setting up of national, subregional and regional institutions aimed at accelerating Africa's industrialization is strongly emphasized in the Second IDDA Programme.
- 336. The evaluation noted that despite financial constraints many governments have managed to establish an impressive range of institutional infrastructure for the furtherance of industrialization. Such institutions exist in all sub-regions of Africa and in various fields of activity of which the following are a few examples:
- National Centre for Assistance to Small and Medium-Size Enterprises (CAPME) in the Cameroon;
- Technical Design and Construction Office (BERETEC) in the Congo;
- Sub-regional Multisectoral Institute of Applied Technology and Project Planning and Evaluation (ISTA) in Gabon;
- Botswana Technology Centre; Rural Industries Promotion Company (RIPCO) in Botswana;
- Scientific and Industrial Research and Development Centre (SIRDCO) in Zimbabwe;
- Africa Capacity Building Foundation (ACBF) in Zimbabwe;
- Bureau of Standards in Kenya, Uganda and United Republic of Tanzania;
- Export Processing Zones in Namibia, Kenya, Ghana;
- Private Enterprise Foundation in Ghana, Uganda.

337. These institutions range from agro-based industries, through scientific and industrial research to entrepreneurial development. It is hoped that their sustainability is constantly monitored and more cooperation and exchanges will be established or strengthened especially among national institutions including associations of manufacturers and bankers, industrial development corporations, research and development institutions. Similar linkages with counterparts outside the region should be created with a view to enhancing institutional capabilities and promoting technology transfer into the region.

I. Sub-regional Co-operation

- 338. The sub-regional groupings recognized under IDDA II programming consist of Southern and Eastern Africa, West Africa, Central Africa and North Africa. Sub-regional programmes were based on those of IDDA I taking into account existing programmes of sub-regional economic groupings and updated at different sub-regional meetings organized by UNIDO, ECA and OAU.
- 339. Programmes for each sub-region consisted of priority industrial sub-sectors and selected institutions were charged with their implementation, coordination and monitoring.
- 340. The evaluation noted that the level of performance in the implementation of programmes varied from sub-region to sub-region depending essentially on the organizational strength and capability of the economic grouping and their institutional mechanisms, the financial resources available, and the political stability within each sub-region.
- 341. In Eastern and Southern Africa, with regard to institution building, COMESA has promoted the establishment of the Leather and Leather Products Institute, the Metallurgical Technology Centre, and the Eastern and Southern Africa Business Organization (ESABO) which is an association of private business enterprises. SADC co-exists with COMESA as they harmonize and coordinate their programmes for their respective members in the spirit of the Abuja Treaty establishing the African Economic Community. The evaluation team noted that countries within the Eastern and Southern African Sub-region have embraced Structural Adjustment Programmes in their industrialization policies and programmes, and this has brought into question the continued relevance of LPA and IDDA II objectives of self-reliance and self-containment. The respective roles of governments and their private sectors will have to be redefined with strong emphasis on governments establishing mechanisms that will ensure political and social stability within which the private sector can operate effectively.
- 342. In the Central Africa Sub-region a long list of organizations and institutions were designated to support, coordinate and implement the sub-regional programme. These included the Economic Community of Central African States (ECCAS), the Bank of Central African States (BDEAC), Central African Customs and Economic Union (UDEAC), Economic Community of the Great Lakes Countries (CEPGL), etc. The evaluation team noted that the priority programme for preparing a sub-regional industrialization master plan has not yet been implemented, neither was the proposed community investment code under ECCAS. Programmes and institutions under CEPGL have either collapsed or not been established because of political instability of the area, resulting in the transfer of PTA Development Bank from Bujumbura to Nairobi and the MULPOC sub-regional office from Gisenyi to Addis Ababa.
- 343. ECCAS which was expected to implement, coordinate and monitor the sub-regional programme has virtually collapsed.
- 344. Some of the major problems hindering economic activities of the sub-region are of infrastructural nature:

- poor roads (inaccessible during rainy seasons)
- outdated railway system
- poor telecommunication systems
- different financial and banking systems
- political and social instability

as well as a shortage of qualified manpower at all levels, especially technical skills and entrepreneurship.

- 345. Due to physical and manpower constraints, coupled with civil wars and general instability of the sub-region, most of the 49 projects listed under the IDDA II Programme have not been implemented.
- 346. It has been noted that, unlike countries of Eastern and Southern Africa, structural adjustment programmes have not been implemented to any meaningful extent in the Central Africa Sub-region. Apart from the devaluation of CFA in 1994, very few reform measures have been introduced in the financial sector. Privatization has not advanced beyond identification of enterprises, except in Cameroon where the telecommunication system has been privatized. Privatization of water, electricity and the railway corporation in Gabon is imminent.
- 347. The Second IDDA Programme appears to have failed in the sub-region for lack of promotion, lack of implementation mechanisms, weak institutional structures and a hostile environment.
- 348. The evaluation team observed that in the North African Sub-region reform programmes have worked well in Egypt and Morocco, while Algeria is still beset by civil unrest. Industrial growth and cooperation in the sub-region have been supported through government policies of liberalization, packages of incentives, development of support institutions and capacity building services aimed at enhancing the role of the private sector and promoting investments.
- 349. SAPs and Uruguay Round Agreements form the framework within which some of the subregional countries are operating as trade is directed towards Europe within the context of Euro-Mediterranean Partnership. There are very few intra-sub-regional trading relations.
- 350. Considerable financial resources have followed the adoption of SAPs and the Euro-Mediterranean Partnership, and this has overshadowed the Programme of IDDA II.
- 351. The UMA is the established authority to co-ordinate economic and industrial development activities in the sub-region and in the implementation of the IDDA Programme.
- 352. In the West Africa sub-region, some 43 core projects were identified under the IDDA II Sub-regional Programme in industrial sub-sectors such as agro-based industries, metallurgical industries, chemicals, engineering, building materials and support services. Organizations and institutions in support of the sub-regional programme included ECOWAS, CEAO, MRU, MULPOC and ARCEDEM. The weakness of these institutions was mainly responsible for lack of effective implementation of the Programme. Although the Treaty of ECOWAS was revised in 1993, it has not managed to establish a customs union nor a common market. There are still a number of tariff and non-tariff barriers to trade, and protocols signed on agriculture, industry and energy are yet to be implemented. It is also observed that political decisions made at high political levels have not been translated into implementable packages at the national level. In this regard, member states would need to recommit themselves to a fuller co-operation under the ECOWAS Treaty. Such financial institutions as the West African Monetary Agency (WAMA) and the ECOWAS Fund for Co-operation and Development hold promise but have not settled into active forces to promote trade and industrialization.

- 353. UEMOA which absorbed CEAO as a monetary union (with a single currency, CFA) appears to be in a stronger position than ECOWAS in terms of policies and programmes which should be translated, consolidated and implemented as sub-regional projects. The Mano River Union has suffered from political instability within the area.
- 354. The major findings in this sub-region are that programmes identified under IDDA II have more or less been submerged into national programmes and are being shaped by structural adjustment programmes which focus on stabilization and liberalization. The IDDA concepts of self-reliance, self-sustainment and industrial transformation through targeting core industries may need to be reexamined in the light of overall dependence on financial backing of SAPs. The tendency is that the sub-region might be transformed into a trading zone with very little production which can only materialize by harmonized policies, complementarity and specialization within the sub-region.

7. THE WAY FORWARD: Lessons and Recommendations

- 355. In drawing up recommendations, the evaluation team felt that certain factors which have affected the credibility and the future prospects of the IDDA II Programme should be spelt out. These factors have had a direct bearing on the acceptance of and support for the Programme and have adversely affected its implementation. Although some of these factors were noted in the evaluation report of the First IDDA, it appears that insufficient attention was paid to them in the preparation of the IDDA II Programme.
- 356. The IDDA II Programme, like the Lagos Plan of Action and the IDDA I, was an expression of intentions and desires for Africa's industrialization. These intentions were not translated into concrete and practical programmes and specific projects for which funds could be mobilized for implementation.
- 357. Special implementation mechanisms at the country level recommended under IDDA II were not established, expecting that the existing national machinery, i.e. Ministries of Industry and Trade or Planning could undertake that responsibility as part of their normal activities. This has not worked.
- 358. Although IDDA II was recommended by African governments and endorsed by the UNIDO General Conference and the United Nations General Assembly, the Programme itself did not involve the donor community and other UN agencies in its design and consequently there was no endorsement and commitment on their part for implementation. Here it must be observed that the IDDA II Programme touched on various areas within the legislative mandate of some UN agencies (physical infrastructure the World Bank/UNDP; human resource development the ILO/UNDP; etc.) and yet the Programme and its promoters did not establish any inter-agency cooperative mechanism for the Programme's coordination and implementation. On the part of the donor community the impression received was that until statements of intentions were translated into specific action programmes and specific implementable projects, it could not provide any financial support.
- 359. Some of the UNDP Resident Coordinators met in the course of field missions confirmed that while industrialization was essential for Africa's overall development, it must be based on specific prerequisites such as manpower and skills development, infrastructural (physical and institutional) development, stable political conditions induced by good governance, etc. and that mere statements of intentions with broad, unwieldy and unfocused programmes of industrialization will not materialize. This was also the view of some donor country representatives.
- 360. The role of the Bureau of CAMI and the Joint Secretariat (UNIDO, ECA and OAU) to monitor the implementation of the Programme appeared to have been absorbed in servicing the various meetings of CAMI rather than promoting the involvement of UN specialized agencies and the World Bank Group in the Programme and assisting African countries in seeking bilateral donor community support for those aspects of the national programmes in which individual donors are likely to show interest.
- 361. The IDDA II biennial allocations of around US\$ 10 million to UNIDO were grossly inadequate to support the Programme. The funds became indistinguishable from normal UNIDO operations. The evaluation team noted that an alternative use of these funds could have been assistance to African countries in mobilizing donor support for specific programmes and projects within the IDDA Programme. This assistance could take the form of specific project identification and preparation, background support in annual bilateral negotiations between African countries and donors, participation in the Paris Consultative Group Meetings, and assistance in building absorptive capacity of government agencies responsible for multilateral and bilateral funds utilization and disbursements.

- 362. The evaluation team got the impression that in designing the IDDA II Programme no special consideration was given to the diversity of African country situations and conditions. A continent-wide Programme without any differentiation between the Northern African countries with relatively higher levels of development and different orientation towards the Middle East and Europe, the land-locked and the very least developed countries with peculiar problems, and the rest of Africa (excluding South Africa) was bound to end up with different approaches and emphasis, leaving the Programme as merely a general guide to industrialization. Although more than 50% of African countries are classified as LDCs, the IDDA Programme made only passing references to them without giving special attention.
- 363. The country missions, which covered twenty-one member Countries in all sub-regions of Africa, have reached the conclusion that for many of these countries, the Second IDDA, like the First, existed only as an initiative. While goals and ideals remain relevant, valid and laudable, the Second IDDA has been devoid of any implementation mechanism which would translate public proclamations into visible action programmes, and goals and objectives into attainable results. For these countries these goals, objectives and ideals made no difference because they were proclaimed in 1980 by the Lagos Plan of Action and repeated by IDDA I. Before IDDA II was officially launched, some of these goals and ideals had already been reflected either in medium-term national development plans or in their long-term national visions. Consequently, the IDDA II Programme, which had no funds to back it up, was regarded as superfluous and somehow dispensable.
- 364. After four years (1993-1997) of existence IDDA II, conceived as a Programme for spearheading Africa's industrialization, has lost its identity within the economic and political vision of its national promoters. Whatever remained of the Programme itself became associated with normal operations of UNIDO, ECA and OAU.
- 365. The IDDA II Programme was launched at a time when Africa was already facing serious internal and external challenges of economic transformation, political transition, globalization and a new environment of international competitiveness. Member states have practically been brought to their knees by the heavy burden of international indebtedness, increasing budget deficits resulting from deteriorating terms of trade and debt servicing, by slackening flow of development assistance and of direct foreign investments. All these factors contributed to a painful period of restlessness that led in some cases to political upheavals, social unrest and civil conflicts.
- 366. Meanwhile, signs of deterioration were seen at every turn. Health delivery services could hardly keep pace with increasing poverty, hunger and disease; physical infrastructure could not be maintained and utility services kept falling apart. In some countries this unstable situation was made worse by persistent indulgence in poor governance engulfed in corruption, nepotism and abuse of individual rights and liberty.
- 367. The first order of business for African governments was to seek for survival. To most African countries the answer lay in the adoption of Structural Adjustment Programmes with strict adherence to macro-economic policies which were accepted by a few with conviction and by many with reluctance. The implementation of the adjustment programmes was accompanied by the practice of "good governance" which was expected to bring about an atmosphere of peace, stability and security.
- 368. In this atmosphere of near-desperation African governments became more and more preoccupied with policies and programmes that were backed with ready financing and which could produce visible results in a short period of time. On the other hand, the Programme of IDDA II had no financing backup and could therefore wait until the immediate crisis had abated. Be that as it may, the concept and the principal objective of IDDA II, namely industrialization as the major engine of Africa's economic growth and development, have remained as the embodiment of the hopes and aspirations of African nations.

This is the spirit that must be recaptured and rendered much more relevant to the present socio-economic circumstances of African countries for the remaining period of the Development Decade.

- 369. To believe that the initial scope of IDDA II Programme can still be adhered to will be unrealistic, especially at a time when the perception of the Programme as originally conceived has been overshadowed by subsequent events, including the recognition of the private sector as the lead actor in industrial development.
- 370. Bearing in mind the major findings of the evaluation, realizing that the Development Decade is almost half way through, and taking into account the financial and capacity constraints that most African countries are likely to face for the rest of the Decade and beyond, it is recommended:
- that the scope and coverage of the Programme be narrowed to make it compatible with the country's capacity to mobilize its human, natural and financial resources for implementation;
- that governments concentrate on those segments of the Programme that are considered as necessary prerequisites for industrial development while leaving the productive and manufacturing aspects to the private sector;
- that governments intensify their efforts at promoting the necessary economic environment and political stability that will make private sector initiatives possible and productive;
- that the private sector and the industrial community as a whole be given support to build up their capacity and enhance their competitiveness; and
- that the donor community especially be invited to have a serious look at the revised priorities, including the Alliance programme for private sector enhancement through partnerships, and to request the opinion, endorsement and participation of the private sector and the chambers of commerce and other related business associations.
- 371. These recommendations take into account the diverse conditions of growth and relative prosperity prevailing in different parts of the region. Emphasis and priorities will therefore differ from country to country and from sub-region to sub-region. But for most countries including many LDCs and landlocked areas, the limited scope approach will appear reasonable.
- 372. At the end of the Development Decade which is only five years hence, each member state may wish to have something for IDDA II to show. Accordingly it is recommended that national attention now be focused on the following specific areas:
 - A. Private sector development;
 - B. Human resource development;
 - C. Regional and sub-regional co-operation and integration;
 - D. Environmental concerns.
- 373. The following two areas are closely related to the above focus of IDDA:
 - Financial sector development
 - Physical infrastructure development
- 374. An overriding issue which affects the industrialization process although it is not an integral part of IDDA but has to be addressed by governments is:

- Governance, political liberalization and commitment
- Within the above context, governments should establish their own priorities depending on the level of development volume of available resources, size and importance of the private sector, etc.

A. Private Sector Development

- 376. African governments now recognize the role of the private sector as complementary in socioeconomic transformation. Both the public and the private sectors must work in partnership, redefine
 their respective roles and responsibilities, while each sector concentrates on areas where it is most suited
 to be efficient and productive. The traditional role of governments must remain intact, namely: to
 maintain law and order (social harmony), to formulate policies and strategies for the guidance of the
 developmental process, to protect the environment, to provide basic education and public health services,
 to provide necessary public infrastructures, and to create an enabling environment within which
 economic and social development can peacefully take place. The private sector's role is essentially to
 mobilize resources, invest and manage these resources in the productive sectors of the economy, while
 contributing to national policy formulation.
- 377. By accepting this reality the public sector cannot do everything from building physical infrastructure to manufacturing consumer products. African governments now consider it desirable and efficient to leave the private sector to determine production priorities based on market forces. But in fostering their relationship with the private sector, African governments may wish to bear the following in mind:
- that sharing economic power and production initiatives with the private sector should not be considered as a favour. One of the lessons from structural reforms currently underway in many countries is the realization that in the way of political and economic liberalization, the roles played by governments and their private sectors are complementary;
- that the strength and effectiveness of the private sector as an innovative and active development partner is yet to develop and it is incumbent on governments to assist in building institutional structures and providing facilities and services that will enhance the private sector capacity to play a meaningful and constructive role in national development;
- that the role of the private sector is that of partnership and not a substitute for government action.
- 378. In Africa's present circumstances governments will continue to provide guidance and leadership in the economic transformation process. If this principle is accepted, suspicion on both sides will give way to mutual respect and understanding, thereby creating the right atmosphere for confidence-building, co-operation and joint efforts in nation-building.
- 379. The private sector base in practically all member countries is small. The critical mass of industrialists to push industrialization forward is yet to emerge. For the rest of the Decade and beyond, African governments may wish to focus on two inter-related areas whose development will help broaden the private sector base and deepen its sphere of influence. These are entrepreneurship development and small and medium-scale sector development.
- a. <u>Entrepreneurship</u>: The strength of the private sector depends on the size and quality of indigenous entrepreneurs. With increasing and significant role now being assigned to the private sector, the importance of identifying potential entrepreneurs and giving them the necessary training and support

services cannot be overemphasized. Such services may include institutional and infrastructural requirements in the form of financial establishments, extension services, consultancy assistance, provision and dissemination of research and development findings and general industrial information, on market opportunities, sources of raw materials and technological changes. Potential entrepreneurs are scattered within the informal sector, and policy measures and mechanisms may be needed for their identification and training in order to initiate them into the formal sector as middle level links in the private sector.

- b. <u>Small and Middle-Scale Sector Development</u>: The training ground for indigenous entrepreneurs is the small and medium-size enterprises, the development of which must remain a major preoccupation in formulating industrial policies and strategies. In almost all African countries this sector constitutes the largest number of establishments, providing most of the industrial employment and accounting for a sizeable amount of manufacturing output. Most of agro-related industries are found in this sector. Other advantages for concentration on small enterprises are that they are mostly rural-based and contribute to industrial dispersion and rural development policies, while arresting urbanization and increasing women entrepreneurship and participation in industry.
- 380. It is recommended that renewed emphasis be placed on the promotion of this sector by establishing and/or strengthening support organizations and facilities such as:
 - industrial estates or export processing zones with common services;
 - industrial cooperatives, inter-firm linkages and partnership;
 - concessional financing and loan guarantees;
 - small industries support departments or organizations to provide project development services including feasibility studies, training and management support;
 - leasing of equipment etc.
- 381. The future of Africa's industrialization depends largely on its indigenous entrepreneurs being capable of managing small and medium-scale enterprises. Given the right economic and political environment and support in technological and managerial capacity-building small and medium-size enterprises can be as competitive and efficient as any large-scale enterprises. The experience in developed countries shows that small- and medium-scale enterprises can be formidable forces in driving forward the industrialization process.

B. Human Resource Development

- 382. Private sector capacity to deliver will largely depend on availability of adequately trained and skilled manpower. It is recommended that for the rest of the Development Decade attention be focused on providing the quantity and quality of industrial manpower needed by the productive sectors of the economy. The development of Africa's human resources will also place member countries in a state of preparedness to accept the challenges of rapid technological changes in production, management techniques, and communications systems.
- 383. Human resource development is by no means limited to the male segment of society. In this regard the role of women in the development process must be underscored.

- 384. Women constitute the majority of the rural population and make crucial contribution to the food production sub-sector. They dominate the informal sector not only in agriculture but also in small and medium-size agro-based industries, marketing and retail trading. Accordingly, education and training of women must be on a par with that of their male counterparts, especially at the technical, professional and managerial levels. This will enable women to have access to income through formal or wage employment, and consequently enhance their role and contribution to industry-related activities.
- 385. This will require restructuring of formal education system; strengthening of technical, vocational and professional training institutions; building the capacity of science, technology and research centres, and ensuring that curricula of these institutions and centres reflect specializations of relevance to industry and industrial support services. The provision of skilled industrial manpower must be seen as part of the enabling environment without which the private sector can hardly operate.
- 386. South East Asian Tigers could not be where they are today without the requisite manpower resources adequately trained in various industrial disciplines and imbued with the spirit of patriotism and dedication buttressed by adequate incentive packages.

C. Regional and Sub-regional Co-operation and Integration

- 387. Sub-regional co-operation in Africa's industrialization is vital and indispensable. Yet sub-regional programmes and projects developed under both IDDA I and II have had very limited success, if any. There are many reasons for failure, but the major cause is the overambitious nature of these programmes. The emphasis on harmonization of national codes and legislation appears to be overplayed, bearing in mind that some member governments are only now attempting to update and modernize national legislation and codes of practice some of which have remained as legacies of the colonial past.
- 388. Attainment of sub-regional integration cannot be achieved rapidly, and therefore attention should be focused for the time being on industrial co-operation where operational mechanisms and procedures may call for less uniformity. Attempts at sub-regional integration through the establishment of common investment projects appear to have made little or no headway as sub-regional organizations and national policy-makers spend considerable amount of time at meetings and conferences arguing about such issues as industrial location and financing mechanisms.
- 389. Taking into account that financial resources for Africa's industrialization will continue to be limited that sub-regional integration cannot be achieved by overambitious programmes involving too many large-scale investment projects, and that sub-regional industrial co-operation must of necessity precede any attempt at integration, the following recommendations are submitted for the remaining part of the Development Decade:
- That the promotion of sub-regional industrial investment projects be left to the private sector which should be encouraged to involve multinational partners outside the sub-regions, while governments play a catalytic role where possible in drawing in international financing institutions:
- That sub-regional organizations concentrate on creating appropriate environment and necessary conditions to enable the private sector and their enterprises to operate across national borders. These initiatives, some of which are already underway, will involve specific steps among which are:
 - a. Relaxation of national barriers, and prompt customs clearances;

- b. Linking of physical infrastructural support services, such as transport and communications;
- c. Provision of those industry-related services, which are cost-effective in finance and skills at sub-regional rather than national level such as scientific and industrial research institutions, quality control and standards institutes, data banks for industrial information, etc.;
- d. Promotion of partnerships and subcontracting arrangements as assistance to private sector development at national, sub-regional and regional levels. Where such partnerships can be established with multinational corporations, the private sector will gain in acquiring new skills and technology in addition to achieving economies of scale and market expansion.
- e. That regional projects be limited to institutional infrastructure development which will provide training and common services to member countries. The experiences of such institutions as the African Regional Centre for Technology (ARCT), African Regional Centre for Engineering, Design and Manufacturing (ARCEDEM) and African Regional Organization for Standardization (ARSO) testify to the need for concentrating on what is already in hand rather than venturing into new areas. Regional efforts should concentrate on transforming them into institutions of excellence capable of providing necessary technological support for Africa's industrialization.
- 390. The importance of sub-regional co-operation is by no means down-sized by shifting emphasis from the pursuit of multinational investment projects to the promotion of sub-regional enabling environment that will support private sector efforts in developing those very investment projects that sub-regional organizations can hardly find the funds and management capacity to implement. If this shift in orientation can work at the national level, it should also work at the sub-regional level where uniformity and unanimity are still formidable obstacles to integration.
- 391. The role of sub-regional economic groupings (COMESA, SADC, UMA, ECCAS, ECOWAS) should be to promote and co-ordinate intra-regional trade and investment and to strengthen and integrate the business relationships among various partners in the region through joint projects and programmes under the umbrella of the African Economic Community.

D. Environmental Concerns

392. Industrialization has its price in terms of social and specific environmental problems such as desertification, effluent disposal, industrial refuse and natural resource degradation. While some of these issues may not be pronounced in the short term, environmental implications of industrial projects should be uppermost in the minds of policy-makers. It is therefore timely to draw attention to the need for assessing these implications with a view to devising appropriate regulatory measures to address environmental concerns as the impact of industrialization begins to be felt. Policy measures to combat desertification are already in place in some affected countries, while environment-related equipment and materials for pollution reduction and waste treatment have also been introduced in other countries. Whether these attempts are adequate to arrest environmental degradation, or other measures are required at the policy level either in the form of economic incentives and/or in the form of adoption of cleaner technologies without unduly hurting the competitive position, will require a detailed assessment. It is important that African Governments are aware of the challenges ahead and prepare themselves in ensuring to get the right mix of integration between environmental and industrial policy.

Financial Sector Development

- 393. The private sector and the business community rely not only on trained and skilled manpower, but also access to development banking institutions and financial intermediaries for long-term investible funds and working capital. Although mobilization of development finance is the responsibility of the private sector promoters, the establishment of the institutional framework must, in the early stages of national development be promoted and supported by governments. Financial markets are essential for financial mobility required by both domestic and foreign investors. Industrialization will continue to be hampered without the necessary financial institutions and services, as well as specialists to operate them.
- 394. It should be noted that Africans with the requisite knowledge and experience in financial administration and management, including stock market operations, are found in many world centres of finance and in international organizations. It is strongly recommended that African governments devise ways and means of appealing to these national experts and specialists and to encourage them to participate in the struggle for economic independence. The experience of the Republic of Korea in the seventies shows that this is a viable option provided African governments and their public officials are ready and willing to provide the necessary conditions and incentives.

Physical Infrastructure Development

- 395. One of the most important preconditions for industrial development which was highlighted by the IDDA II Programme is physical infrastructure support services. They form part of the general environment to be provided by governments and the public sector. Hardly would any investor, local or foreign, venture into a country or a region where physical infrastructure has not been developed. It is recommended that governments continue to devote a good portion of their development budgets to building and maintaining infrastructural facilities upon which all industrial activities will depend.
- 396. The importance of physical and institutional infrastructure to economic and industrial development has been dealt with elsewhere in the report. What is being emphasized here is the need to accord a high priority to physical infrastructure, particularly transport and communications, water, energy and power generation and distribution, and telecommunications which are the essential dynamic instruments for industrialization. The costs for providing these services are extremely high, but many countries are already benefiting from the World Bank and other international agency support and bilateral donor assistance in this area. While bilateral donors are reluctant to put funds into direct industrial processes such as manufacturing, which they consider to be the province of the private sector, their support in building industrial preconditions is desirable and must be encouraged. Assistance from the UN system for preparing studies and development packages in the area of physical infrastructure for donor consideration and financing should be vigorously pursued. Similar assistance should be sought in strengthening government capacity for quick disbursement of development assistance funds.

Governance, Political Liberalization and Commitment

397. Africa's industrialization will not make any headway without political leadership, determination and commitment. These and good governance are the guiding principles and the driving force behind all known industrial success stories. Weak and indecisive leadership, corrupt practices, constant changes in national policies, negative attitude and lack of work ethics, divisive tendencies in public appointments and failure to recognize and reward success have all contributed to Africa's lack of potency for advancement. Development is like a country at war and calls for collective responsibility of all citizens;

but it requires dedicated and selfless political leadership which is ready to stimulate and arouse national consciousness for the common task of nation-building.

- 398. The First Industrial Development Decade failed not necessarily because the Programme was imposed on African member states as it is often alleged, or else the Second Development Decade Programme the design and preparation of which was the responsibility of governments themselves would have shown signs of success.
- 399. Structural Adjustment Programmes which are also alleged to have been imposed like bitter pills or reluctantly accepted by the many African governments now appear to be embraced and yielding some results. It may therefore be said that initiatives and programmes fail essentially because of the absence of political will and commitment to embrace them, rally the forces of society behind them and call for sacrifices as may be required for their implementation while political leadership leads the way by precept and example.
- 400. If the recommended limited scope of the IDDA II Programme can be successfully implemented by the end of the Development Decade, it would be an achievement upon which the industrialization superstructure could be built. But it is a task that goes far beyond the adoption of conference resolutions. It is a task that calls for concerted national action led by political leadership whose overriding concern and preoccupation is Africa's economic emancipation when self-reliance and self-sustainment will no longer be a dream but a reality of African life.

Alliance for Africa's Industrialization

- 401. Although there is no experience with the above programme, the evaluation attempted to establish a relationship between the IDDA and the Alliance.
- 402. It is recognized that the African private sector is still weak and ineffective to play a leading role as a development partner in the industrialization process. It has been recommended that governments devise policy measures and provide institutional and other support services to enhance the capacity and capability of the private sector, and the industrial community thereby promoting confidence, dialogue and self-assertion. The private sector is also required to organize itself at national levels and provide competent and effective leadership that would command respect and clout not only vis-a-vis their own governments and public service operators, but also in the development of partnership relationships with counterparts at sub-regional, regional and international levels. It is primarily to support and strengthen these efforts that the Alliance for Africa's Industrialization was proposed and launched in October 1996. The Alliance is therefore seen not as a replacement of the IDDA II Programme but rather a supplementary mechanism to existing government support services and promotional efforts by helping to mobilize additional financial resources, arranging technology acquisition and transfer of higher management skills, exploring external market opportunities for African industry, and arranging the formation of partnerships between African and foreign-based enterprises as a means of promoting quality upgrading, product design and packaging, etc. The Strategic Management of Industrial Development (SMID) approach used in a number of developing countries by UNIDO has proved itself to further the dialogue between the public and private sectors and therefore, should be more widely applied by African Countries.
- 403. The Alliance is a mechanism focusing the attention of African decision-makers, both public and private and the international community on the region's industrial development potential and the need to build on this potential. The focus should be on enterprise to enterprise co-operation. The initial focus of the Alliance will be agro-industrial linkages. This is fully in line with the importance of agriculture

in African economies, the potential for international comparative advantage of agro-based production, the multiple linkages between the two sectors and their potential socio-economic impact through small-and medium-size enterprises and informal sector activities and with the evaluation proposal for IDDA II. The Alliance programme will eventually extend to other productive areas of private sector activity such as natural resource-based mineral processing industries once the previous starting base is consolidated.

- 404. The Alliance for Africa's Industrialization will be demand-driven and its programme will be expected to create yet another opportunity to make a contribution to Africa's economic and industrial transformation and an effective integration into the globalized industrial economy characterized by free flows of investment, trade and technology. By this approach the Alliance will be facilitating partnerships through dialogue and negotiations between African industry and governments on the one hand, and companies, governments and NGOs of industrialized countries on the other. Being essentially private sector-led initiative, the Alliance is expected to arouse the enthusiasm and support of development partners in industrialized countries whose own private sector operators are also beneficiaries of the Alliance programme. African private sector and business community will stand to gain in confidence and stature if they and their governments will courageously accept the Alliance as their own mechanism working for their own interest in partnership with external sources of finance, technology, markets and higher managerial skills. In short, the Alliance should be used as a tool to reach the goals of IDDA.
- 405. The above provides only a relationship between IDDA and the Alliance. During their implementation, national governments, with eventual support from the participating organizations, should ensure that IDDA and the Alliance are complementary.

Annex 1

Revised version

Terms of Reference

IDDA PROGRAMME EVALUATION

by the

Secretariats of UNIDO, ECA and OAU

I. INTRODUCTION

The Conference of African Ministers of Industry at its twelfth meeting held in Gaborone, Botswana in May/June 1995 in its resolution 1 (XII) requested African countries, the Economic Commission for Africa, the United Nations Industrial Development Organization and the Organization of African Unity to undertake in 1997 a mid-term evaluation of the implementation of the programme for the Second IDDA at national, subregion and regional levels and prepare country, subregional and regional reports to its next meeting.

The Fourteenth Session of the Industrial Development Board (26-30 June 1995) adopted decision IDB.14/Dec.16 to conduct an in-depth evaluation of the IDDA Programme and its activities. The IDB also called for updating the national, sub-regional programmes within the framework of the IDDA II programme of activities. The envisaged work of updating these national and sub-regional programmes will be harmonized with the activities planned under the scope of this proposed in-depth evaluation exercise of the IDDA II programme.

Also in response to the call by the 12th meeting of the Conference of Ministers of Industry, the OAU, the UN General Assembly and the legislative bodies of UNIDO to focus attention on Africa, UNIDO has initiated an "Alliance for Africa's Industrialization". This Alliance was officially launched in Abidjan, Cote d'Ivoire on 23 October, 1996.

The findings, conclusions and recommendations of the evaluation will contribute to the re-orientation and more effective implementation of the IDDA incorporating the Alliance for Africa's Industrialization as a mechanism for the attainment of the goals and objectives of the IDDA. This will involve establishment of country priorities, identification of a mechanism/strategy for the formulation of integrated programmes of technical cooperation activities, the modalities of their implementation and mobilization of financial resources.

A programme of the mid-term evaluation will be undertaken jointly by UNIDO, OAU and ECA. The evaluation will be forward looking. It will determine what follow-up actions are required and what lessons can be drawn for future use at a preparatory, operational, organizational and policy level.

II. BACKGROUND INFORMATION

A. Lagos Plan of Action

Adopted in April 1980 at Lagos, Nigeria by the Assembly of Heads of State and Government of the Organization of African Unity (OAU), the Lagos Plan of Action accords a major role to industrialization, in view to its impact on meeting the basic needs of the population, ensuring the integration of the economy and the modernization of society. In order to accelerate the achievement of these objectives, the Lagos Economic Summit adopted the years 1980-1990 as the Industrial Development Decade for Africa. At its thirty-fifth session the General Assembly of UN reinforced the decision by adopting resolution 35/66B which proclaimed the 1980s as the Industrial Development Decade for Africa (IDDA).

B. First IDDA

On the basis of the deliberations during the Assembly of the OAU in Lagos, the IDDA Programme was prepared by UNIDO, ECA and OAU as an instrument to translate the Lagos Plan of Action into a programme for industrialisation to be pursued by African nations with the assistance of the international community. It was issued as "A Programme for the Industrial Development Decade for Africa" (ID/287) in 1982. This document set out a detailed framework for the formulation and implementation of the Programme, guidelines for the formulation of the Programme.

The Eighth CAMI held at Bujumbura (Burundi) in September 1986 called for an independent mid-term evaluation. It requested UNDP to finance the evaluation which responded positively and approved a project (DP/RAF/88/034) amounting to US\$ 292 000. An independent six-member team commenced work in August 1988 and the report was issued in April 1989 (DP/ID/SER.B/645). The report was presented to the Ninth CAMI held at Harare, Zimbabwe in May 1989 through the meeting of the Intergovernmental Committee of Experts of the Whole on Industrialization in Africa. The CAMI took note of the evaluation which highlighted the limited results of the IDDA Programme.

C. Second IDDA (1993-2002)

In September 1986, the Eighth CAMI recommended the proclamation of the Second IDDA to accelerate the industrialization of Africa. The General Conference of UNIDO in its resolution GC 4/Res. 8(November 1991) and the GA in its resolution 47/177 (December 1992) endorsed the need for the second IDDA, covering the period 1993 to 2002. The Framework and Guidelines for the preparation of the second IDDA Programme (PPD. 161) were prepared jointly by UNIDO, ECA and OAU in 1990. This framework emphasized that the basic objectives of the first IDDA remained valid. In this respect the framework categorized the proposed programmes and activities into four distinct groups, i.e. consolidation of programmes, new investments, support programmes and special programmes for LDCs. In reflecting on these groups of activities, it draws on previous work undertaken by ECA, UNIDO, OAU and other UN organizations. While highlighting key ares, it emphasizes the need for analyses and reviews of the situation at the country level. Subsequently, the interim period (1991-92) before the start of the second Decade in 1993 was used to prepare national and sub-regional programmes.

III. THE PROGRAMME FOR THE SECOND IDDA

A. Evolving Background

It must be recalled that since the adoption of the programme for the Second IDDA a number of important initiatives have been adopted by African member States. The most important of these include:

1. The Treaty establishing the African Economic community:
The African Economic Community, signed in Abuja, Nigeria 3rd June 1991 which is aimed at promoting economic, social and cultural development in Africa is a significant step forward towards economic and industrial integration in Africa.

The purpose of promoting industrial development will, among others, harmonize the industrial policies of the Member States, create a solid basis for industrialization, promote collective self-reliance, modernize the priority sectors and establish joint industrial development projects at regional and continental levels as well as the creation of African Multinational enterprises in priority industrial sub-sectors.

2. The Cairo Agenda for Action:

The Cairo Agenda for Action for relaunching Africa's economic and social development adopted on 26-28 June 1995, reaffirms that Africa's development is first and foremost the responsibility of our government and people.

Africa is determined to lay a firm foundation for a human-centered, equitable and sustained development on the basis of sound economic policies social justice and collective self-reliance in order to achieve accelerated structural transformation of Africa's economies. To this effect, industrial development is centered to structural change and transformation of Africa's economies, to the increase in incomes and employment and to the diversification of exports.

3. The UN System-wide Plan of Action for African Economic recovery and development:

The objectives of the System-wide Plan of Action for African Economic Recovery and Development are to provide a dynamic and flexible framework for concerted and coordinated action by the United Nations system to assist Africa in achieving sustained and sustainable growth and development in the 1990s and beyond, assist in the creation of a more supportive international economic environment of economic growth and social development, assist in the restructuring of African economies, help promote employment generation, channel the sectoral expertise of the United Nations organs, agencies and bodies in efforts to foster African recovery and development, and enhance global awareness of the continent's economic crisis.

The System-wide Plan was revised in 1994. The purpose of the revision of the System-wide Plan is two fold. Firstly, to draw a closer linkage between the System-wide Plan and the United Nations New Agenda for the Development of Africa in the 1990s by reflecting in the System-wide Plan the new imperatives that have emerged from the review of the New Agenda, and secondly to establish the key priority sectors that will be the focus of cooperative action by organizations of the United Nations system over the remaining period of the System-wide Plan. The revision also seeks to address the concern that gender issues were not adequately reflected in the original Plan. Thus, a gender dimension has been introduced into the revised Plan in recognition of the fact that

a more sustained development strategy requires reliance on all human resources, of which women form a major group.

4. The Alliance for Africa's Industrialization:

The Alliance for AFRICA's industrialization is a new initiative by UNIDO as an endeavour to focus the attention of African decision makers and the international community on the region's industrial development potentials and the need to mobilize these potentials through enhancing the role of the private sector and facilitating the integration in the dynamic world economy. The alliance therefore, gives primary attention to resource-based industries. The Alliance is said to be a direct follow-up of the UN System-wide Special Initiative on Africa launched by the Secretary-General. Its major objective is to focus on the industrial component of the initiative and other industry-related aspects of African development.

UNIDO intends to make the Alliance one of the pillars of its new programme (restructured). In that line UNECA should be closely associated with the development and implementation of the Alliance programme, bearing in mind the ECA/UNIDO joint sponsorship of the programme of the Industrial development Decade for Africa (IDDA-II), 1993-2002.

The salient elements of all these programmes must be taken into account in the evaluation of the IDDA.

B. THE APPROACH

Both the first and second IDDA programme made an attempt to design and construct internal engines of growth (national and multinational) by providing a common framework for guiding African countries, individually and collectively, in formulating programmes for the Decade within the overall context of their own national development plans. In order to achieve the long, medium and short-term development objectives of the programme, African countries were urged to develop national, sub-regional and regional plans and programmes.

The strategic approach for the Second IDDA has the following components:

- 1. optimizing inter-linkage between industry and agriculture.
- 2. addressing the issue of poor performance of existing industrial investment s plan for rehabilitation and need to improve public enterprise performance
- 3. Strengthening African entrepreneurial base
- 4. Underlining the importance of the informal and small scale sectors
- 5. putting new emphasis on Market orientation
- 6. Stimulating industrial culture
- 7. Promoting subregional and regional cooperation
- 8. Addressing environmental issues

The action oriented programme for the Second IDDA encompasses (i) consolidation programmes, (ii) industrial expansion programmes (iii) promotion of small and medium industries and entrepreneurship development and (iv) support services. The programme also spels out modalities for its financing including, in particular, domestic resources, foreign investment and foreign aid.

The programme for the Second IDDA based primary responsibility on member states. Thus, a more realistic view was taken of the priorities, projects and programmes which were built from the grass roots level on the basis of individual national programmes. However, the programme for the Second IDDA was also conceived as a participative exercise allowing many contributing parties such as government, policy makers, private sector actors, financial and banking institutions, national and international institutions as well as non governmental organization.

IV. PURPOSE, SCOPE AND METHODS OF EVALUATION

A. Purpose

To enable African Member States, UNIDO, OAU and ECA to make objective assessment of the achievement of the Programme against set objectives and to make informed decisions on the future course of action.

B. Scope

The evaluation will:

- assess the achievement of the Programme against its objectives and expected outputs at the national, sub-regional and regional levels
- identify and assess the factors that have facilitated the achievements of the Programme's objectives, as well as those factors that have impeded the fulfilment of those objectives
- propose a course of action, both at policy and programme implementation level and to draw lessons thereof for the design, implementation and monitoring of the Programme for the future.

In order to achieve the purpose in Section III A and cover the scope in Section III B, the evaluation will address the following specific aspects in Sections III C - III E.

C. Evolving Circumstances and relevance

domestic macro-economic framework including SAPs, private sector participation etc.

- . major trends in the global economy affecting the industrialization process in Africa, in particular impact of the Uruguay Round of Multilateral Trade Negotiations
- . continued relevance of the objectives of the IDDA II Programme; in the evolving domestic and global context;
- . Complementarity of IDDA activities with Africa's other programmes in particular the Cairo Agenda of action, The Treaty and the System-wide Initiative for Africa;
- complementarity between the "Alliance for Africa's Industrialization" and the IDDA II Programme, with a view to determine on how the Alliance can contribute to the attainment of the IDDA II programme objectives;
- . the relevance of the Programme activities planned, under implementation, or completed in relation to IDDA objectives and targets;
- the extent to which on-going and planned programmes have been planned in accordance with priorities established in the IDDA, especially food and agro, building, metallurgical, mechanized, electric and electronics, chemical, forest and energy industries;
- the extent of participation of the countries (governments and private sector), in the formulation and implementation of programme activities;
- the relevance of the objectives of the programmes of the Secretariats of UNIDO, ECA and OAU.

D. Attainment of goals and efficiency assessment

- the degree to which the programme activities produced the planned outputs at subregional and national levels and factors which facilitated the production of outputs, as well as those factors that impeded planned actions, in particular:
 - the degree of co-operation and co-ordination between OAU, ECA, UNIDO and the Conference of African Ministers of Industry in the planning and implementation of the programme,
 - (ii) the effectiveness of the mechanisms, networks and linkages, etc. expressly set up by the IDDA organizing bodies to promote the Decade;
 - (iii) the extent to which the contributions of the institutions of the United Nations system, other international organizations and

African organizations have been coordinated to support joint IDDA programmes,

- the degree of compliance with procedures as indicated in the project cycle management system for the identifications, design, approval, implementation, monitoring and evaluation of projects under the IDDA Programme.
- the role of UNIDO Secretariat and UNIDO field offices, ECA and ECA MULPOCs and OAU in the implementation of the projects,

E. Assessment of Impact of Activities

To assess the impact of the IDDA activities on specific issues specified in the 16th meeting of the Joint Committee of the OAU, ECA and UNIDO secretariats on the implementation of the Programme for the Industrial Development Decade for Africa (IDDA), Addis Ababa Ethiopia 27-29 March 1996 (Annex I of the Report). The evaluation should also examine the forward looking issues that will make the implementation of the IDDA more effective. The role of Regional Economic Communities (RECs) in the implementation of the IDDA should be fully assessed and recommendations made therefor. In this respect, the evaluation should focus on the following:

1. Industrial Competitiveness

- Examine the extent to which the IDDA Programme has supported the development of entrepreneurial capabilities;
- Strengthen and suggest ways of building critical capacities, especially technical skills at the member states level so as to rapidly adjust to the market demands imposed particularly by the ISO 9000 and ISO 14000 family of quality standards in conformity to the Uruguay Round Agreement;
- Promote the exploitation of Africa's comparative advantage, particularly its human resources and vast natural resources
- Gender main-streaming in manufacturing activities, so as to harness the productive capacity of 50% of the African population;

2. Human Resources for Industrial Development

Areas of focus

The role of the government and the private sector enterprises in the elaboration of national training programmes to make them more development related

- Extent to which member states, UN bodies and other development partners ensure the viability and sustainability of Africa industrial training organizations such as ARCT, ARSO and ARCEDEM

3. Industrial Private Sector Development

- Extent to which the private sector especially the SMIs and SSIs are promoted and given the required support
- The quality and support given to strengthen African institutions in the areas of research and development (R&D), with special emphasis on relations between universities, research institutions and industries
- The role of the State and the private sector in the formulation and implementation of national industrial policies and strategies

4. Mobilization of Financial Resources for Industry

- Identify domestic and foreign sources for financing programmes/projects

5. Regional Co-operation

Areas of focus

- The extent to which support and promotion is given to promote the industrial activities of African sub-regional and regional economic communities
- The support and strengthening of African regional and sub-regional technological and scientific institutions
- Enhance opportunities for intra-African trade and cross-border investments in industrial and technological activities

6. International Co-operation

Areas of focus

- Forge alliances with the international community to promote Africa's industrialization
- Flow of domestic and FDI into industrialization
- Flow of ODA through bilateral and multilateral sources

- Mechanisms for project identification and promotion
- Investment promotion activities by ECA, UNIDO and OAU
- Promotion of information technology that enhance acquisition of technologies

F. Method

IDDA programme evaluation will be divided into three phases:

- i. The conceptual stage (desk review);
- ii. Field missions, and,
- iii. Preparation of the final evaluation report.

FIELD MISSIONS

COUNTRIES	MISSION TEAM	TIMING
Zimbabwe Botswana Namibia	Mr. Adusei-Poku - UNIDO consultant Mr. L. Buzingo - OAU Mr. N. Biering, UCD, Zambia (UNIDO) Mrs. Fall - ECA	19/1 to 23/1/97 24/1 to 29/1/97 30/1 to 2/2/97
Ghana Nigeria Ethiopia	Mr. Adusei-Poku - UNIDO consultant Mr. Ugbor- UNIDO s/m Mr. Barry - ECA	3/2 to 6/2/97 6/2 to 12/2/97 12/2 to 15/2/97
Kenya Uganda Tanzania Zambia Ethiopia	Mr. G. Kopolo - UNIDO s/m Mr. M'wango - ECA Mr. J.O. Mowencha - ECA	16/1 to 19/2/97 5/2 to 10/2/97 11/2 to 15/2/97 1/2 to 4/2/97 20/2 to 24/2/97
Vienna Cameroon Congo Gabon Côte d'Ivoire	Mr. G. M'Bengue - UNIDO consultant Mr. Demba - ECA Mr. Bela - ECA Mr. Danso - UNIDO (UCD)	14/1 to 15/1/97 20/1 to 23/1/97 23/1 to 27/1/97 27/1/97 to 30/1/97 30/1 to 2/2/97
Mali Sénégal Guinea Niger Ethiopia	Mr. G. M'Bengue - UNIDO consultant Mr. Addu - ECA Mr. Fukori - ECA Mr. L. Buzingo - OAU Mr. Sako - UNIDO (UCD)	9/2 to 12/2/97 13/2 to 16/2/97 17/2 to 20/2/97 21/2 to 24/2/97
Egypt Tunisia Morocco	Mr. O. Gonzalez - UNIDO s/m Mr. Kouroma - ECA Mr. Kamara - ECA	20/1 to 24/1/97 27/1 to 28/1/97 29/1 to 31/1/97

j	Addis Ababa
Finalization of Regional Reports	24/2/97 - 7/3/97

	Vienna
Finalization of Evaluation Report	8/3/97 - 31/3/97

National Programmes for the Second

Industrial Development Decade for Africa (IDDA)

prepared for the following countries:

Benin

Botswana

Burkina Faso

Burundi

Cameroon

Cape Verde

Central African Republic

Chad

Comoros

Congo

Côte d'Ivoire

Ethiopia

Gambia

Ghana

Guinea

Guinea-Bissau

Kenya

Lesotho

Madagascar

Malawi

Mali

Mauritania

Mauritius

Mozambique

Niger

Nigeria

Sénégal

Seychelles

Sierra Leone

Swaziland

Tanzania

Togo

Uganda

Zaire

Zambia

Zimbabwe

Annex 4.a

List of IDDA Technical Assistance (XA) Projects 1994-1995 Biennium

Project No.	Title	Total Allot.
XA/RAF/95/605	Seminar on Development of investment project for women entrepreneurs, Yaounde, Cameroon, 12-23 June 1995	17,000
XA/CMR/95/607	Rehabilitation of UNIDO cassava processing prototype plant	10,800
XA/CVI/94/611	Upgrading and expansion of pharmaceutical production in Cape Verde	47,309
XA/ETH/94/602	Promotion of Akaki spare parts and hand tools factory (ASPF) as a regional centre	6,502
XA/ETH/94/627	Training of regional industrial officers in decentralized industrial development strategies	52,750
XA/GAB/95/604	Assistance préliminaire à l'élaboration d'un cadre de référence du developpement industriel	75,000
XA/GAM/94/618	Training programme for Women- entrepreneurs in Food Processing Sector	85, 855
XA/GUI/94/620	Assistance préparatoire: Appui à l'entreprenariat féminin en matière d'investissement	127,500
XA/GUI/94/624	Programme development and support services in Guinea	45,000
XA/MAG/94/636	Programme pour le développement industriel écologiquement durable (DIED)	45,0 00
XA/MAR/94/619	Productivity centre for the garment industry	38,000
XA/MAU/95/602	Séminaire national sur le développement de la comptabilite analytique dans les entreprises industrielles, Mauritanie	25,500

XA/MOZ/94/601	High-level advisory services for the promotion of the industrial development programme	104,000
XA/NIR/94/625	Feasibility study for the extension of the Cement Company of Northern Nigeria, Sokoto, Nigeria (CCNN)	8,660
XA/PRC/95/609	Assistance à la gestion stratégique du plan integré de développement agro-industriel (PIDA)	84,000
XA/RAB/94/628	Workshops on the enhancement of national project execution capabilities through the development of industry-related & consulting services in Tunisia, Algeria & Morocco	141,600
XA/RAF/94/606	Feasibility study for the establishment and operation of a national tissutheque in West African LDCs	52,023
XA/RAF/94/607	The application of biogas technology to the treatent of industrial wastes in selected African countries	9,287
XA/RAF/94/608	Assistance à l'industrie du cuir (Phase II)	145,360
XA/RAF/94/610	Etude des moyens de renforcement de la cooperation technique dans le domaine des industries de la pêche entre les pays africains riverains de l'Ocean atlantique	127,600
XA/RAF/94/614	Sechage solaire des fruits et legumes au Burkina Faso, en Guinee Conakry et au Mali	7,315
XA/RAF/94/616	Promotion de la cooperation industrielle dans la sous-region de l'Ocean indien	6,215
XA/RAF/94/618	Etude de faisabilité en vue de l'expansion d'une unite industrielle integrée et proposition de régionalisation du centre national d'etudes et d'experiment de machin. agric.	6,914
XA/RAF/94/619	Development and Dissemination of appropriate food processing equipment for rural women in Sub-Saharan Africa	502,717
XA/RAF/94/621	Assistance à l'organisation pour la mise en valeur du fleuve Senegal	8,862

XA/RAF%94%623	Assessment of the solar and wind energy utilization in Africa	30,153
XA/RAF/,947,629	Programme sous-regional de mobilisation des ressources d'investissement pour la création et/ou la rehabilitation des PME/PMI (2eme phase)	30,000
XA/RAF/94/.630	Prog. sous-regional "de mobilisation de ressources d'investissement" pour la création et/ou rehabilitation de PME/PMI dans des pays moins avancés (PMA)	9,300
XA/RAF/94/633	Improvement of management performance through applications of computers and training for SMEs (IMPACT)	178,500
XA/RAF/94/634	Production du materiel didactique et formation de formateurs dans le domaine de l'emballage et le conditionnement des produits agroalimentaires, Casablanca, Maroc, 16/10-3/11/95	204,600
XA/RAF/94/635	INTIB/AFRICA ind. & technological info. network, phase II. Install. of INTIB/IRMS standard database & reg. seminar on Data MNGMT and networking, Dakar, Senegal	121,000
XA/RAF/94/638	IDDA technology for dev. prog. (6th African Ties Meeting, Nov. 1995)	260,000
XA/RAF/94/639	Assistance to the PTA leather and leather products institute (LLPI). (Training course: Addis Ababa, Ethiopia, 1st QRT 1995)	94,000
XA/RAF/94/640	Waste minimization demonstrations and policy advice in four African countries	233,100
XA/RAF/94/642	Demonstration pilot programme for the design, manufacturing and testing of hydraulic maintenance presses, IBADAN, Nigeria, 13/3 - 19/5/95	303,500
XA/RAF/94/643	Regional SME development programme for Africa	551,400
XA/RAF/94/644	Follow-up to industrial sector programme review missions in Africa	750,000

XA/RAF/95/601	Reg. workshop on ind. policies & strategies. The experience of Rep. of Korea and its relevance to African countries; Seoul, 14-23/9/95	140,688
XA/RAF/95/603	Demonstration pilot prog. for the design, manufacturing and testing of hammer mills; Ibadan, Nigeria, Aug. 1995	332.000
XA/RAF/95/610	Regional Africa Leather and Footwear Industry Scheme training, Thika, Kenya, 1) cutting/stitching, Aug.'95, 2) designer/pattern making, Oct/Nov.'95	143,960
XA/RAF/95/611	Rehabilitation and establishment of finished articles manufacture (REFAM)	145,000
XA/RWA/94/602	Amelioration des capacites de recherche et developpement a l'OPYRWA pour la mise au point d'une technique de raffinage du pyrethre produit	33,053
XA/SEN/94/613	Etude technico-economique des capacites et des besoins de l'ind. halieutique du Senegal pour la mise au point d'une gamme de nouveaux produits	34,648
XA/STP/94/626	Assistance à l'organisation d'une table ronde sur le développement industriel à Sao Tomé et Principe	42,000
XA/SUD/95/613	P.A. for the industrial utilization of medicinal and aromatic plants in Sudan	40,300
XA/SWA/94/622	Advisory support in the dev. of an export processing zone (EPZ) in Swaziland	41,500
XA/TOG/94/617	Assistance préliminaire à la formulation de programmes d'actions dans le cadre d'une gestion stratégique du developpement industriel	72,000
XA/UGA/95/608	Handing over of a system to develop small and cottage industries at grass-root level to the nataional authorities	39,500

XA/ZAM/94/631	National Industrial Statistics Programme (NISP)	121,050
'XA/ZIM/94/609	Feasibility study for the establishment of a citrus fruit and tomato processing plant in Zimbabwe	61,968
XA/RAF/95/612	Production of SME promotion tools	117,000

Annex 4.b

List of IDDA Supplementary Activities (YA) Projects 1994-1995 Biennium

Project No	Project Title	Total Allot.
AD HOC BY DG	EGM on Women in Food Processing Industries, Lusaka, 17-20/01/94	66,500
YA/RAF/95/X31	Participation of Women Entrepreneurs in the Private Sector Forum	10,800
AD HOC BY DG	Industrial Automation/Quality Management	5,100
YA/RAF/94/X27	Workshop on the impact of the devaluation of CFA Franc on the manufacturing sectors of the countries of UEMOA, Bamako, Mali, 25-27 April 1995	48,900
YA/RAF/94/001	IDDA Programme Development	40,000
YA/RAF/94/X37	UNIDO/OAU/ECA Joint Committee on IDDA	8,100
YA/RAF/94/X35	Industrial Sector Programme Review Missions	73,800
YA/RAF/94/X46	Meeting of the 26th Ordinary Session of the OAU	3,400
YA/RAF/94/X54	Symposium on the role of NGOs in the implementation of UN-NADAF, Dakar, Senegal, 9-11 November 1994	8,700
YA/RAF/94/X55	IDDA Inter-Agency Coordination	7,400
YA/RAF/94/X60	Private Sector Seminar (ABID) in Cote d'Ivoire	51,200
YA/RAF/95/X03	Bureau of CAMI/Committee of 10	19,700
YA/RAF/95/X08	Consultation with South Africa in the preparation of an integrated programme and in support of UNIDO/South Africa cooperation (COMFAR for the savings)	94,600

YA/RAF/95/X11	Arrears of the tenth meeting of the Conference of African Ministers of Industry	36,500
YA/RAF/95/X13	Arrears of the 11th meeting of the Conference of African Ministers of Industry, Port Louis, Mauritius, 31/5-3/6/93	4,000
YA/RAF/95/X16	Implications of the Uruguay Round Preparatory Assistance	21,600
YA/RAF/95/X21	Industrial sector programme review missions (Cameroon, Central African Republic, Chad, Ethiopia, Gabon, and Madagascar)	77,500
YA/RAF/95/X23	UNIDO/ECA cooperation	9,000
YA/RAF/95/X22	Industrial development reviews of Angola, Cote d'Ivoire, Ethiopia, Ghana, Kenya, Mozambique, Tanzania and Uganda	200,000
YA/RAF/95/X26	Twelfth Meeting of the Conference of African Ministers of Industry, 6-8 June 1995, and preceding meeting of Inter-Governmental Committee of Experts, Gaborone, Botswana	40,000
YA/RAF/94/X40	UNIDO/ADB cooperation	39,800
YA/INT/94/X40	An in-depth evaluation of the Industrial Partners Programme for Africa	22,900
YA/RAF/94/X39	Workshop on the impact of the devaluation of CFA Franc on the manufacturing sector of countries of CEMAC, Brazzaville, 28/11-1/12/94	99,300
YA/RAF/94/X28	Production of a film for the Africa Industrialization Day 1994	77,000
YA/RAF/94/X34	Industry Africa/Afrique Industrie (issues 8&9)	83,500
YA/RAF/94/X50	Four posters for Africa Industrialization Day	18,000
YA/RAF/95/X20	Video production on human resource development	95,000
AD HOC BY DG	Privatization workshop for the French-speaking LDCs, Bissau	69,000

711/7/11			
YA/RAF/94/X61	Preparation and edition of a manual on diagnostic and restructuring of agro-based industrial enterprises in Africa	96,000	
YA/RAF/95/X25	Development of an integrated UNIDO support programme in the agrobased industries	178,000	
YA/RAF/94/X56	Diversification of African industries	40,900	
YA/RAF/95/X24	Private Sector Forum	292,400	
YA/RAF/95/X29	Development of a regional privatization programme	219,000	
AD HOC BY DG	Nutritional survey in Tanzania	3,100	
YA/RAF/94/X58	EGM on the utilization of wastes and by-products in the meat and fish industries in Africa	138,500	
AD HOC BY DG	Workshop on the health care technology in the Sub-Saharan region	1,800	
YA/RAF/94/X15	Application of the GMP Requirements in the pharmaceutical industry	25,000	
YA/RAF/95/X15	Promotion of Artisanal Mining in Africa: Participation in International Roundtable on Artisanal Mining, World Bank, Washington D.C., 17-19/5/95	25,100	✓
YA/RAF/95/X19	Regional Workshop on Development Strategies for Fishing Systems	125,000	✓
AD HOC BY DG	PTA Investor's Forum, Lusaka, 18- 21/01/94	60,400	
YA/MOR/94/X07	Forum pour la promotion de projets d'investissement au Maroc	67,000	
YA/RAF/94/X45	Investment Promotion Programme, including Investment Forum for selected countries	221,500	
YA/RAF/94/X08	Investor's Forum for Ghana	120,000	····
YA/RAF/94/X43	Programme sous-régional de mobilisation des ressources d'investissement pour la création et/ou la rehabilitation des PME/PMI (2eme phase)	115,000	_
YA/RAF/94/X22	TECHMART PTA	120,000	

YA/RAF/95/X04	Symposium on the development of partnership programmes in the dairy industry: from research to industrial operation, Juiz de Fora, Minas Gerais, Brazil, Abidjan and Cote d'Ivoire	91,800
YA/RAF/95/X27	Study on the Automotive industry in Africa	113,000
YA/RAF/95/X32	Country level assessment of the impact of the Uruguay Round agreements on the export supply capabilities in Africa 122,000	122,000

List of IDDA Technical Assistance (XA) Projects Approved for the 1996-1997 biennium as at 28 February 1997

Project no.	Project title	Budget
XA/ALG/96/622	Elaboration d'un programme intégré d'appui et d'assistance à la restructuration industrielle et au redressement des entreprises industrielles en Alger	90,000
ХА/DJI/96/621	Etude de faisabilité portant sur l'exploitation industrielle de la perlite dans le gisement d'egerlayta ainsi que de ses sous-produits	96,700
XA/GAB/95/604 XA/GAB/96/604	Assistance préliminaire à l'élaboration d'un cadre de référence du développement industriel	3,498
XA/GAM/96/624	National Industrial Statistics Programme (NISP)	120,000
XA/GHA/96/639	High-level advice to the Ministry of Lands and Forestry on coordination of technical support for the forestry master plan	24,000
XA/GLO/96/043 XA/GLO/96/636	Facility to implement training programmes on preparation and appraisal of investment projects in developing countries (national level)	25,000
XA/GUI/94/620 XA/GUI/96/620	Assistance préparatoire: appui à l'entreprenariat féminin en matière d'investissement	7,500
XA/IVC/96/623	Mise à jour et finalisation des travaux du schéma directeur d'industrialisation de la Côte d'Ivoire - Phase 1	140,000
XA/IVC/96/649	Agro-machinery programme: preparatory assistance	60,000
XA/KEN/96/649	Agro-machinery programme for Kenya: preparatory assistance	60,000
XA/KEN/97/602	Jua Kali Demonstration, Training and Technology Transfer Centre	94,820
SF/MAG/94/002 XA/MAG/96/615	Programme pour le développement industriel écologiquement durable (DIED), Phase II	97,000
XA/MAG/96/601 XA/MAG/94/636 SF/MAG/94/001	Programme pour le développement industriel écologiquement durable (DIED)	52,196
XA/MAR/96/619 XA/MAR/94/619 XA/MAR/93/619	Productivity Centre for the Garment Industry	5,436
XA/MAU/96/630	Assistance d'urgence aux industries de la pêche en Mauritanie	177,000
XA/ML1/96/641	Advisory services on an action programme for the development of mango processing	36,870
XA/MI.W/96/627	National Leather and Footwear Industry Scheme (NALFIS), part of the Regional Africa Leather and Footwear Industry Scheme	361,000
XA/MOZ/96/603 XA/MOZ/94/601	High-level advisory services for the promotion of the industrial development programme	8.000
XA/PRC/96/609 XA/PRC/95/609	Assistance a la gestion stratégique du Plan Intégré de Développement Agro-industriel (PIDA)	45.700

Project no.	Project title	Budget
XA/RAB/96/628	Programme for the development of industry-related and consulting services in Morocco, Tunisia and Algeria	74,700
US/RAF/96/060 XA/RAF/96/613	1	
XA/RAF/96/611	Renforcement de la capacité institutionnelle du cafard pour organiser des programmes de formation pour les entrepreneurs et consultants/formateurs dan le domaine de la création et de la gestion des PMIS agro-alimentaires dans les pays les moins avances de l'afrique francophone	139,500
XA/RAF/96/612 XA/RAF/94/612	Production of SME promotion tools	44,359
XA/RAF/96/629 XA/RAF/94/629	Programme sous-regional de mobilisation des ressources d'investissement pour la création et/ou la réhabilitation des PME/PMI (2ème phase)	27,193
XA/RAF/96/636	A joint (UNIDO,OAU, ECA) in-depth evaluation of the IDDA II Programme	150,000
XA/RAF/96/637	Support for "Ecologically Sustainable Industrial Development" (ESID) Programme for regional Africa	89,000
XA/RAF/96/638	Développement et diffusion de matériels de transformation des produits alimentaires pour les femmes en milieu rural - Phase II	206,700
XA/RAF/96/640	Consultative meeting of african IPAs and UNIDO-IPSS	62,500
XA/RAF/96/643 XA/RAF/94/643	Regional small and medium enterprise (SME) development programme for Africa	221,714
XA/RAF/96/645	AF/96/645 Needs assessment for establishing an effective information network system in African LDCs: Phase I	
XA/RAF/96/646	Investment, technology and information programme for selected English-speaking African countries	1,768,500
XA/SIL/96/614 XA/SIL/95/614		
XA/TOG/96/616	Assistance to the government of Togo in the field of export processing zones 71	
XA/7,AM/96/626	National Leather and Footwear Industry Scheme (NALFIS) part of the regional Africa Leather and Footwear Industry Scheme (RALFIS)	
XA/ZAM/96/631 XA/ZAM/94/631	National Industrial Statistics Programme (NISP)	63,864
XA/ZIM/96/632	National Industrial Statistics Programme (NISP)	117,500

Project no.	Project title	Budget
YA/ANG/96/X01	Etude sur la mise en valeur des ressources humaines pour le développement des capacité de management et d'entrepreneuriat - IDDA supplementary activity	25,160
YA/ANG/96/X02	High level assistance for the modernization of the pharmaceutical sector in the framework of the reconstruction of the national economy - IDDA supplementary activity	
YA/BDI/96/X01	Consultation/Programme formulation mission to Burundi	41,500
YA/BDI/96/X02	Participation of women entrepreneurs at the first global women entrepreneurs trade fair and investment forum, Accra, Ghana, 26 June - 3 July 1996	11,000
YA/CMR/96/X02	Diagnostic study for introducing new technologies for mercury abatement in Cameroon - IDDA supplementary activity	4,000
YA/CMR/96/X07	Forum pour la promotion de projets d'investissement au Cameroun - IDDA supplementary activity	165,000
YA/ETH/96/X08	First conference of african ministers responsible for sustainable development and environment; and first meeting of the committee of experts, 4 to 9 March 1996	2,882
YA/IVC/96/X08	Forum de l'investissement pour la Côte d'Ivoire	170,000
YA/KEN/97/X01	Programme preparation for implementation of sessional paper on industrial development: Kenya	80,000
YA/RAF/95/X53 YA/RAF/96/X53	, , , , , ,	
YA/RAF/96/X09	High-level consultation/programme formulation mission to Eritrea - IDDA supplementary activity	19,000
YA/RAF/96/X11	/RAF/96/X11 Bureau of CAMI and joint UNIDO/ECA/OAU Committee on the IDDA - IDDA supplementary activity	
YA/RAF/96/X12	Assistance to African countries in the revision/preparation of their national programmes for the second IDDA	310,000
YA/RAF/96/X14	Assistance to the East Africa Confederation of Informal Sector Organizations (CISO) in Arusha/Tanzania - IDDA supplementary activity	46,492
YA/RAF/96/X21	UNIDO/OAU cooperation	17,150
YA/RAF/96/X23	Study on industrial development and preparation of industrial development strategy	
YA/RAF/96/X24	Agro-related industrial development in Africa enhancing capacity-building and competitiveness based on potential comparative advantages	95.000

Project no.	Project title	Budget
YA/RAF/96/X25	Promotion campaign for African Industry Day	166,500
YA/RAF/96/X30	Development and launching of the "Alliance for Africa's Industrialization"	256,000
YA/RAF/96/X35	Ecoworld '96 - Industry and the Environment in Africa	21,100
YA/RAF/96/X46	Investment Promotion Programme, including an investment forum for selected Southern African countries	
YA/RAF/96/X52	Assistance to the UDEAC in the preparation of an integrated industrial policy and programme for the Central African subregion	82,900
YA/RAF/96/X60	Round-table of African Investment Promotion Agencies from selected francophone countries	
YA/RAF/96/X62	Promotion of quality improvement of aromatic plant products in African LDCs, Harare, Zimbabwe, 8-10 September 1997	63,900
YA/RAF/96/X63	Development of two integrated packages of technical cooperation projects in the field of investment, technology and industrial information for English and French/Portuguese-speaking African cities and mobilization of financial resources for their implementation	
YA/RAF/96/X64	Partnership development between the city of Rio De Janeiro and selected African countries	84,000
YA/RAF/96/X66	Thirteenth meeting of the Conference of African Ministers of Industry and preceding meeting of the Intergovernmental Committee of Experts of the whole on the industrialization in Africa, Accra, Ghana, May/June 1997	
YA/URT/96/X04	Rural technologies programme for the informal metal working sector: preparatory assistance	37,000

List of people met in the Secretariats of OAU, ECA and UNIDO

UNIDO:

Mr. JM. Deroy	Mr. K. Yumkella
Mr. A. Akpa	Mr. K. Goodspeed
Mr. F. Ugbor	Mr. S. Ndam
Mr. A. Nickels	Mr. Y. Maruno
Mr. O. Gonzalez-Hernandez	Mr. J. Navratil
Mr. I. Farooque	Ms. F. Bennani Baiti

UNIDO Advisory Committee Members:

Mr. A. Andrasevic	Mr. B. Diallo
Ms. J. Jensen	Mr. T. Roepstorff
Mr. C. Scaratti	Mr. B. van Burik
Mr. K. Vencatachellum	Mr. A. Volodin
Mr. A. Zeweri	

UNIDO Country Directors:

Mr. N. Biering	Mr. M. Cissé
Mr. A.P. Dansou	Mr. G.E. Mariki
Mr. C.T. Sakho	Mr. G. Tabah
Mr. D. Tommy	

ECA:

Industry and Human Settlements Division:

Mr. P.K. Bugumbe	, Officer-in-Charge, DMD
Mr. V. Bobrov,	DMD
Mr. Iro Addo,	10
Mr. E. Tiagha,	n
Mr. P. Demba,	n
Ms. Melkrist,	Ħ
Mr. M. Kourouma,	н
Mr. D. Kamara,	u
Mr. I. Barry,	tt

Evaluation Division:

Ms. M. Schmelzer Mr. Farazi.

OAU

Mr. L. Buzingo, Head, Industry, Energy and Mineral Resources Division

List of People Met and Institutions Visited during the Field Mission

BOTSWANA

Ministry of Commerce and Industry

S. S. Sibanda, Deputy Director, Department of Industrial Affairs (DIA)

Florence K. Matome, Senior Industrial Officer, DIA

Ministry of Finance and Development Planning

G. Kayira, Employment Coordinator

Terence F. Kelly, Economic Consultant

Department of Trade and Investment Promotion (TIPA)

Tlhabolgo Ndzinge, Deputy Director, Investment Promotion Division

Botswana National Productivity Centre (BNPC)

Boyce Sebetela, Productivity Promotions Manager

Exporters' Association of Botswana

Eitlhopha Mosinyi, President

Loago Raditedu, Director

Rural Industries Promotion Company (RIPCO)

K.V. Morei, Managing Director

Botswana Development Corporation

Michael O. Molefane, Managing Director

Boikhutso M. Disele, Deputy Managing Director & General Manager, Marketing

Southern African Development Committee (SADC)

H.-M. Stahl

The Ginger Group

Brian Newell, Botswana National Productivity Centre (BNPC)

Botswana Technology Centre

David Inger, Managing Director

UNDP

Deborah Landey, UN Resident Coordinator

Teedzani Woto, National Economist

CAMEROUN

Ministère du Développement Industriel et Commercial (MINIDC)

Justin Dioro, Ministre

Obio W. Obiosang, Directeur de l'Industrie

Ministère de l'Economie et des Finances (MINEFI)

Pierre Titti, Directeur du Budget Angos Zangue, Chef, Service des Equipements Epoupa Bosambo, Sous-Directeur des Investissements et des Opérations Isaac Jouonang, Chef, Service BIP

Ministère de l'Agriculture

Paul Félix Bangoweni, Directeur des Projets Agricoles Martin Kenfack, Inspecteur Général

Ministère de l'Elevage, des Pêches et des Industries Animales (MINEPIA)

Pierre Nanda, Directeur de la Production Animale
Jean Fezeu Dawak, Chargé des Etudes, Direction des Productions Animales
Timothé Lekedji, Sous-Directeur des Pâturages et de l'Hydraulique Pastorale

Ministère de l'Urbanisme et de l'Habitat (MINUH)

Thomas Mbondo Kanga, Directeur de l'Urbanisme et de l'Habitat Jean-Claude Mbwentchou, Sous-Directeur des Etudes et Projets

Chambre de Commerce, d'Industrie et des Mines du Cameroun (CCIM)

Saidou A. Bobbo, Secrétaire Général Hubert Amenya, Division des Etudes Edouard Nitcheu, Division de la Promotion

Société Nationale d'Investissements (SNI)

Ambroise Ondoa Onana, Directeur Général Adjoint
Bassoro Amidou, Directeur des Etudes et Projets
Efoua Mbozo'o, Directeur des Opérations
Magloire Hand Bahiol, Directeur Adjoint des Projets
Roger Mbassa Ndine, Président du Comité de Réhabilitation des Entreprises du Secteur Public et Parapublic

Centre d'Appui au Développement Alimentaire (CADA) Claude Nana Sundjo, Directeur

Direction du Développement Industriel et Commercial Laurentine Monkan

SYNDUSTRCAN

Samuel Kondo, Président

SITABAC

James Onobiono, Présidente Directeur Général Amadou Vamoulke, Conseiller du PDG

Programme Prioritaire des PME Byll Arhini, Expert National

PNUD

Aurelien Agbenonci, Représentant Résident

CONGO

Ministère du Développement Industriel
Wilfrid Jérôme Peleka, Directeur de Cabinet
Bernard Toukoulou, Conseiller à l'industrie

Felix Ngoulou, Directeur Général de l'Economie (Mini Plan)

Maconjo, Secrétaire d'Etat PME

Ngoulou, Directeur Général du Commerce

Gestion Stratégique du Développement Industriel Joseph Mounzingoula, Directeur

Jean Cristophe Boungou-Bazika, Directeur Général du Plan

Chambre de Commerce de Brazzaville Odzali, Président

Fumey, Président de UNICONGO

Union Nationale des Opérateurs Congolais Miakassissa, Président

Agence Nationale de l'Artisanat Okimi, Directeur

Forum des Jeunes Entrepreneurs Le Père La Breteche

PNUD

Dominique Aitouyahia-McAdams, Représentant Résident

CÔTE D'IVOIRE

Chambre de Commerce et d'Industrie N'Dia Georges, 1er Vice Président Komenan Mouogo, Direction de l'Industrie

Ministère du Plan et de l' Industrie Ahoua N'Doli, Ministre Délégué

CEPICI

Kofivit, Directeur Accueil Amani, Attaché Direction

Fédération Nationale des Industries Turquetil, Secrétaire Général

National Patronat Côte Ivoire Diallo, Conseiller Banque Africaine de Développement

M'hamed Amdiss, Chef, Division des Opérations Région Ouest Habib Kamoun, Analyste Financier

Chambre d'Agriculture

Kanon Seraphin, Directeur Général Adjoint Sekan Emile, Chef, Service de la Formation et Chargé des Projets

Centre Appui Secteur Privé BNETD

Sylla, Directeur

Sylain Kacou, Economiste

EGYPT

General Organization for Industrialization (GOFI)

Hassan Safwat, Deputy Chairman

M. A. Abdellatif, Under-Secretary for Foreign Agreements

M.A. Badawy, Head, Central Department for Licensing, Registration & Information Center

Y. A. Moustafa, Central Department for Foreign Agreements

UNDP/UNIDO

O Gomaa, Regional Recruitment Officer

L. Cartini, Junior Professional Officer

ETHIOPIA

Ministry of Trade and Industry

Kasahun Ayele, Minister

Ministry of Economic Development and Co-operation

Andualem Abebe, Head, Multilateral Co-operation Department

Chamber of Commerce

Kassahun Jembere, Secretary General

Ethiopian Investment Authority

Tadesse Haile, General Manager

Ethiopian Science & Technology Commission

Asrat Bulbula, Commissioner

Development Bank of Ethiopia

Bekele Tilahun, Manager, Planning & Research Department

Development Agency for Handicrafts & Small-Scale Industries

Askenachew Gebreyes, General Manager

Leather & Leather Products Institute (LLPI)

R Arunga, Director

Ethiopian Private Industries Association Berhane Mewa, President

Ethiopian Privatization Agency
Beshah Azmite, General Manager

Akaki Spare Parts & Handtools Factory Daniel Kebede, General Manager

UNDP

Lehembre, Deputy Resident Representative

GABON

Ministère du Commerce et de l'Industrie de la Privatisation

André Dieudonné Berre, Ministre

Noel Landry Posso, Directeur de Cabinet du Ministre du Commerce et de l'Industrie Fabien Mbeng Ekorezok, Conseiller du Ministre du Commerce et de l'Industrie Michel Boulingui, Chargé d'Etudes au Cabinet

M. Tidzani, Directeur Général de l'Industrie

Ministère des Finances et de l'Economie

Meye Bekourou, Conseiller du Ministre, & Secrétaire Du Comité de la Privatisation

Ministère du Plan

Albert Ovono Ndong, Directeur de la Planification Générale Masse, Conseiller Commissaire Général au Plan

Chambre de Commerce et d'Industrie

Jean François Aveyra, Vice-Président Albert Engonga Bikoro, Conseiller Président, Chambre de Commerce Ossouala, Conseiller Président, Chambre de Commerce

Confédération Nationale du Patronat Gabonais Jean Pierre Tchoua, Président

Confédération Patronate Gabonaise

Eric Messerchmitt, Secrétaire Général

Fédération Syndicale du Patronat Gabonais Jean Claude Mefe. Président

Syndicat Industriel du Gabon

François Ambourouet, Président

Office National de l'Emploi (O.N.E.)

Henri Jacob Oyono Ngomo, Directeur Général Baloghe, Conseiller Negrel, Conseiller Ondo Ossa, Coordinateur du projet prospective "Réflexion stratégique à long terme - Gabon 2025"

PNUD

Toon Vissers, Représentant Résident Didier Mebalry, Expert

GUINEA

Ministère des Affaires Sociales, de la Promotion Féminine et de l'Enfance Kaba Saran Daraba, Ministre

Ministère du Plan et de la Coopération Camara Aminata Barry, Secrétaire Général Ousmane Kaba, Ministre

Ministère de la Promotion du Secteur Privé, du Commerce et de l'Industrie Mamadou Niaré, Secrétaire Général Madi Kala Camara, Ministre

Ministère de l'Energie et de Ressources Naturelles Fofana Fassine, Ministre

Ministère d'Economie et des Finances Houssein Traore, Secrétaire Général

Ministère de l'Industrie Pepé Guilavogui, Directeur

Conseil National de Patronat Hadj Boubacar Hann, Président

Union des Industriels de Guinée Aboubacar Boubou Kaba

Groupement des Femmes Entrepreneurs de Guinée Hadia Yarie Kaba

Association Jeunesse Développement Abdourhamane Sano

Office pour la Promotion des Investissements Privés Diallo Abdoulaye Touré Hawa Keita

PNUD/ONUDI

Cyr Mathieu Samaké, Représentant Résident Cissé Namata, Directeur de l'ONUDI pour la Guinée

KENYA

Ministry of Commerce and Industry (MCI)

S. Musila, Director of Industries

J.M. Masila, Director of Industries

S. Mwanangoro, Senior Assistant Director of Industries

E.N. Warui, Senior Industrial Development Officer

M.M. Nzomo, Deputy Director

E.N. Warui, Senior Industrial Development Officer

Ministry of Planning and Development

N. Gitu, Director of Planning, Office of Planning, Office of the Vice President

Investment Promotion Council (IPC)

Francis E. Oyugi, General Manager

Sicily Kariuki, Investment Promotion, Manager

Industrial Development Bank (IDB)

Lawrence A. Masaviru, Managing Director

B.N. Maina, Chief, Corporate Planning

Kenya Industrial Research and Development Institute (KIRDI)

B.W. Maina, Director

H.L. Kaane, Director

P. Muturi, Deputy Director

Kenya Bureau of Standards (KBS)

H.M. Mokaya, Assistant Director

C.O. Oyalo, Assistant Director

E.L. Songole, Assistant Director

J.D. Gikandi, Assistant Director

S.K. Kivagu, Assistant Director

D.M. Ndivo, Assistant Director

E.S. Halwenge, Principal Legal Officer

Export Promotion Council (EPC)

Peter W. Muthoka, Secretary/Chief Executive

Zakary E. Gichohi, Manager, Research Policy

Operations, Export Processing Zones Authority (EPZA)

R.I. Kigunda, Senior Manager

J.O.B. Akara, Research Policy and Planning

Kenya Chamber of Commerce and Industry, (KCCI)

K. Owango, Chairman

Clement O Hala, Chairman, Economic and Development Committee

Federation of Kenya Employers (FKE)

G.N. Konditi, Deputy Executive Director

T.M. Waithaka, Research Officer

Kenva Industrial Estates (KIE)

B.M. Kikuvi, Managing Director

C.N. Orare, Finance Manager

J.H.O. Owade, General Manager, Operations

J.M. Ndoto, DSR-Manager

J.O. Obiero, Projects Implementation Manager

F.M. Ndirangu, Projects Supervision Manager

M.M. W. Munene, Project Preparation and Appraisal Manager

Kenya Organization of Micro Small and Medium Enterprises (KOSME)

T.K. Mugarana, National Chairman

J.A. Omboga, Chief Executive

UNIDO

G. Tabah, Country Director

MALI

Ministère de l'Industrie, de l'Artisanat et du Tourisme

C. Fatou Haidara, Ministre de l'Industrie, de l'Artisanat et du Tourisme

H. Niang, Secrétaire Général de l'Industrie, de l'Artisanat et du Tourisme Adama Konate, Directeur de l'Industrie

Commissariat au Plan

Bacary Diara, Chargé de Mission

Chambre de Commerce, d'Agriculture et d'Industrie

Daba Traore, Secrétaire Général

Salif Diara, Responsable du Bureau Chargé de Commerce

Organisation Patronale des Industriels du Mali

Mamadou Makalou, Secrétaire Général

Agence pour la Promotion des Filières Agricoles (APROFA)

Dominique Traore, Directeur Général

Centre National de Promotion des Investissements

Namori Traoré, Directeur Général Adjoint

Bureau des Entreprises Publiques

Seydou Camara, Directeur P.I.

Agence pour la Promotion des Entreprises Privées (APEP)

Ousman Sidibé, Directeur Général

Banque Malienne de Crédit et de Dépôts

Salif Nambala Keita, Directeur d'Exploitation et des Agences

Fédération Nationale des Employeurs du Mali

Moussa M.B. Coulibaly, Président

MOROCCO

Ministère du Commerce, de l'Industrie et de l'Artisanat

Azzedine El Ghissassi, Chef de la Division de la Planification Industrielle Hamdaoui Quahbia, Direction des Etudes et Planification

Arab Industrial Development and Mining Organization

T. Al Dafer, Director General Assistant

A. Abbas Abbas, Director, Country Services Section Anouar

Centre Africain de Formation et de Recherche Administratives pour le Développement Mohammad Ahmad Wali, Directeur Général

L. Michael, Programme de Formation, Recherche et Consultation

Office pour le Développement Industriel

J. Hevert, Senior Advisor

Multinational Programming and Operational Centre (MULPOC)-Regional Development Centre

E. Ben Saoud, Director

De Pedro José

O. Mariko

UNDP/UNIDO

A.C. Sultanoglu, Deputy Resident Representative

S. Flettner, Programme Officer

NAMIBIA

Ministry of Finance

J. J. Roberts, Commissioner for Inland Revenue Desmond Bason, Deputy Director, Treasury

Ministry of Fisheries and Marine Resources

Shihaleni Ellis Ndjaba, Director of Operations

Ministry of Trade and Industry

Steve Motinga, Director of Industrial Development Ulrich F. Gaoseb, Chief Economist, Industrial Planning Marianne R. Kaoses, Chief Statistician, Industrial Planning Hidipo Hamutenya, Minister of Trade and Industry Hanno Rumpf, Permanent Secretary Stephen Adei, UN Resident Coordinator

Namibia Development Corporation (NDC)

Anna N. Shiweda, General Manager, Agriculture Development David Nuyoma, Manager, Small Enterprise Development

National Planning Commission (NPC)

Andrew Paulus Ndishishi, Permanent Secretary

Namibia National Chamber of Commerce and Industry (NNCCI)

Sam Geiseb, Bureau Director TIPS

Figaii, Staff member dealing with SMEs

D. F. Reissner, Manager, Export/Market Development, Namibia Breweries Limited

Namibia Non-Governmental Organization Forum (NANGOF)

Naftali Uriab, Chairperson

Kahepako Uariua-Kakujaha, Programme Manager

Charles Truebody, Director, Private Sector Foundation (PSF)

Offshore Development Company (Ptv.) Ltd. (ODC)

Milton Lowu, Acting Chief Executive Officer

Bunny Jasrotia

Phillip Namunchobo

William Nkuruh

UNDP

Stephen Adei, UN Resident Coordinator Oddvar Jakobsen, Deputy Resident Representative Lulesa Abadura, Programme Officer

NIGER

Ministère de l'Industrie et de l'Energie

Kada Labo

Rabiou Hassane, Directeur de l'Energie

Gérome Trapsida, Directeur du Développement Industriel

Ministère de l'Agriculture et de l'Élevage

Hassane Saley, Directeur des Etudes et de la Programmation

Ministère des Relations Extérieures

Adani Illo, Secrétaire Général

Migin Y.Aïssa Nee Manzo, Directrice de Participation et des Entreprises Publiques

Nouvelle Imprimerie du Niger

Maman Habou, Directeur Général

AFELEN

Lawaly Chaibou, Chef, Division Communication et Environnement

Bonzougou Abdoulave, Directeur de Développement du Secteur Privé

Maurice Davignon, Chef, Projet Programme, Cadre de Promotion du Secteur Privé

Chambre de Commerce, d'Agriculture, d'Industrie et d'Artisanat

Moustapha Moumouni, Secrétaire Général P.I.

MULPOC, Niamey Souma Henri, Directeur Cissé Birane

Hamid Ahmed, Cellule de Coordination du Programme de Privatisation

PNUD

Alberic Kakou, Représentant Résident Adjoint

NIGERIA

National Committee for Industrial Development (NCID)
Rasheed Gbadamosi, Chairman

Economic Community of West African States (ECOWAS)
Limane Barrage, Chef de Division Industrie

Federal Ministry of Solid Minerals Development (FMSMD)

Alhaji Muhammad Kaloma Ali, Minister of Solid Minerals Development

PNUD

B. Chinsman, Resident Representative

SÉNÉGAL

Ministère de l'Energie, des Mines et de l'Industrie A. Baila Niang, Directeur du Cabine Mamadou Keita, Directeur de l'Industrie

Coopération Economique et Financière Papa Salla M'Boup, Directeur

Abdoulaye Mamadou Ball, Directeur de l'Industrialisation

SONEPI

Al Hadj Mamadou Ly, Directeur Général Damba Faye

Cellule d'appui à l'Environnement des Entreprises Christian Sow

Conseil National de Patronat Sénégalais Tala Kane

Cadre de Promotion du Secteur Privé Cimon Dioh, Responsable du Programme

Chambre de Commerce d'Industrie et d'Artisanat-Région de Dakar Bayo N'Diaye. Secrétaire Général

PNUD

Odile Sorgho-Moulinier, Représentant Résident

TANZANIA

Ministry of Trade and Industry (MTI)

- B.S. Mchomvu, Principal Secretary
- E.E. Mangesho

Board of External Trade (BET)

- M.K. Mwandoro, Director General
- T. Asfaw, Senior Trade Promotion Adviser (CTA)
- R.S. Mwasha, Director of Research and Planning

SADC Industry and Trade Coordination Division (SITCD)

- Z.J. Masanja, Principal Industrial Economist
- J.S.K. Mhina, Principal Industrial Economist
- M. Rweyemamu, Ag. Director

Tanzania Bureau of Standards (TBS)

- B Mutabazi
- C. Ekelege, Head, Engineering Standards Department

Tanzania Industrial Research and Development Organization (TIRDO)

A.P. Nanyaro, Director General

Tanzania Commission for Science and Technology (COSTES)

F.J. Mkwawa, Technology Analyst

Tanzania Chamber of Commerce, Industry and Agriculture (TCCIA)

- M.M. Kalanje, Executive Director
- D. Mwaibulla, President

Cleaner Production Center of Tanzania (CPCT)

- C.L.C. Migiro, Director
- B.S.E. Mndewa, Deputy Director, Demonstration and Training
- Ms. A.N. Magashi, Deputy Director, Information and Policy

Small Industries Development Organization (SIDO)

- E.B. Joroka, Director General
- J. Mwene-Milao, Director, Planning and Technology Development

Confederation of Tanzania Industries (CTI)

- J.A. Mwapachu, Chairman
- O. Kilindu, Chief, Executive

World Bank

V. Rwechungara, Programme Officer (Industry), World Bank Resident Mission

International Monetary Fund

F.L. Osunsade, Senior Resident Representative

TUNISIA

Ministère des Affaires Etrangères

S. Fourati, Directeur, Coopération Internationale

Drig, Chargé de Programme, ONUDI/Agence Atomique

Ministère de l'Industrie

N.L.B. Lazreg, Directeur de la Promotion et de la Programmation Industrielle Agence de Promotion de l'Industrie

T. Ennaifar, Président Directeur Général

M. Zalila, Directeur Principal

Gondji, Officier

Jaloul, Officier

Consultant

Ben Abda, Président Directeur Général, Les Ateliers du Bois Industriel de Tunisie

UNIDO

M. Garzelli, Country Director

UGANDA

Ministry of Trade and Industry (MTI)

A. Rwendeire MP, Minister of State

Permanent Secretary

- J. Kapasi-Kakama, Director of Industry
- J. Mambule, Principal Industrial Officer

Ministry of Finance (MOF)

S. Manso, SFO/Industry

Max Ochai, Economist

O. Obella, Assistant Commissioner

Uganda Investment Authority

W.T. Muhairwe, Deputy Executive Director

Uganda Export Promotion Board (UEPB)

R.K. Rutaagi, Executive Director

C. Namwanga, Trade Information Officer

Ogwal. Research Officer

S Ojakol, Research Officer

Bank of Uganda

L.A. Kasekende, Executive Director, Research and Policy Function

Uganda Investment Authority

A.B. Abariwano, Chairman

Uganda Bureau of Standards

M. Ekisa, Administrative Officer

East African Development Bank (EADB)

- F.R. Tibeita, Director General
- J.G. Nduati, Director, Operations

Uganda National Chamber of Commerce and Agriculture (UNCCIA)

- T. Byara, Secretary General
- G. Rujojo, Deputy Secretary General
- S. Mukyne, Chief Field Officer
- J.L. Walama, Secretary
- P. Wamanga, Ag. Chief Accountant
- S. Twbone, Jinja
- M. Wanume, Director, Eastern Region

MOPED

W. Van Rij, Adviser, Micro and Small Enterprises Development Policy Unit MOPED

Private Sector Foundation (PSF)

- S.B. Rutega, Executive Director
- G. Musoke Lwanga, Project Co-ordinator

Leather Industries of Uganda Ltd.

Bashir Bata

Uganda Industrial Research Institute

A.R. Kaahwa, Co-ordinator

UNDP/UNIDO

- B. Thomas, Resident Representative
- A.C. Chuma, Deputy Resident Representative
- P. Tremmel, Programme Officer
- L. Atekit, Programme Assistant

ZAMBIA

Ministry of Commerce, Trade and Industry (MCTI)

- K. Muleya, Permanent Secretary
- K. Maddison, Industrial Adviser
- D. Mauzu, Director of Industry
- M.F. Malundu, Senior Economist

Ministry of Finance

J. Mtongo, Permanent Secretary

Zambia Chamber of Commerce and Industry (ZACCI)

R. Mukala, Information Specialist

Common Market for Eastern and Southern Africa

J.E.O. Mwencha, Ag. Secretary General

J.A. Alele Opio, Ag. Director, Industry, Energy and Environment

M. Kanyangarara, Metals and Engineering Industrial Development

M. Sichilima, Statistician/Economist

E. Mashasha, Computer Operator

Technology Development Advisory Unit (TDAU)

A.W. K. Tembo, Acting Manager

H.M. Mwenda, NPPD

Zambia Investment Centre

M.S. Chilwesa, Vice Chairman, Board of Directors Bwalya Ngandu, Director General

Small Scale Industries Association of Zambia (SSIAZ)

R.M. Namala, Trustee

M.S. Chilwesa, Executive Secretary

Zambia Privatization Authority (ZPA)

Ian Fraser

Multinational Programming and Operational Centre (MULPOC)

S.A. Ochola, Acting Director, ECA

W. Lombe, Economic Affairs Officer, ECA

B.C. Gondwe, Economic Affairs Officer, ECA

UNDP

M.C. Soko, Assistant Resident Representative

O. Oyaide, Deputy Resident Representative

J.C. Charlier, Programme Officer

Siteke G. Mwale, Executive Chairman, Consultant

N. Biering, Country Director

ZIMBABWE

Ministry of Industry and Commerce

O.M. Tshabangu, Deputy Secretary, Enterprise Development & Consumer Affairs F. Bango, Head, Chemicals and Industrial Co-operation Section

Employers' Confederation of Zimbabwe (EmCoZ)

Harrison T. Mudede, Director, Training Services

Scientific and Industrial Research and Development Centre (SIRDC)

C.J. Chetsanga, Director General

Phanuel Munezvenyu, Deputy Director General

Mukai T. Musarurwa, Director of Building Technology Institute

Industrial Development Corporation (IDC)

L. A. Munyawarana, Deputy General Manager, Business Development

The African Capacity Building Foundation (ACBF)
Abel L. Thoahlane, Executive Secretary
Gene Ogiogio, Principal Programme Officer

Small Enterprises Development Corporation (SEDCO)

Muchada, Development Manager

Central Statistical Office (CSO)

Lazarus Machirovi, Director

M. B. Nyoni, Deputy Director

K.R.N. Shoniwa, Assistant Director

Inponente Tannery, Superior Footwear Group of Companies Carlos J.O. Costa, Managing Director

Zimbabwe Investment Centre (ZIC)
Eliana Ambayi, Assistant Director, Research and Policy Analysis
Tapiwa Huye, Research and Policy Analysis Executive

UNDP

Carlos Lopes, United Nations Resident Coordinator